OP \$965.00 90176960

TRADEMARK ASSIGNMENT COVER SHEET

Electronic Version v1.1 Stylesheet Version v1.2 ETAS ID: TM644411

SUBMISSION TYPE:	NEW ASSIGNMENT
NATURE OF CONVEYANCE:	MERGER
EFFECTIVE DATE:	11/20/2020
SEQUENCE:	2

CONVEYING PARTY DATA

Name	Formerly	Execution Date	Entity Type
Onyx Enterprises Int'l, Corp.		11/20/2020	Corporation:

RECEIVING PARTY DATA

Name:	Excel Merger Sub II, LLC
Street Address:	1 Corporate Drive
City:	Cranbury
State/Country:	NEW JERSEY
Postal Code:	08512
Entity Type:	Limited Liability Company: DELAWARE

PROPERTY NUMBERS Total: 38

Property Type	Number	Word Mark	
Serial Number:	90176960	PARTSID	
Serial Number:	90242594	RIXXU	
Serial Number:	90242260	TORXE	
Serial Number:	90162198	LEGASEE	
Serial Number:	90158620		
Serial Number:	90309826	TIREID	
Serial Number:	90309806	RIMSID	
Serial Number:	90384986	ID AUTO	
Serial Number:	90006703	DON'T SETTLE FOR STOCK	
Serial Number:	88457485	REALLY LOVE	
Serial Number:	88457469	REALLY, REALLY LOVE	
Serial Number:	88734396	ID	
Serial Number:	88211256	ID	
Serial Number:	88303678	RACINGID	
Serial Number:	88732234	CARID	
Serial Number:	88352524	WHAT DRIVES YOU	
Serial Number:	88303637	STREETID	

TRADEMARK REEL: 007280 FRAME: 0633

900614436

Property Type	Number	Word Mark
Serial Number:	88238749	RIXXU
Serial Number:	88238723	TORXE
Serial Number:	88238670	TRUCKID
Serial Number:	88238623	TOOLSID
Serial Number:	88238588	RECREATIONID
Serial Number:	88238532	MOTORCYCLEID
Serial Number:	88238501	POWERSPORTSID
Serial Number:	88238468	CAMPERID
Serial Number:	88238416	BOATID
Serial Number:	87412406	ID
Serial Number:	86253327	ID
Serial Number:	86043553	ONYX
Serial Number:	85437004	STREETID
Serial Number:	85117082	STREETID
Serial Number:	85317984	TRUCKID
Serial Number:	77718242	CARID
Serial Number:	77718230	CARID
Serial Number:	77654112	WOODVIEW
Serial Number:	77654103	WOODVIEW
Serial Number:	77608200	WOODVIEW AUTOMOTIVE ACCESSORIES
Serial Number:	77595889	WOODVIEW AUTOMOTIVE ACCESSORIES

CORRESPONDENCE DATA

Fax Number:

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent using a fax number, if provided; if that is unsuccessful, it will be sent via US Mail.

Phone: 7204414217
Email: nina@apb-law.com

Correspondent Name: Nina Brewer

Address Line 1: 2701 Lawrence Street

Address Line 4: Denver, COLORADO 80205

NAME OF SUBMITTER:	Nina Brewer
SIGNATURE:	/Nina Brewer/
DATE SIGNED:	05/05/2021

Total Attachments: 147

source=BE_Copies_0100992057_210122203031 (1)#page1.tif source=BE_Copies_0100992057_210122203031 (1)#page2.tif source=BE_Copies_0100992057_210122203031 (1)#page3.tif source=BE_Copies_0100992057_210122203031 (1)#page4.tif source=BE_Copies_0100992057_210122203031 (1)#page5.tif

```
source=BE Copies 0100992057 210122203031 (1)#page6.tif
source=BE Copies 0100992057 210122203031 (1)#page7.tif
source=BE Copies 0100992057 210122203031 (1)#page8.tif
source=BE Copies 0100992057 210122203031 (1)#page9.tif
source=BE_Copies_0100992057_210122203031 (1)#page10.tif
source=BE Copies 0100992057 210122203031 (1)#page11.tif
source=BE Copies 0100992057 210122203031 (1)#page12.tif
source=BE_Copies_0100992057_210122203031 (1)#page13.tif
source=BE_Copies_0100992057_210122203031 (1)#page14.tif
source=BE Copies 0100992057 210122203031 (1)#page15.tif
source=BE Copies 0100992057 210122203031 (1)#page16.tif
source=BE_Copies_0100992057_210122203031 (1)#page17.tif
source=BE Copies 0100992057 210122203031 (1)#page18.tif
source=BE Copies 0100992057 210122203031 (1)#page19.tif
source=BE_Copies_0100992057_210122203031 (1)#page20.tif
source=BE_Copies_0100992057_210122203031 (1)#page21.tif
source=BE Copies 0100992057 210122203031 (1)#page22.tif
source=BE Copies 0100992057 210122203031 (1)#page23.tif
source=BE Copies 0100992057 210122203031 (1)#page24.tif
source=BE Copies 0100992057 210122203031 (1)#page25.tif
source=BE Copies 0100992057 210122203031 (1)#page26.tif
source=BE Copies 0100992057 210122203031 (1)#page27.tif
source=BE Copies 0100992057 210122203031 (1)#page28.tif
source=BE Copies 0100992057 210122203031 (1)#page29.tif
source=BE Copies 0100992057 210122203031 (1)#page30.tif
source=BE_Copies_0100992057_210122203031 (1)#page31.tif
source=BE Copies 0100992057 210122203031 (1)#page32.tif
source=BE Copies 0100992057 210122203031 (1)#page33.tif
source=BE Copies 0100992057 210122203031 (1)#page34.tif
source=BE_Copies_0100992057_210122203031 (1)#page35.tif
source=BE Copies 0100992057 210122203031 (1)#page36.tif
source=BE Copies 0100992057 210122203031 (1)#page37.tif
source=BE_Copies_0100992057_210122203031 (1)#page38.tif
source=BE_Copies_0100992057_210122203031 (1)#page39.tif
source=BE_Copies_0100992057_210122203031 (1)#page40.tif
source=BE Copies 0100992057 210122203031 (1)#page41.tif
source=BE_Copies_0100992057_210122203031 (1)#page42.tif
source=BE_Copies_0100992057_210122203031 (1)#page43.tif
source=BE Copies 0100992057 210122203031 (1)#page44.tif
source=BE Copies 0100992057 210122203031 (1)#page45.tif
source=BE_Copies_0100992057_210122203031 (1)#page46.tif
source=BE Copies 0100992057 210122203031 (1)#page47.tif
source=BE Copies 0100992057 210122203031 (1)#page48.tif
source=BE Copies 0100992057 210122203031 (1)#page49.tif
source=BE Copies 0100992057 210122203031 (1)#page50.tif
source=BE Copies 0100992057 210122203031 (1)#page51.tif
source=BE_Copies_0100992057_210122203031 (1)#page52.tif
source=BE_Copies_0100992057_210122203031 (1)#page53.tif
```

```
source=BE Copies 0100992057 210122203031 (1)#page54.tif
source=BE Copies 0100992057 210122203031 (1)#page55.tif
source=BE Copies 0100992057 210122203031 (1)#page56.tif
source=BE Copies 0100992057 210122203031 (1)#page57.tif
source=BE_Copies_0100992057_210122203031 (1)#page58.tif
source=BE Copies 0100992057 210122203031 (1)#page59.tif
source=BE Copies 0100992057 210122203031 (1)#page60.tif
source=BE_Copies_0100992057_210122203031 (1)#page61.tif
source=BE_Copies_0100992057_210122203031 (1)#page62.tif
source=BE Copies 0100992057 210122203031 (1)#page63.tif
source=BE Copies 0100992057 210122203031 (1)#page64.tif
source=BE_Copies_0100992057_210122203031 (1)#page65.tif
source=BE Copies 0100992057 210122203031 (1)#page66.tif
source=BE Copies 0100992057 210122203031 (1)#page67.tif
source=BE_Copies_0100992057_210122203031 (1)#page68.tif
source=BE_Copies_0100992057_210122203031 (1)#page69.tif
source=BE Copies 0100992057 210122203031 (1)#page70.tif
source=BE Copies 0100992057 210122203031 (1)#page71.tif
source=BE_Copies_0100992057_210122203031 (1)#page72.tif
source=BE Copies 0100992057 210122203031 (1)#page73.tif
source=BE Copies 0100992057 210122203031 (1)#page74.tif
source=BE Copies 0100992057 210122203031 (1)#page75.tif
source=BE Copies 0100992057 210122203031 (1)#page76.tif
source=BE Copies 0100992057 210122203031 (1)#page77.tif
source=BE Copies 0100992057 210122203031 (1)#page78.tif
source=BE_Copies_0100992057_210122203031 (1)#page79.tif
source=BE Copies 0100992057 210122203031 (1)#page80.tif
source=BE Copies 0100992057 210122203031 (1)#page81.tif
source=BE Copies 0100992057 210122203031 (1)#page82.tif
source=BE_Copies_0100992057_210122203031 (1)#page83.tif
source=BE Copies 0100992057 210122203031 (1)#page84.tif
source=BE Copies 0100992057 210122203031 (1)#page85.tif
source=BE_Copies_0100992057_210122203031 (1)#page86.tif
source=BE_Copies_0100992057_210122203031 (1)#page87.tif
source=BE_Copies_0100992057_210122203031 (1)#page88.tif
source=BE Copies 0100992057 210122203031 (1)#page89.tif
source=BE_Copies_0100992057_210122203031 (1)#page90.tif
source=BE_Copies_0100992057_210122203031 (1)#page91.tif
source=BE Copies 0100992057 210122203031 (1)#page92.tif
source=BE Copies 0100992057 210122203031 (1)#page93.tif
source=BE_Copies_0100992057_210122203031 (1)#page94.tif
source=BE Copies 0100992057 210122203031 (1)#page95.tif
source=BE Copies 0100992057 210122203031 (1)#page96.tif
source=BE Copies 0100992057 210122203031 (1)#page97.tif
source=BE Copies 0100992057 210122203031 (1)#page98.tif
source=BE Copies 0100992057 210122203031 (1)#page99.tif
source=BE_Copies_0100992057_210122203031 (1)#page100.tif
source=BE Copies 0100992057 210122203031 (1)#page101.tif
```

```
source=BE Copies 0100992057 210122203031 (1)#page102.tif
source=BE Copies 0100992057 210122203031 (1)#page103.tif
source=BE Copies 0100992057 210122203031 (1)#page104.tif
source=BE Copies 0100992057 210122203031 (1)#page105.tif
source=BE_Copies_0100992057_210122203031 (1)#page106.tif
source=BE Copies 0100992057 210122203031 (1)#page107.tif
source=BE Copies 0100992057 210122203031 (1)#page108.tif
source=BE Copies 0100992057 210122203031 (1)#page109.tif
source=BE_Copies_0100992057_210122203031 (1)#page110.tif
source=BE Copies 0100992057 210122203031 (1)#page111.tif
source=BE Copies 0100992057 210122203031 (1)#page112.tif
source=BE_Copies_0100992057_210122203031 (1)#page113.tif
source=BE Copies 0100992057 210122203031 (1)#page114.tif
source=BE Copies 0100992057 210122203031 (1)#page115.tif
source=BE_Copies_0100992057_210122203031 (1)#page116.tif
source=BE_Copies_0100992057_210122203031 (1)#page117.tif
source=BE Copies 0100992057 210122203031 (1)#page118.tif
source=BE Copies 0100992057 210122203031 (1)#page119.tif
source=BE Copies 0100992057 210122203031 (1)#page120.tif
source=BE Copies 0100992057 210122203031 (1)#page121.tif
source=BE Copies 0100992057 210122203031 (1)#page122.tif
source=BE Copies 0100992057 210122203031 (1)#page123.tif
source=BE Copies 0100992057 210122203031 (1)#page124.tif
source=BE Copies 0100992057 210122203031 (1)#page125.tif
source=BE Copies 0100992057 210122203031 (1)#page126.tif
source=BE_Copies_0100992057_210122203031 (1)#page127.tif
source=BE Copies 0100992057 210122203031 (1)#page128.tif
source=BE Copies 0100992057 210122203031 (1)#page129.tif
source=BE Copies 0100992057 210122203031 (1)#page130.tif
source=BE_Copies_0100992057_210122203031 (1)#page131.tif
source=BE Copies 0100992057 210122203031 (1)#page132.tif
source=BE Copies 0100992057 210122203031 (1)#page133.tif
source=BE_Copies_0100992057_210122203031 (1)#page134.tif
source=BE_Copies_0100992057_210122203031 (1)#page135.tif
source=BE Copies 0100992057 210122203031 (1)#page136.tif
source=BE Copies 0100992057 210122203031 (1)#page137.tif
source=BE_Copies_0100992057_210122203031 (1)#page138.tif
source=BE_Copies_0100992057_210122203031 (1)#page139.tif
source=BE Copies 0100992057 210122203031 (1)#page140.tif
source=BE Copies 0100992057 210122203031 (1)#page141.tif
source=BE_Copies_0100992057_210122203031 (1)#page142.tif
source=BE Copies 0100992057 210122203031 (1)#page143.tif
source=BE Copies 0100992057 210122203031 (1)#page144.tif
source=BE Copies 0100992057 210122203031 (1)#page145.tif
source=BE Copies 0100992057 210122203031 (1)#page146.tif
source=BE Copies 0100992057 210122203031 (1)#page147.tif
```

STATE OF NEW JERSEY DEPARTMENT OF THE TREASURY FILING CERTIFICATE (CERTIFIED COPY)

Corporation Name: ONYX ENTERPRISES INT'L, CORP.

Business Id: 0100992057 Certificate Number: 6000135232

I, THE TREASURER OF THE STATE OF NEW JERSEY, DO HEREBY CERTIFY, THAT THE ABOVE NAMED BUSINESS DID FILE AND RECORD IN THIS DEPARTMENT A MERGER ON November 23, 2020 AND THAT THE ATTACHED IS A TRUE COPY OF THIS DOCUMENT AS THE SAME IS TAKEN FROM AND COMPARED WITH THE ORIGINAL(S) FILED IN THIS OFFICE AND NOW REMAINING ON FILE AND OF RECORD.

IN TESTIMONY WHEREOF, I HAVE HEREUNTO SET MY HAND AND AFFIXED MY OFFICIAL SEAL AT TRENTON, THIS

January 12, 2021 A.D.



Sulp of Muso
ELIZABETH MAHER MUOIO
STATE TREASURER

VERIFY THIS CERTIFICATE ONLINE AT

https://wwwl.state.nj.us/TYTR_StandingCert/JSP/Verify_Cert.jsp

MRG

FILED

NOV 2 3 2020

STATE TREASURER

Certificate of Merger of

Excel Merger Sub I, Inc. (a Delaware corporation) into and with

Onyx Enterprises Int'l, Corp. (a New Jersey corporation)

The undersigned corporations have adopted an agreement and plan of merger pursuant to 14A:10-1 and pursuant to which Excel Merger Sub I, Inc., a Delaware corporation, will be merged into and with Onyx Enterprises Int'l, Corp., a New Jersey corporation. The agreement and plan of merger is the Business Combination Agreement among Legacy Acquisition Corp., Excel Merger Sub I, Inc., Excel Merger Sub II, LLC, Onyx Enterprises Int'l, Corp., and Shareholder Representative Services LLC, as stockholder representative, entered into on September 18, 2020, attached to this Certificate of Merger.

- 1. This is a merger filing.
- 2. Name of Surviving Business Entity: Onyx Enterprises Int'l, Corp.
- 3. Names and Jurisdictions of all participating corporations including Surviving Corporation:

Name	Jurisdiction	Identification # Assigned by Treasurer (if applicable)
Onyx Enterprises Int'l, Corp. Excel Merger Sub I, Inc.	New Jersey Delaware	0100992057 N/A

- Date Merger Adopted: November 3, 2020
- 5. Voting:
 - A. Onyx Enterprises Int'l, Corp.

Outstanding Shares:

The number of shares outstanding is 1,000,417 shares, 417 shares of which is common stock, and 1,000,000 shares of which is preferred stock. All of the authorized shares of preferred stock are designated "Series A Preferred Stock."

The number of shares entitled to vote on the merger plan was 417 shares of common stock and 1,000,000 shares of preferred stock.

Voting For:

217 shares of common stock voted <u>for</u> adopting the plan of merger. 1,000,000 shares of preferred stock voted for adopting the plan of merger.

Voting Against:

200 shares of common stock voted against adopting the plan of merger.

S 3109375 75448197

B. Excel Merger Sub I, Inc.

Outstanding Shares:

The number of shares outstanding is 1,000 shares of common stock. The number of shares entitled to vote on the merger plan was 1,000 shares of common stock.

Merger plan was adopted by the unanimous written consent of the sole shareholder without a meeting.

The applicable provisions of the laws of Delaware have been complied with.

- 6. The surviving business entity agrees that it may be served with process in this State in any action, suit, or proceeding for the enforcement of any obligation of any domestic or foreign corporation, previously amenable to suit in this State, which is party to this merger, and in any proceeding for the enforcement of the rights of a dissenting shareholder of such domestic corporation against the surviving corporation.
 The surviving business entity also agrees that it will promptly pay the dissenting shareholders of any such domestic corporation the amount, if any, to which they may be entitled under the provisions of Title 14A.
- 7. The effective time and date of the merger is the time and date when filed
- 8. Simultaneous with the effective time and date of the merger, the amended and restated articles of incorporation of Onyx Enterprises Int'l, Corp. attached to this Certificate of Merger shall become the articles of incorporation of the surviving corporation.

Each of the undersigned individuals signed this Certificate of Merger on the 20th day of November, 2020 on behalf of the respective corporation to the merger listed above his name.

Excel Merger Sub I, Inc.	Onyx Enterprises Int'l, Corp.
Name: Dayfyl McCall Title: President	Name: Antonino Ciappina Title: Interim General Manager, acting President, acting Chief Executive Officer, and acting Secretary
	Onyx Enterprises Int'l, Corp.

Name: Mark Atwater Title: Vice President B. Excel Merger Sub I, Inc.

Outstanding Shares:

The number of shares outstanding is 1,000 shares of common stock. The number of shares entitled to vote on the merger plan was 1,000 shares of common stock.

Merger plan was adopted by the unanimous written consent of the sole shareholder without a meeting.

The applicable provisions of the laws of Delaware have been complied with.

- 6. The surviving business entity agrees that it may be served with process in this State in any action, suit, or proceeding for the enforcement of any obligation of any domestic or foreign corporation, previously amenable to suit in this State, which is party to this merger, and in any proceeding for the enforcement of the rights of a dissenting shareholder of such domestic corporation against the surviving corporation.
 The surviving business entity also agrees that it will promptly pay the dissenting shareholders of any such domestic corporation the amount, if any, to which they may be entitled under the provisions of Title 14A.
- 7. The effective time and date of the merger is ____ the time and date when filed
- 8. Simultaneous with the effective time and date of the merger, the amended and restated articles of incorporation of Onyx Enterprises Int'l, Corp. attached to this Certificate of Merger shall become the articles of incorporation of the surviving corporation.

Each of the undersigned individuals signed this Certificate of Merger on the 20th day of November, 2020 on behalf of the respective corporation to the merger listed above his name.

Excel Merger Sub I, Inc.	Onyk Enterprises Int'l, Corp.
Name: Darryl McCall	Name: Antonino Ciappina
Title: President	Title: Interim General Manager, acting
	President, acting Chief Executive Officer, and acting Secretary

Onyx Enterprises Int'l, Corp.

Name: Mark Atwater Title: Vice President

STATE OF NEW JERSEY DEPARTMENT OF THE TREASURY FILING CERTIFICATE (CERTIFIED COPY)

Corporation Name: ONYX ENTERPRISES INT'L, CORP.

Business Id: 0100992057 Certificate Number: 6000135231

I, THE TREASURER OF THE STATE OF NEW JERSEY, DO HEREBY CERTIFY, THAT THE ABOVE NAMED BUSINESS DID FILE AND RECORD IN THIS DEPARTMENT A MERGER ON November 20, 2020 AND THAT THE ATTACHED IS A TRUE COPY OF THIS DOCUMENT AS THE SAME IS TAKEN FROM AND COMPARED WITH THE ORIGINAL(S) FILED IN THIS OFFICE AND NOW REMAINING ON FILE AND OF RECORD.

IN TESTIMONY WHEREOF, I HAVE HEREUNTO SET MY HAND AND AFFIXED MY OFFICIAL SEAL AT TRENTON, THIS

January 12, 2021 A.D.



Sulp of Muso
ELIZABETH MAHER MUOIO
STATE TREASURER

VERIFY THIS CERTIFICATE ONLINE AT

https://wwwl.state.nj.us/TYTR_StandingCert/JSP/Verify_Cert.jsp

MRG FILED NOV 2 0 2020

Certificate of Merger of

Excel Merger Sub I, Inc. (a Delaware corporation) into and with

Onyx Enterprises Int'l, Corp. (a New Jersey corporation) 010099205

The undersigned corporations have adopted an agreement and plan of merger pursuant to 14A:10-1 and pursuant to which Excel Merger Sub I, Inc., a Delaware corporation, will be merged into and with Onyx Enterprises Int'l, Corp., a New Jersey corporation. The agreement and plan of merger is the Business Combination Agreement among Legacy Acquisition Corp., Excel Merger Sub I, Inc., Excel Merger Sub II, LLC, Onyx Enterprises Int'l, Corp., and Shareholder Representative Services LLC, as stockholder representative, entered into on September 18, 2020, attached to this Certificate of Merger.

- 1. This is a merger filing.
- 2. Name of Surviving Business Entity: Onyx Enterprises Int'l, Corp.
- 3. Names and Jurisdictions of all participating corporations including Surviving Corporation:

Name	Jurisdiction	Identification # Assigned by Treasurer (if applicable)
Onyx Enterprises Int'l, Corp.	New Jersey	0100992057
Excel Merger Sub I, Inc.	Delaware	N/A

- 4. Date Merger Adopted: November 3, 2020
- 5. Voting:
 - A. Onyx Enterprises Int'l, Corp.

Outstanding Shares:

The number of shares outstanding is 1,000,417 shares, 417 shares of which is common stock, and 1,000,000 shares of which is preferred stock. All of the authorized shares of preferred stock are designated "Series A Preferred Stock."

The number of shares entitled to vote on the merger plan was 417 shares of common stock and 1,000,000 shares of preferred stock.

Voting For:

217 shares of common stock voted <u>for</u> adopting the plan of merger. 1,000,000 shares of preferred stock voted <u>for</u> adopting the plan of merger.

Voting Against:

200 shares of common stock voted against adopting the plan of merger.

B. Excel Merger Sub I, Inc.

Outstanding Shares:

The number of shares outstanding is 1,000 shares of common stock. The number of shares entitled to vote on the merger plan was 1,000 shares of common stock.

Merger plan was adopted by the unanimous written consent of the sole shareholder without a meeting.

The applicable provisions of the laws of Delaware have been complied with.

- 6. The surviving business entity agrees that it may be served with process in this State in any action, suit, or proceeding for the enforcement of any obligation of any domestic or foreign corporation, previously amenable to suit in this State, which is party to this merger, and in any proceeding for the enforcement of the rights of a dissenting shareholder of such domestic corporation against the surviving corporation. The surviving business entity also agrees that it will promptly pay the dissenting shareholders of any such domestic corporation the amount, if any, to which they may be entitled under the provisions of Title 14A.
- The effective time and date of the merger is the time and date when filed 7.
- Simultaneous with the effective time and date of the merger, the amended and restated 8. articles of incorporation of Onyx Enterprises Int'l, Corp. attached to this Certificate of Merger shall become the articles of incorporation of the surviving corporation.

Each of the undersigned individuals signed this Certificate of Merger on the 20th day of November , 2020 on behalf of the respective corporation to the merger listed above his name.

<i>•</i> /	er Sub I, Inc.
Name: Day	yl McCall

Title: President

Onyx Enterprises Int'l, Corp.

Name: Antonino Ciappina

Title: Interim General Manager, acting President, acting Chief Executive Officer,

and acting Secretary

Onyx Enterprises Int'l, Corp.

Name: Mark Atwater Title: Vice President

B. Excel Merger Sub I, Inc.

Outstanding Shares:

The number of shares outstanding is 1,000 shares of common stock. The number of shares entitled to vote on the merger plan was 1,000 shares of common stock.

Merger plan was adopted by the unanimous written consent of the sole shareholder without a meeting.

The applicable provisions of the laws of Delaware have been complied with.

- 6. The surviving business entity agrees that it may be served with process in this State in any action, suit, or proceeding for the enforcement of any obligation of any domestic or foreign corporation, previously amenable to suit in this State, which is party to this merger, and in any proceeding for the enforcement of the rights of a dissenting shareholder of such domestic corporation against the surviving corporation.
 The surviving business entity also agrees that it will promptly pay the dissenting shareholders of any such domestic corporation the amount, if any, to which they may be entitled under the provisions of Title 14A.
- 7. The effective time and date of the merger is ____ the time and date when filed
- 8. Simultaneous with the effective time and date of the merger, the amended and restated articles of incorporation of Onyx Enterprises Int'l, Corp. attached to this Certificate of Merger shall become the articles of incorporation of the surviving corporation.

Each of the undersigned individuals signed this Certificate of Merger on the 20th day of November, 2020 on behalf of the respective corporation to the merger listed above his name.

Name: Darryl McCall
Title: President

Title: President

Name: Antonino Ciappina
Title: Interim General Manager, acting President, acting Chief Executive Officer, and acting Secretary

Onyx Enterprises Int'l, Corp.

Name: Mark Atwater Title: Vice President

BUSINESS COMBINATION AGREEMENT

among

LEGACY ACQUISITION CORP.,

EXCEL MERGER SUB I, INC.,

EXCEL MERGER SUB II, LLC,

and

ONYX ENTERPRISES INT'L CORP.,

and

SHAREHOLDER REPRESENTATIVE SERVICES LLC, AS STOCKHOLDER REPRESENTATIVE

Dated as of September 18, 2020

TABLE OF CONTENTS

AR	TICLE	E 1 THE MERGERS	2
	1.1	Merger; Surviving Company	
	1.2	Effective Times	2
	1.3	Effects of the Mergers	
	1.4	Conversion of First Surviving Corporation Shares	4
	1.5	Fractional Shares	.,. 4
	1.6	Exchange of Shares	
	1.7	Notices to Stockholders; Dissenting Shares	6
	1.8	Transfer Taxes	6
AR	TICLE	2 AGGREGATE PURCHASE PRICE; CLOSING; AND POST-CLOSING	
AD		IENT	7
	2.1	Aggregate Purchase Price.	
	2.2	Calculation of Estimated Purchase Price for Closing	
	2.3	Closing	
	2.4	Closing Payments	
	2.5	Purchase Price Adjustment	
	2.6	Withholding	
	2.7	Lock-Up Period; Stop-Transfer Notices	
	2.8	Equity Incentive Plan	13
AR		3 REPRESENTATIONS AND WARRANTIES OF THE COMPANY	
	3.1	Organization and Good Standing	
	3.2	Capitalization	.14
	3.3	Authority and Authorization; Conflicts; Consents	.14
	3.4	Financial Statements and Undisclosed Liabilities	
	3.5	Taxes	
	3.6	Litigation and Orders.	
	3.7	Compliance with Law; Permits	
	3.8	Contracts	
	3.9	Title to and Condition of Certain Assets	
	3.10	Certain Accounts	
	3.11	Real Property	
	3.12	Environmental Matters	
	3.13	Intellectual Property	21
	3.14	Insurance.	
	3.15	Absence of Certain Events	
	3.16	Employee Benefits	
	3.17	Employees and Labor Relations	
	3.18	Certain Business Relationships	
	3.19	Vendors and Subcontractors	
	3.20	Brokers	
	3.21	No Other Representations or Warranties	.26
		4 REPRESENTATIONS AND WARRANTIES OF MERGER SUB AND MERGER	
SUE	3 2		
	4.1	Organization and Good Standing	
	4.2	Authority and Authorization; Conflicts; Consents	
	4.3	Capitalization	
	4.4	Litigation and Orders	.28

4.	5 Brokers	28
ARTIC	CLE 5 REPRESENTATIONS AND WARRANTIES OF BUYER	28
5.		
5.	the state of the s	
5.	11	
5.	4 Capitalization	29
5.		
5.		
5.		
5.		
5.	9 Buyer Major Contracts	33
5.	10 Properties	34
5.	11 Insurance	34
5.	12 Absence of Certain Events	34
5.	13 Employees and Employee Benefit Plans	34
5.	14 Certain Business Relationships	34
5.	15 Investment Company Act	
5.	16 Ownership of Transaction Consideration	
5.	17 Trust Account	
5.	18 Lock-Up Agreements	35
5.	19 Securities	35
5.3		
5.2		
5.2		
ARTIC	LE 6 CERTAIN COVENANTS	
6.		
6.3		
6.3	Further Assurances	39
6.4	4 Confidentiality and Publicity	39
6.:		
6.0	Post-Closing Board of Directors	40
	LE 7 CERTAIN COVENANTS OF THE COMPANY AND STOCKHOLDERS	
7.		
7.2		42
7.3	1 · · · · · · · · · · · · · · · · · · ·	
7.4	· · · · · · · · · · · · · · · · · · ·	
7.5	Stockholders Agreement	43
A TO DETECT	T TO CENTS AN COMMAND OF DAMED	43
	LE 8 CERTAIN COVENANTS OF BUYER	
8.3		
8.2	To the second se	
8.3		
8.4	, , ,	
8.5		
8.6		
8.7	. O O O O	
8.8		
8.9		
8.1	Stockholder Consent and Information Statement	49

	8.11	Listing of Buyer Common Stock	51
	8.12	Buyer Public Filings	
	8.13	Documents and Information	
	8.14	Post-Closing Board of Directors and Executive Officers	51
	8.15	Tax Status of Merger Sub 2	
	8.16	Warrant Amendment	
AR	TICLE	9 CLOSING DELIVERIES AND TERMINATION	53
***	9.1	Closing Deliveries by the Company	
	9.2	Closing Deliveries by Buyer, Merger Sub and Merger Sub 2	53
	9.3	Termination of Agreement	54
	9.4	Effect of Termination	
A ID	TICLE	10 CONDITIONS TO OBLIGATIONS TO CLOSE	55
AR	10.1	Conditions to Obligations of Each Party to Close	
	10.1	Conditions to Obligation of Buyer, Merger Sub and Merger Sub 2 to Close	
	10.2	Conditions to Obligation of the Company to Close	
	10.5	Conditions to Congation of the Company to Close	0
AR	TICLE	11 CERTAIN GENERAL TERMS AND OTHER AGREEMENTS	
	11.1	Waiver of Claims Against Trust Account	
	11.2	Notices	
	11.3	No Survival	60
	11.4	Expenses	60
	11.5	Interpretation; Construction	60
	11.6	No Third-Party Beneficiaries	61
	11.7	Governing Law	61
	11.8	Jurisdiction, Venue and Waiver of Jury Trial	61
	11.9	Entire Agreement; Amendment; Waiver	61
	11.10	Assignment; Binding Effect	62
	11.11	Severability; Blue-Pencil	62
	11.12	Counterparts	62
		Disclosure Schedules.	
		Legal Representation; Conflicts Waiver	
		Specific Enforcement	
		Appointment of Stockholder Representative	
ΑĐ	ייו בייו	12 CERTAIN DEFINITIONS	<i>(2</i>
AK	ユエレレル	14 CENTAIN DEFINITIONS	

GLOSSARY OF DEFINED TERMS

Definition	Location
Adjustment Reserve Amount	Section 2 1/h)
Affiliate	
Agreement Dynama Dries	
Aggregate Purchase Price.	Section 2.7(a)
Allocation Schedule	Section 2.2(c)
Amended Buyer Charter	A:-1- 12
Ancillary Document	
Annual Balance Sheet Date	
Antitust Laws	
Applicable Law	
Book-Entry Shares	Section 1.3(k)
Business	Introduction
Business Combination	Section 11.1
Business Day	
Buyer	
Buyer Certificate of Incorporation	
Buyer Class F Common Stock	
Buyer Common Stock	
Buyer Disclosure Schedule	
Buyer Financials	Section 5.5(c)
Buyer Major Contract	Section 5.9(a)
Buyer Preferred Stock	Article 12
Buyer Private Placement Warrants	Section 8.16(a)
Buyer Public Unit	Article 12
Buyer Public Warrants	Article 12
Buyer Required Stockholder Approvals	Article 12
Buyer Securities	Article 12
Buyer Sponsor	Article 12
Buyer Stockholder Approval Matters	Section 8.10(b)
Buyer Stockholders' Written Consent	Section 8.10(a)
Buyer Transaction Expenses	
Buyer's Proposed Calculations	
Cash	
Certificate	
Certificates of Merger	
Clayton Act	Article 12
Closing	Section 2.3
Closing Date	Section 2.3
Closing Filing	Section 8 6/0 Vii)
Closing Indebtedness	
Closing Net Working Capital.	
Closing Press Release	
Closing Share Consideration	
COBRA COVERNO	Castian 9 1(a)
COBRA Coverage	A 4:1-1-10
Code Shar Mana Caraidantian	
Common Share Merger Consideration	Section 1.3(1)
Common Stockholders	Article 12

Company	
Company Common Stock	Article 12
Company Intellectual Property	
Company IP Agreements	Article 12
Company IT Systems	A ricle 12
Company Owned Software	
Company Plan	
Company Preferred Stock	
Company Takeover Proposal	A wiple 12
Company Transaction Expenses	Castion 2.2/h)
Contract	
D&O Indermifiable Matters	
Deferred Underwriting Amount	
DGCL	
Disclosure Schedule.	
Dissenting Shares	
DLLCA	
DOJ	
Encumbrance	x and x and x and x are x and x and x and x are x and x and x are x and x and x are x and x are x and x are x and x and x are x a
Enforcement Limitation.	
Environmental Claim.	
Environmental Law	
Environmental Permits	
ERISA	
ERISA Affiliate	
Estimated Purchase Price	
Estimated Purchase Price Calculation Statement	
Exchanged Amounts	
Expiration Time	
Faegre Drinker	Section 2.3
Federal Trade Commission Act	Article 12
Federal Trade Commission Act	
Final Purchase Price Calculation Statement. Financial Statements.	
Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger.	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1
Federal Trade Commission Act	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1
Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger.	Article 12Section 2.5(a)Section 3.4(a)(ii)Section 1.1Section 1.2
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger	Article 12Section 2.5(a)Section 3.4(a)(ii)Section 1.1Section 1.2Introduction
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger. First Effective Time	Article 12Section 2.5(a)Section 3.4(a)(ii)Section 1.1IntroductionSection 1.1
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger. First Effective Time. First Merger. First Surviving Company Founder Shares	
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger. First Effective Time. First Merger. First Surviving Company Founder Shares Founders Shares Agreement	
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount	
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount FTC	
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount FTC Fully Diluted Shares	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Section 6.1(b) Article 12
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount FTC Fully Diluted Shares GAAP	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Article 12 Article 12 Article 12 Article 12 Article 12 Article 12
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founder Shares Agreement Fractional Share Amount FTC Fully Diluted Shares GAAP Governmental Authority	
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount FTC Fully Diluted Shares GAAP Governmental Authority Hazardous Activity	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Section 1.5 Section 6.1(b) Article 12 Article 12 Section 3.12(b)
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount FTC Fully Diluted Shares GAAP Govenmental Authority Hazardous Substance	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Section 6.1(b) Article 12 Article 12 Article 12 Section 3.12(b) Article 12
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount. FTC Fully Diluted Shares GAAP Governmental Authority Hazardous Activity Hazardous Substance Holder	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Section 6.1(b) Article 12 Article 12 Article 12 Section 3.12(b) Article 12 Section 2.7(c)
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount FTC Fully Diluted Shares GAAP Governmental Authority Hazardous Activity Hazardous Substance Holder HSR Act	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Section 6.1(b) Article 12 Article 12 Article 12 Section 3.12(b) Article 12 Section 2.7(c) Article 12
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founder Shares Agreement Fractional Share Amount FTC Fully Diluted Shares GAAP Governmental Authority Hazardous Activity Hazardous Substance Holder HSR Act. Income Tax	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Section 6.1(b) Article 12 Article 12 Article 12 Section 3.12(b) Article 12 Section 2.7(c) Article 12
Federal Trade Commission Act Final Purchase Price Calculation Statement Financial Statements First Certificate of Merger First Effective Time First Merger First Surviving Company Founder Shares Founders Shares Agreement Fractional Share Amount FTC Fully Diluted Shares GAAP Governmental Authority Hazardous Activity Hazardous Substance Holder HSR Act	Article 12 Section 2.5(a) Section 3.4(a)(ii) Section 1.1 Section 1.2 Introduction Section 1.1 Article 12 Introduction Section 1.5 Section 6.1(b) Article 12 Article 12 Article 12 Section 3.12(b) Article 12 Section 2.7(c) Article 12

Independent Accounting Firm	Section 2.5(b)(ii)
Initial Expiration Date	Section 8 6(a)
Intellectual Property	Article 12
Interim Balance Sheet Date	Saction 3 4(a)(ii)
Interim Balance Sheets	Postion 3 4(a)(ii)
Interim Financial Statements.	
Interim Period	
IPO	
IPO Prospectus	Section 5.12
IPO Underwriters	Aricle 12
IRS	
Knowledge	Article 12
Leased Real Property	
Letter of Transmittal	
Liability	
Licensed Intellectual Property	Article 12
Lock-Up Period	Section 2.7(a)
Lock-Up Shares	Section 2.7(c)
Major Contract	
Material Adverse Effect	Article 12
Material Unregistered IP	
Merger Sub	Preamble
Merger Sub 2	
Mergers	
Multiemployer Plan	
Net Working Capital	Article 12
Net Working Capital Target	Promble
Nondisclosure Agreement.	Section 6,4(a)
Notice of Alternative Transaction	
Notice of Objection	Section 2.5(b)(1)
Offer to Purchase	
Offer Documents	Section 8.6(d)
Offering Shares	Article 12
Open Soure Code	Section 3.13(k)
Order	Article 12
Ordinary Commercial Agreement	Section 3.5(h)
Ordinary Course of Business	Article 12
Organizational Document.	
Outside Date	
Owned Real Property	
Party	
Permit	Article 12
Permitted Encumbrance	Article 12
Person	
Plan	
Platform Agreements	
Post-Closing Buyer Board	Article 12
Post-Closing Buyer Board	Article 12 Article 12
Post-Closing Buyer Board	Article 12 Article 12 Section 2.8(a)
Post-Closing Buyer Board	Article 12Article 12Section 2.8(a)Article 12

Dro formed Daymont	
Preferred Payment Private Warrant Amendment	Section 8 16(a)
Pro Rata Share	Section 2.5(c)(i)
Proceeding.	A rich 12
Public Certifications	Section 5 5(a)
Public Distributions	
Public Stockholders	
Public Warrant Amendment	Section 8 16(a)
Real Property	
Real Property Lease	
Redeemed Share	
Redemption	
Redemption Percentage	Article 17
Reference Time	Article 12
Registered IP	
Registration Rights Agreement	Section 9.7
Released Claims	Casties 11.1
Representative Losses	Section 11.10(b)
Representatives	A-4:-1- 12
Requisite Stockholders	Faction 9 (4)
Schedule TO.	Section 8.0(d)
SEC	
SEC Documents	
SEC Reports	
Second Certificate of Merger.	
Second Effective Time	
Second Merger	
Securities Act	
Shares	Article 12
Sherman Act.	
Signing Filing	Section 8.6(g)(i)
Signing Press Release.	
Software	
Sponsor Lock-up Agreements	
Stockholder	
Stockholder Litigation	Article 12
Stockholder Representative	
Stockholder Representative Reserve Fund	Section 11.16(c)
Stockholder Support Agreement	Article 12
Stockholders Agreement	
Straddle Period	
Subsidiary	
Surviving Company	
Tax	
Tax Return	
Tender Consideration	
Tender Offer	
Threatened	
Third Party Target.	
TO Commencement Date	
Transfer Transfer	
Transfer Taxes	Section 1.8

Trust Account	Article 12
Trust Agreement	
VWAP	
WARNAct	
Warrant Agreement	
Warrant Amendments	Section 8.16(a)
Warrant Holder Support Agreement	Section 8.16(b)
Warrantholder Consent Solicitation Statement	Section 8.16(c)
Wire Trans fer Instructions	Section 2.2(a)
Written Consent	Introduction

US.128937618.22

BUSINESS COMBINATION AGREEMENT

THIS BUSINESS COMBINATION AGREEMENT (this "Agreement") is entered into as of September 18, 2020, by and among Legacy Acquisition Corp., a Delaware corporation ("Buyer"), Excel Merger Sub 1, Inc., a Delaware corporation and an indirect wholly owned Subsidiary of Buyer and directly owned Subsidiary of Merger Sub 2 ("Merger Sub"), Excel Merger Sub II, LLC, a Delaware limited liability company and direct wholly owned Subsidiary of Buyer ("Merger Sub 2"), Onyx Enterprises Int'l Corp., a New Jersey corporation (the "Company"), and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as the Stockholder Representative pursuant to the terms of Section 11.16 of this Agreement. Certain capitalized terms used in this Agreement are defined in Article 12 of this Agreement. This Agreement constitutes a plan of merger for purposes of Section 14A:10-1 of the New Jersey Business Corporations Act (the "NJBCA").

INTRODUCTION

- A. The Company is generally engaged in the business of operating an e-commerce platform for the automotive parts and accessories retail market (the "Business").
- B. As an inducement to the Company to enter into this Agreement, concurrently with the consummation of the transactions contemplated by this Agreement, each holder of the Founder Shares will enter into a Sponsor Support Agreement, substantially in the form of Exhibit A hereto (the "Sponsor Support Agreement").
- C. The Company, Buyer and Merger Sub intend to effect a merger of Merger Sub with and into the Company (the "First Merger") in accordance with this Agreement, the NJBCA, and the DGCL and the Limited Liability Company Act of the State of Delaware (the "DLLCA"), where upon consummation of the First Merger, Merger Sub will cease to exist, and the Company will become a Subsidiary of Merger Sub 2. Immediately following the First Merger, the Company, as the surviving company of the First Merger, will merge with and into Merger Sub 2 (the "Second Merger") in accordance with this Agreement, the NJBCA, the DGCL, and the DLLCA, whereupon consummation of the Second Merger, Merger Sub 2 will be the surviving company and the Company will cease to exist. The First Merger and the Second Merger are together referred to in this Agreement as the "Mergers."
- D. The Board of Directors of the Company has (i) approved this Agreement and the transactions contemplated by this Agreement upon the terms and conditions set forth in this Agreement, (ii) determined that this Agreement and the transactions contemplated by this Agreement are fair to, and in the best interests of, the Company and the Stockholders, and (iii) resolved to recommend that the Stockholders approve this Agreement and the transactions contemplated by this Agreement.
- E. The respective boards of directors of each of Buyer, Merger Sub, and the Manager of Merger Sub 2 have approved this Agreement and the transactions contemplated by this Agreement, including the Merger, upon the terms and conditions set forth in this Agreement.
- F. The Parties intend that the Mergers (*i.e.*, the First Merger and Second Merger taken together) will qualify as a "reorganization" within the meaning of Section 368(a) of the Code and that this Agreement be, and is hereby adopted as, a "plan of reorganization" for purposes of Sections 354 and 361 of the Code.
- G. This Agreement will be adopted, and the transactions contemplated by this Agreement will be approved, by the written consent of the Requisite Stockholders (the "Written Consent") as promptly as practicable but not later than 11:59 pm (Eastern Time) on the day on which this Agreement is executed and delivered by all Parties.

AGREEMENT

Each Party agrees as follows:

ARTICLE 1 THE MERGERS

- 1.1 Merger; Surviving Company. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the applicable provisions of the DGCL and the NJBCA, at the First Effective Time, Merger Sub will merge with and into the Company. As a result of the First Merger, the Company will continue as the surviving company of the First Merger (the "First Surviving Company") and will continue its corporate existence under the laws of the state of New Jersey as a direct wholly owned Subsidiary of Merger Sub 2, and an indirect wholly owned Subsidiary of Buyer, and the separate corporate existence of Merger Sub will cease. Immediately thereafter, and as part of the same plan, upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the applicable provisions of the NJBCA and the DLLCA, at the Second Effective Time, the First Surviving Company shall be merged with and into Merger Sub 2. As a result of the Second Merger, Merger Sub 2 will continue as the surviving company of the Second Merger (the "Surviving Company") and will continue its existence under the laws of the state of Delaware as a wholly owned Subsidiary of Buyer, and the separate corporate existence of the First Surviving Company will cease.
- Effective Times. On the Closing Date, subject to the payment by Buyer of the consideration required to be paid by Buyer at the Closing pursuant to this Agreement, the Parties will cause the First Merger to be consummated by filing a certificate of merger, in a form mutually agreeable to Buyer and the Company (the "First Certificate of Merger"), to be properly executed and filed with the Secretary of State of the State of Delaware in accordance with the terms and conditions of the DGCL and with the Secretary of State of the State of New Jersey in accordance with the terms and conditions of the NJBCA and will take all such other actions as may be required by Applicable Laws to make the First Merger effective as promptly as practicable. The First Merger will become effective at the time that the First Certificate of Merger is accepted for filing by both the Secretary of State of the State of Delaware and the Secretary of State of the State of New Jersey, or at such later date and time as is specified in the First Certificate of Merger (such time and date being referred to herein as the "First Effective Time"). Immediately following the filing of the First Certificate of Merger, Merger Sub 2 shall cause a certificate of merger with respect to the Second Merger (the "Second Certificate of Merger" and, together with the First Certificate of Merger, the "Certificates of Merger") to be filed with the Secretary of State of the State of Delaware in accordance with the terms and conditions of the DLLCA and with the Secretary of State of the State of New Jersey in accordance with the terms and conditions of the NJBCA and will take all such other actions as may be required by Applicable Laws to make the Second Merger effective as promptly as practicable. The Second Merger will become effective at the time that the Second Certificate of Merger is accepted for filing by both the Secretary of State of the State of Delaware and the Secretary of State of the State of New Jersey, or at such later date and time as is specified in the Second Certificate of Merger (such time and date being referred to herein as the "Second Effective Time"). The First Effective Time shall precede in all events the Second Effective Time.

1.3 Effects of the Mergers.

(a) The Mergers will have the effects set forth in this Agreement, the Certificates of Merger and the applicable provisions of the NJBCA, DGCL and DLLCA. Without limiting the generality of the foregoing and subject thereto, by virtue of the First Merger and without further act or deed, at the First Effective Time, all the property, rights, privileges, immunities, powers, franchises, licenses and authority of the Company and Merger Sub will vest in the First Surviving Company, and all Liabilities and duties of the Company and Merger Sub will become the Liabilities and duties of the First Surviving Company. Also without limiting the generality of the foregoing and subject thereto, by virtue of the Second Merger and without further act or deed, at the Second Effective Time, all the property, rights, privileges, immunities, powers, franchises, licenses and authority of the First Surviving Company and Merger Sub 2 will vest in the

Business Combination Agreement Page 2

Surviving Company, and all Liabilities and duties of the First Surviving Company and Merger Sub 2 will become the Liabilities and duties of the Surviving Company.

- (b) At the First Effective Time, the certificate of incorporation of the First Surviving Company will, by virtue of the First Merger and pursuant to the First Certificate of Merger, be amended and restated to read as the certificate of incorporation of Merger Sub, as in effect immediately prior to the First Effective Time, except that at the First Effective Time, (i) "Article I" of such certificate of the First Surviving Company will be amended and restated in its entirety to read as follows: "The name of the corporation is PARTS iD, Inc.," and (ii) the limitations of liability and indemnification provisions will be revised to the extent necessary to comply with Section 8.2 of this Agreement. At the First Effective Time, the bylaws of the First Surviving Company will be amended and restated to read as the bylaws of Merger Sub, as in effect immediately prior to the First Effective Time, except that the Merger Sub's bylaws will be revised to the extent necessary to comply with Section 8.2 of this Agreement.
- (c) At the Second Effective Time, the certificate of formation and limited liability company agreement of Merger Sub 2 will, by virtue of the Second Merger and pursuant to the Second Certificate of Merger, remain in effect as the certificate of formation and limited liability company agreement of the Surviving Company, except that at the Second Effective Time, (i) "Article I" of such certificate of formation of the Surviving Company will be amended and restated in its entirety to read as follows: "The name of the limited liability company is "PARTS iD, LLC," and (ii) the limitations of liability and indemnification provisions will be revised to the extent necessary to comply with Section 8.2 of this Agreement.
- (d) Each of the Parties will take all necessary action to cause the directors of Merger Sub immediately prior to the First Effective Time to be the directors of the First Surviving Company immediately following the First Effective Time, to hold office until their respective successors are duly elected or appointed and qualified or their earlier death, resignation or removal in accordance with the certificate of incorporation and bylaws of the First Surviving Company. Each of the Parties will take all necessary action to cause the manager of Merger Sub 2 immediately prior to the Second Effective Time to become the manager of the Surviving Company as of the Second Effective Time.
- (e) The officers (i) of the Company immediately prior to the First Effective Time, who shall be the Persons named on **Schedule 1.3(e)**, will be the officers of the Surviving Company immediately following the First Effective Time and (ii) of Merger Sub 2 immediately prior to the Second Effective Time, who shall be the persons named on **Schedule 1.3(e)**, will be the officers of the Surviving Company immediately following the Second Effective Time, and such officers shall hold office until their respective successors are duly elected or appointed and qualified or their earlier death, resignation or removal in accordance with the certificate of formation and limited liability company agreement of the Surviving Company.
- (f) At the First Effective Time, by virtue of the First Merger and without any action on the part of any Party and subject to delivery by the relevant Stockholder of the Certificates or Book-Entry Shares representing such Stockholder's Shares, each duly endorsed in blank or accompanied by a separate executed stock power or similar document legally sufficient to assign the Shares to Buyer:
 - (i) each share of common stock, par value \$0.01 per share, of Merger Sub issued and outstanding immediately prior to the First Effective Time will be converted into and become one validly issued, fully paid and nonassessable share of common stock, par value \$0.01 per share, of the First Surviving Company;

- (ii) the Shares of Company Preferred Stock (other than Dissenting Shares and Cancelled Shares) shall be automatically converted into the right to receive an amount in cash equal to the Preferred Payment;
- (iii) each Share of Company Common Stock (other than Dissenting Shares and Cancelled Shares) shall be automatically converted into the right to receive (i) a number of shares of Buyer Common Stock equal to the Estimated Purchase Price divided by \$10.00 (in each case, the "Closing Share Consideration"), (ii) the per share amount of payments, if any, to Common Stockholders under Section 2.5(c), and (iii) the per share amount of any other release to Common Stockholders generally under this Agreement, in each case determined by dividing the aggregate amount of such released amounts by the number of Fully Diluted Shares (collectively, clauses (i) (iii), the "Common Share Merger Consideration"). For the avoidance of doubt, each component of the Common Share Merger Consideration will be payable only when, as and if, and in the forms contemplated by the applicable provisions of this Agreement; and
- (iv) each Share that is owned or held in treasury by the Company or that is owned by Buyer or any of its wholly owned Subsidiaries shall no longer be outstanding and shall automatically be cancelled and shall cease to exist (the "Cancelled Shares") and no consideration shall be delivered in exchange therefor.
- Common Share Merger Consideration pursuant to this Article 1 shall no longer be outstanding and upon the conversion thereof shall cease to exist as of the First Effective Time, and uncertificated Shares represented by book entry form ("Book-Entry Shares") and each certificate that, immediately prior to the First Effective Time represented any such Shares (each, a "Certificate") shall thereafter represent only the right to receive the Preferred Payment or the Common Share Merger Consideration into which the Shares represented by such Book-Entry Share or Certificate have been converted pursuant to this Section 1.3, as well as any dividends or other distributions to which Stockholders become entitled pursuant to Section 1.6.
- 1.4 Conversion of First Surviving Corporation Shares. At the Second Effective Time, by virtue of the Second Merger and without any action on the part of any of the Parties or holders of any securities of the First Surviving Company or of Merger Sub 2, (i) each membership interest of Merger Sub 2 issued and outstanding immediately prior to the Second Effective Time shall remain issued and outstanding as a membership interest of the Surviving Company and (ii) all shares of common stock of the First Surviving Company shall no longer be issued and outstanding and shall automatically be cancelled and shall cease to exist without any consideration being payable therefor.
- 1.5 <u>Fractional Shares</u>. No fraction of a share of Buyer Common Stock will be issued by virtue of the transactions contemplated by this Agreement. In lieu of any fractional share of Buyer Common Stock that otherwise would be issuable pursuant to this Agreement, each Stockholder who otherwise would be entitled to receive a fraction of a share of Buyer Common Stock (after aggregating all fractional shares of Buyer Common Stock that otherwise be received by that holder) will be paid an amount in cash determined by multiplying (a) \$10.00 by (b) the fraction of a share of Buyer Common Stock to which such Stockholder would otherwise be entitled on a payment date, which such product shall be rounded up to the nearest penny (the "Fractional Share Cash Amount"). No such holder shall be entitled to dividends, voting rights, or any other rights in respect of any fractional share of Buyer Common Stock. For the avoidance of doubt, the Fractional Share Cash Amount constitutes part of the Common Share Merger Consideration.

Page 4

1.6 Exchange of Shares.

- Certificates or Book-Entry Shares to Buyer, together with the documents specified in Section 1.3(f), duly completed and validly executed, the holder of such Certificates or Book-Entry Shares shall be entitled to receive in exchange therefor the Common Share Merger Consideration or Preferred Payment, as applicable. No interest shall be paid or shall accrue on any Fractional Share Cash Amount payable upon surrender of any Certificate or Book-Entry Share. Until surrendered as contemplated by this Section 1.6(a), each Certificate and Book-Entry Share shall be deemed at any time after the First Effective Time to represent only the right to receive, upon such surrender, the Common Share Merger Consideration or Preferred Payment, as applicable. Notwithstanding anything to the contrary in this Agreement, any holder of Book-Entry Shares shall not be required to deliver a Certificate; instead, the Book-Entry Shares of such holder shall be deemed surrendered and shall forthwith be cancelled.
- (b) Treatment of Unexchanged Shares. No dividends or other distributions, if any, with a record date after the First Effective Time with respect to Buyer Common Stock, shall be paid to the holder of any unsurrendered Share of Company Common Stock to be converted into the right to receive shares of Buyer Common Stock pursuant to Section 1.3 until such holder shall surrender such Shares in accordance with this Section 1.6. After the surrender in accordance with this Section 1.6 of a Share to be converted into the right to receive shares of Buyer Common Stock pursuant to Section 1.3, the holder thereof shall be entitled to receive (in addition to the Common Share Merger Consideration payable to such holder pursuant to this Section 1.6) any such dividends or other distributions, without any interest thereon, which theretofore had become payable with respect to the Buyer Common Stock issuable in respect of such Share of Company Common Stock.
- No Further Ownership Rights in Capital Stock. From and after the First Effective Time, (i) all holders of Certificates and Book-Entry Shares shall cease to have any rights as stockholders of the Company other than the right to receive the Preferred Payment or Common Share Merger Consideration into which the Shares represented by such Certificates or Book-Entry Shares have been converted pursuant to this Agreement upon the surrender of such Certificate or Book-Entry Share in accordance with Section 1.6(a) (together with any dividends or other distributions to which such Certificates or Book-Entry Shares become entitled in accordance with Section 1.6(b)), without interest, and (ii) the stock transfer books of the Company shall be closed with respect to all Shares outstanding immediately prior to the First Effective Time. From and after the First Effective Time, the stock transfer books of the Company shall remain closed, and there shall be no further registration of transfers on the stock transfer books of the Surviving Company of Shares that were outstanding immediately prior to the First Effective Time. If, at any time after the First Effective Time, any Certificates or Book-Entry Shares formerly representing Shares are presented to the Surviving Company or the Buyer for any reason, such Certificates or Book-Entry Shares shall be cancelled and exchanged as provided in this Article 1, subject to Applicable Laws in the case of Dissenting Shares.
- (d) Unclaimed Amounts. Any portion of the Common Share Merger Consideration or the Preferred Payment (including any interest or other amounts received with respect thereto) that remains unclaimed by, or otherwise undistributed to, the holders of Certificates and Book-Entry Shares as of the second anniversary of the Closing shall, to the extent permitted by Applicable Law, become the property of the Surviving Company and any holder of Certificates or Book-Entry Shares who has not theretofore complied with this Article 1 shall thereafter look only to Buyer or the Surviving Company (subject to abandoned property, escheat or similar Applicable Laws), as general creditors thereof, for satisfaction of its claim for Common Share Merger

Business Combination Agreement Page 5

Consideration or Preferred Payment which such holder has the right to receive pursuant to this **Article 1** without interest thereon.

(e) Lost Certificates. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed, in form and substance reasonably acceptable to Buyer, the Buyer shall deliver, in exchange for such lost, stolen or destroyed Certificate, the applicable Common Share Merger Consideration or Preferred Payment, as applicable, deliverable in respect thereof pursuant to this Agreement had such lost, stolen or destroyed Certificate been surrendered.

1.7 Notices to Stockholders; Dissenting Shares.

- (a) The Company shall, as promptly as reasonably practicable after the date hereof, but in any event no later than 15 days after the date of this Agreement, mail or deliver to each Stockholder a letter from the Company providing such Stockholder with (i) a brief information statement regarding the Company and the transactions contemplated by this Agreement, (ii) the notification required by Section 14A:5-6 of the NJBCA with respect to the Written Consent, and (iii) notice in the manner contemplated in Section 14A:11-1 of the NJBCA of such Stockholder's right to dissent to the First Merger pursuant Section 14A:11-1 of the NJBCA.
- (b) Notwithstanding any other provision of this Agreement to the contrary, any Shares held by a Stockholder who has neither voted in favor of the First Merger nor consented thereto in writing and who will have properly demanded appraisal for such Shares in accordance with Section 14A:11-1 of the NJBCA (collectively, the "Dissenting Shares") will not be converted into or represent the right to receive the Preferred Payment or Common Share Merger Consideration (or any portion thereof), in respect of such Dissenting Shares. Such Stockholder will instead be entitled to receive payment of the appraised value of such Dissenting Share in accordance with the provisions of Section 14A:11-1 of the NJBCA, except that any Dissenting Share held by a Stockholder who will have failed to perfect or who effectively will have withdrawn or otherwise lost his, her, or its rights to appraisal of such Shares under Section 14A:11-1 of the NJBCA will thereupon be deemed to have been converted into and to have become exchangeable, as of the First Effective Time, for the right to receive, in the manner provided in Section 1.6, the Preferred Payment or Common Share Merger Consideration, as applicable, without any interest thereon.
- (c) At the First Effective Time, any holder of Dissenting Shares will cease to have any rights with respect thereto, except the rights provided in Section 14A:11-1 of the NJBCA.
- (d) The Company shall provide Buyer prompt written notice of any demand received by it for appraisal of Shares, any withdrawal of any such demand and any other demand, notice or instrument delivered to the Company prior to the First Effective Time pursuant to the DGCL that relates to such demand, and Buyer shall have the opportunity and the right to direct all negotiations and Proceedings with respect to such demands. Except with the prior written consent of Buyer, the Company shall not make any payment with respect to, or settle or offer to settle, any such demands.
- 1.8 <u>Transfer Taxes</u>. All stamp, transfer, documentary, sales, use, registration and other such taxes, levies and fees (including any penalties and interest) incurred in connection with this Agreement and the Contemplated Transactions (collectively, the "Transfer Taxes") will be the responsibility of the Stockholders. The Stockholders will, at their own expense, procure any stock transfer stamps required by, and properly file on a timely basis all necessary Tax Returns and other documentation with respect to any Transfer Tax. No Transfer Taxes will be or become a Liability of Buyer or the Surviving Company.

Business Combination Agreement

Page 6

Each Transfer Tax Return will be prepared by the Party that customarily has primary responsibility for filing such Transfer Tax Return pursuant to Applicable Laws. Any Transfer Tax Returns prepared by the Company pursuant to this **Section 1.8** will be made available to Buyer at least ten Business Days before such Transfer Tax Returns are due to be filed.

ARTICLE 2 AGGREGATE PURCHASE PRICE; CLOSING; AND POST-CLOSING ADJUSTMENT

- 2.1 <u>Aggregate Purchase Price</u>. The term, "Aggregate Purchase Price," means an amount equal to the sum of:
 - (a) \$260,000,000,
 - (b) plus the amount, if any, by which the Closing Net Working Capital exceeds the Net Working Capital Target,
 - (c) **minus** the amount, if any, by which the Net Working Capital Target exceeds the Closing Net Working Capital,
 - (d) plus \$25,000,000, which represents Cash that will be retained by the Company,
 - (e) minus the amount of Indebtedness of the Company at the First Effective Time ("Closing Indebtedness"),
 - (f) minus \$20,000,000 paid to the holders of the outstanding Company Preferred Stock at the First Effective Time ("Preferred Payment"),
 - (g) minus the amount of all Company Transaction Expenses,
 - (h) minus \$3,000,000 (the "Adjustment Reserve Amount"),
 - (i) minus \$350,000 for the Stockholder Representative Reserve Fund, and
 - (j) minus, subject to Section 2.5(e)(i), \$7,500,000, which represents potential indemnification expenses that may become payable by the Company relating to the Stockholder Litigation (the "Indemnification Expense Reserve Amount").

2.2 Calculation of Estimated Purchase Price for Closing.

On or before the third Business Day before the Closing Date, the Company will deliver to Buyer a written statement (the "Estimated Purchase Price Calculation Statement") setting forth (i) the Company's good faith estimate of the amount of (A) the Closing Net Working Capital, (B) the Closing Indebtedness, (C) the Company Transaction Expenses, and (ii) the Company's resulting good faith estimate of the Aggregate Purchase Price based on the estimates in clause (i) and the other elements set forth in Section 2.1 (the "Estimated Purchase Price"), together with reasonably detailed documentation supporting such estimates in clauses (i) and (ii), and (iii) a certificate signed by an executive officer of the Company certifying that the estimated Closing Net Working Capital was determined in accordance with the terms of this Agreement. Schedule 2.2 sets forth the format for the calculation of the amounts described in this Section 2.2. Concurrently with delivery of the Estimated Purchase Price Calculation Statement. the Company will deliver to Buyer wire transfer instructions and other delivery information for the payments described in Section 2.4 (as may be amended from time to time thereafter, and which shall be orally confirmed, by the Company (if prior to Closing) or the Stockholder Representative (if after the Closing) prior to each payment hereunder, the "Wire Transfer Instructions"). Following the delivery of the Estimated Purchase Price Calculation Statement. if the Buyer has any objection to any amounts included in the Estimated Purchase Price Calculation Statement, Buyer and the Company shall reasonably cooperate in good faith to resolve such objections. If the Company and the Buyer are unable to resolve any dispute with respect to the Estimated Purchase Price Calculation Statement, such dispute shall not under any circums tance

Business Combination Agreement Page 7

delay the Closing, and the amounts set forth in the Estimated Purchase Price Calculation Statement shall be the amounts used for the purposes of the Closing.

- (b) Not later than one Business Day before the Closing Date, Buyer shall deliver to the Company a statement certified by an executive officer of Buyer setting forth the Redemption Percentage.
- (c) Not later than the third Business Day prior to the Closing Date, the Company will deliver to Buyer a schedule (the "Allocation Schedule") setting forth (i) a list of the Stockholders as of the First Effective Time and, for each Stockholder, the number of Shares owned by that Stockholder and (ii) a calculation of all amounts and shares of Buyer Common Stock payable at the Closing to each Stockholder.

Each of the Company, Stockholder Representative (on behalf of the Stockholders), Buyer and Merger Sub acknowledge and agree that each of them will have the right to rely on the Allocation Schedule as setting forth a true, complete and accurate listing of all amounts due to be paid by Buyer, Merger Sub and the Company at Closing. Buyer, Merger Sub and the Surviving Company will not have any Liability with respect to the allocation of proceeds among the Stockholders resulting from any payments made to such Stockholders pursuant to the Allocation Schedule. The Stockholders, by approving this Agreement, agree that all Stockholders will cooperate with the Stockholder Representative, Buyer, the Surviving Company and the other Stockholders in order to ensure that each Stockholder receives its respective portion of any Preferred Payment or Common Share Merger Consideration that it is entitled to receive pursuant to the terms of this Agreement.

2.3 <u>Closing</u>. Subject to the fulfillment or waiver of the conditions precedent set forth in Article 8, the closing of the transactions contemplated by this Agreement (the "Closing") will take place at the offices of Faegre Drinker Biddle & Reath LLP ("Faegre Drinker"), in Florham Park, New Jersey, at 9:00 a.m. (local time) on a date to be mutually agreed upon by the Company and Buyer, which date will be no later than the third Business Day after all of the conditions set forth in Article 8 have been satisfied or waived (other than those conditions which by their terms are intended to be satisfied at Closing). The date on which the Closing occurs is referred to herein as the "Closing Date." Documents may be delivered at the Closing by electronic delivery, and the receiving Party may rely on the receipt of such documents so delivered as if the original had been received.

2.4 <u>Closing Payments</u>. At Closing, Buyer will:

- (a) pay and discharge (or cause to be paid and discharged), on behalf of the Company all Closing Indebtedness and Company Transaction Expenses set forth on the Estimated Purchase Price Calculation Statement, and the amount of the Stockholder Representative Reserve Fund, by wire transfer of immediately available funds pursuant to the Wire Transfer Instructions;
- (b) hold in reserve that number of shares of Buyer Common Stock equal to the Adjustment Reserve Amount divided by \$10.00, to be held by the Buyer pursuant to the terms of this Agreement;
- (c) subject to Section 2.5(e)(i), hold in reserve that number of shares of Buyer Common Stock equal to the Indemnification Expense Reserve Amount divided by \$10.00, to be held by the Buyer pursuant to the terms of this Agreement; and
- (d) pay to an account for the holder of the Company Preferred Stock as designated in writing by the Company, an amount equal to the Preferred Payment by wire transfer of immediately available funds, in accordance with **Article 1**; and
- (e) deliver to each Common Stockholder its Closing Share Consideration in shares of Buyer Common Stock in book-entry form, in accordance with **Article 1**.

Page 8

2.5 Purchase Price Adjustment.

(a) Buyer's Preparation of the Statement. Within 90 days after the Closing Date, Buyer will prepare and deliver to each of the Stockholder Representative and the Buyer Sponsor a statement (the "Final Purchase Price Calculation Statement") setting forth, in reasonable detail, Buyer's calculations ("Buyer's Proposed Calculations") of (i) the amount of the Closing Net Working Capital, (ii) Closing Indebtedness, (iii) Company Transaction Expenses, and (iv) the resulting Aggregate Purchase Price based on the foregoing and the other elements set forth in Section 2.1.

(b) Objections; Resolution of Disputes.

- (i) Unless one or both of the Stockholder Representative and the Buyer Sponsor notifies Buyer in writing within 45 days after Buyer's delivery of the Final Purchase Price Calculation Statement of any objection to Buyer's Proposed Calculations (the "Notice of Objection"), the Final Purchase Price Calculation Statement will become final and binding on the Parties at the end of such 45-day period. During such 45-day period, Stockholder Representative, the Buyer Sponsor and their respective representatives and advisors will be permitted to review all materials and information used by Buyer in preparing the Final Purchase Price Calculation Statement and Buyer will make available such personnel as are reasonably necessary to assist Stockholder Representative or the Buyer Sponsor in its respective review of the Final Purchase Price Calculation Statement. Any Notice of Objection must specify in reasonable detail the basis for the objections set forth therein.
- If either or both of Stockholder Representative and the Buyer Sponsor (an "Objecting Party") provides a Notice of Objection to Buyer and each other within such 45-day period, Buyer and the Objecting Party (or the Objecting Parties, if both the Stockholder Representative and the Buyer Sponsor deliver a Notice of Objection or elect to participate after the other party delivers a Notice of Objection) will, during the 30-day period following Buyer's receipt of the Notice of Objection, attempt in good faith to resolve the objections of Stockholders. During such 30-day period, Buyer and its independent auditors and other representatives and their advisors will be permitted to review the working papers of the Objecting Party and, if applicable, their representatives relating to the Notice of Objection and the basis therefor. If Buyer and the Objecting Party are unable to resolve all such objections within such 30-day period, the matters remaining in dispute that were properly included in the Notice of Objection will be submitted to RSM US LLP (or, if such firm declines to act, to another nationally recognized public accounting firm mutually agreed upon by Buyer and the Objecting Party in writing and, if Buyer and the Objecting Party are unable to so agree within 10 days after the end of such 30-day period, then Buyer and the Objecting Party will each select such a firm and such firms will jointly select a third nationally recognized firm to resolve the disputed matters (such selected firm being the "Independent Accounting Firm")). Buyer and the Objecting Party will instruct the Independent Accounting Firm to render its decision as promptly as practicable but in no event later than 30 days after its selection. Buyer and the Objecting Party will further instruct the Independent Accounting Firm to deliver a written report containing its calculation of the disputed items and a written explanation in reasonable detail of each required adjustment to the disputed items, including the basis for those adjustments. The Independent Accounting Firm will act as an expert and not an arbitrator to resolve the disputed matters solely based on submissions made on behalf of Stockholders and Buyer that are consistent with the terms of this Agreement (and not by independent review or additional discovery in any form). The Independent Accounting Firm will not assign a value to any item that is greater than the greater value for such item claimed by either Party or less than the lesser value for

Business Combination Agreement Page 9

such item claimed by either Party. The resolution of disputed items by the Independent Accounting Firm will be final and binding, and the determination of the Independent Accounting Firm may be enforced as if it were an arbitral award that is final, binding and non-appealable and upon which a judgment may be entered by a court having jurisdiction over the Party against which such determination is to be enforced, absent manifest error. Any Party may seek specific enforcement or take other necessary legal action to enforce any decision made by the Independent Accounting Firm. The other Party's only defense to such a request for specific enforcement or other legal action shall be manifest error or fraud by or upon the Independent Accounting Firm. The fees and expenses of the Independent Accounting Firm will be apportioned between Buyer, on one hand, and the Objecting Party, on the other hand, based upon the inverse proportion of the amount of the disputed items on the Final Purchase Price Calculation Statement resolved in favor of each of Buyer and such Person (i.e., so that the prevailing Party bears a lesser amount of such fees and expenses), with any expenses that are the responsibility of the Stockholders being paid out of the shares of Buyer Common Stock representing the Adjustment Reserve Amount. The Final Purchase Price Calculation Statement, as adjusted pursuant to this Section 2.5(b)(ii), will be final and binding on the Parties. All information provided by the Parties to each other and the Independent Accounting Firm under this Section 2.5 will be held in confidence and will not be further disclosed by the Parties and the Independent Accounting Firm.

- (c) Adjustment Payment. Upon determination, in accordance with Section 2.5(b), of the Final Purchase Price Calculation Statement and the final calculations of the items to be included therein under Section 2.5(a), the Aggregate Purchase Price will be recalculated using such finally determined amounts. The following payments, if any, will be deemed to be adjustments to the Aggregate Purchase Price for all purposes.
 - (i) If the Aggregate Purchase Price (as determined under this Section 2.5) exceeds the Estimated Purchase Price by more than \$1,500,000, then Buyer will deliver or cause to be delivered, no later than three Business Days after the Final Purchase Price Calculation Statement becomes final and binding under Section 2.5(b), to each Common Stockholder, in book-entry form shares of Buyer Common Stock to each Common Stockholder equal to:
 - (A) (1) the Adjustment Reserve Amount; multiplied by (2) a fraction equal to (x) the number of Shares of Company Common Stock held by that Common Stockholder immediately prior to the First Effective Time divided by (y) the Fully Diluted Shares (each being that Common Stockholder's "Pro Rata Share"); divided by (3) \$10.00; plus
 - (B) (1) the lesser of (x) the amount of such excess (above the \$1,500,000 threshold) or (y) \$3,000,000; multiplied by (2) that Common Stockholder's Pro Rata Share; divided by (3) \$10.00.
 - (ii) If the Estimated Purchase Price exceeds the Aggregate Purchase Price (as determined under this Section 2.5) by more than \$1,500,000, then no later than three Business Days after the Final Purchase Price Calculation Statement becomes final and binding under Section 2.5(b), Stockholder Representative and Buyer Sponsor will provide joint instructions to the Buyer to release from Buyer's reserve established pursuant to Section 2.4(b) a number of shares of Buyer Common Stock equal to the lesser of (1) the amount of such excess (above the \$1,500,000 threshold) divided by \$10.00, and (2) the number of shares of Buyer Common Stock representing the Adjustment Reserve Amount. The Adjustment Reserve Amount is Buyer's sole and exclusive source of recovery of any amount by which the Estimated Purchase Price

Business Combination Agreement Page 10

exceeds the Aggregate Purchase Price by more than \$1,500,000. Such instructions to the Buyer will also provide that any shares of Buyer Common Stock representing the balance of the Adjustment Reserve Amount after the release of reserved shares of Buyer Common Stock under the previous sentence will be delivered in book-entry form to the Common Stockholders, with an amount of shares of Buyer Common Stock so delivered to each Common Stockholder equal to that Common Stockholder's Pro Rata Share of such remaining amount. Each payment to a Common Stockholder will be in accordance with the instructions provided by that Common Stockholder to the Buyer.

- (iii) If the Aggregate Purchase Price (as determined under this Section 2.5) is not more than \$1,500,000 more or less than the Estimated Purchase Price, then no later than three Business Days after the Final Purchase Price Calculation Statement becomes final and binding under Section 2.5(b), Stockholder Representative and Buyer Sponsor will provide joint instructions to the Buyer to release the shares of Buyer Common Stock representing the Adjustment Reserve Amount and deliver in book-entry form such number of shares of Buyer Common Stock to each Common Stockholder equal to that Common Stockholder's Pro Rata Share of such remaining amount. Each payment to a Common Stockholder will be in accordance with the instructions provided by that Common Stockholder to the Buyer.
- (d) Adjustment Reserve Amount. The number of shares of Buyer Common Stock representing the Adjustment Reserve Amount will be held in reserve by Buyer pursuant to Section 2.4(b).

(e) Indemnification Expense Reserve Amount.

- (i) The Indemnification Expense Reserve Amount will be \$0 if (A) the Stockholder Litigation is finally settled or resolved by a court prior to Closing, and (B) the final amount of the D&O Indemnifiable Matters has either been paid by the Company prior to Closing or is included in the calculation of the Estimated Purchase Price, whether as Net Working Capital, Indebtedness, or Company Transaction Expenses.
- The number of shares of Buyer Common Stock representing the Indemnification Expense Reserve Amount will be held in reserve by Buyer pursuant to Section 2.4(c) and this Section 2.5(e). The Indemnification Expense Reserve Amount will be reduced by any amount required (by final settlement or court resolution) to be indemnified by the Company after the Closing Date for any acts or omissions of the Company's current or former officers, directors, governors, members, partners, or managers in connection with the Stockholder Litigation (the "D&O Indemnifiable Matters"), net of any amounts recovered by the Company under any insurance policy (including any directors' and officers' liability insurance) or other indemnity, contribution or similar payment received with respect to such matter. The Company will use reasonable best efforts to collect under all available insurance policies or indemnity, contribution or other similar agreements. For the avoidance of doubt, the right of offset against the Indemnification Expense Reserve Amount shall be the sole and exclusive remedy of the Company and its Affiliates for any D&O Indemnifiable Matters and the Indemnification Expense Reserve Amount shall not be available to the Company for any Liability other than a D&O Indemnifiable Matter.
- (iii) The Company will promptly deliver written notice to the Stockholder Representative and to the Buyer Sponsor of any claim made against the Company with respect to a D&O Indemnifiable Matter. Such written notice will specify in reasonable detail the amount of the claim, the basis therefor, and a summary of rights that are reasonably available to the Company under any insurance policy, indemnity, contribution or similar agreement with respect thereto. The Company will provide the Stockholder Representative and the Buyer Sponsor with any other information reasonably requested

by the Stockholder Representative or by the Buyer Sponsor with respect to any such claim. Any disputes regarding D&O Indemnifiable Matters or any purported offset against the Indemnification Expense Reserve Amount shall be resolved in accordance with **Article 11**.

- (iv) On the earlier to occur of: (A) the final settlement or court resolution of the Stockholder Litigation or (B) the second anniversary of the Closing Date, Buyer will release to each Common Stockholder a number of shares of Buyer Common Stock equal to (x) the remaining Indemnification Expense Reserve Amount, if any, multiplied by (y) that Common Stockholder's Pro Rata Share, divided by (z) \$10.00.
- (v) If, within the five years following the second anniversary of the Closing Date, the Company receives any amounts under any insurance policy, indemnity, contribution or other similar agreement for any D&O Indemnifiable Matters that were previously deducted from the Indemnification Expense Reserve Amount, then the Company will promptly distribute those funds to the Common Stockholders in accordance with their respective Pro Rata Shares.
- 2.6 Withholding. Notwithstanding any other provision of this Agreement, and for the avoidance of doubt, (a) the payor of each payment made pursuant to this Agreement shall be entitled to make such payment net of any Taxes required by Applicable Law to be deducted or withheld from such payment and (b) any amounts deducted or withheld from any such payment shall be timely remitted to the applicable Taxing authority and shall be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding was made. Notwithstanding the foregoing there shall not be any withholding on account of U.S. federal income Taxes on a payment to a Stockholder in respect of the Preferred Payment or in respect of its shares of Common Stock of the Common Share Merger Consideration if the Stockholder timely provides a valid duly executed IRS Form W-9, W-8BEN or W-8BEN-E claiming a complete exemption from backup withholding prior to Closing.

2.7 Lock-Up Period; Stop-Transfer Notices.

- (a) Lock-Up Agreement. Each Common Stockholder agrees that such Common Stockholder shall not Transfer any Buyer Common Stock issued to such Common Stockholder as part of the Closing Share Consideration that may have been issued to such Common Stockholder prior to the earlier of (i) the first anniversary of the Closing Date, (ii) the date, following the 180th day after the Closing Date, on which the VWAP of Buyer Common Stock equals or exceeds \$15.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like), (iii) the date, following the 270th day after the Closing Date, on which the VWAP of Buyer Common Stock equals or exceeds \$13.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like), or (iv) Buyer's completion of a liquidation, merger, stock exchange or other similar transaction that results in all of Buyer's stockholders having the right to exchange their shares of Buyer Common Stock for cash, securities or other property (the "Lock-up Period").
- (b) **Permitted Transfers.** Notwithstanding the foregoing, during the Lock-up Period, Transfers of Closing Share Consideration are permitted to be made (i) in the case of an entity, as a distribution to its partners, stockholders or members; (ii) in the case of an individual, by a bona fide gift to one of the members of the individual's immediate family or to a trust, the beneficiary of which is a member of the individual's immediate family, for estate planning purposes; (iii) in the case of an individual, by virtue of laws of descent and distribution upon death of the individual; (iv) in the case of an individual, pursuant to a qualified domestic relations order; (v) if consented to in advance by the Post-Closing Buyer Board; **provided, however**, that in each case of clauses (i) through (iv), the permitted transferees must enter into a written agreement agreeing to be bound by these transfer restrictions and the other terms described in

Business Combination Agreement

Page 12

this **Section 2.7** to the extent and for the duration that such terms remain in effect at the time of the Transfer.

- Stop-Transfer Notice. Each holder of Buyer Common Stock, inclusive of any Person to whom shares of Buyer Common Stock have been Transferred by a Common Stockholder in accordance with Section 2.7(b)) (each, a "Holder") of Buyer Common Stock issued pursuant to this Agreement hereby authorizes Buyer during each applicable Lock-Up Period to cause its transfer agent for the shares of Buyer Common Stock that are subject to restriction on Transfer during the applicable Lock-Up Period ("Lock-up Shares") to decline to transfer, and to note stop transfer restrictions on the stock register and other records relating to, such Lock-up Shares for which the undersigned is the record holder and, in the case of Lockup Shares for which the Holder is the beneficial holder but not the record holder, such Holder agrees during the applicable Lock-Up Period to cause the record holder to cause the relevant transfer agent to decline to transfer, and to note stop transfer restrictions on the stock register and other records relating to, such Lock-up Shares, if such Transfer would constitute a violation or breach of this Agreement. In addition, any Holder who is not a resident of the United States and is not an accredited investor as defined under Rule 501 of Regulation D under the Securities Act hereby authorizes Buyer to instruct its transfer agent to note that such Transaction Securities were issued under an exemption from registration under Regulation S and to utilize the CREST electronic clearing system for the purpose of compliance with legend and transfer restriction requirements as provided for by Regulation S. In addition, each Holder agrees that Buyer may issue appropriate stop transfer instructions to its transfer agent in order to ensure compliance with the Securities Act, including as to Regulation S as applicable.
- (d) Releases from Lock-Ups. Buyer will not grant an early release or waiver of the restrictions imposed by any lock-up agreement upon the Buyer Sponsor or the Common Stockholders during the Lock-Up Period unless a similar release or waiver, on a pro rata basis, is provided to the Common Stockholders or the Buyer Sponsor, respectively, from any such Person's lock-up agreement. Except as provided in this Section 2.7, Buyer shall not release, amend or waive the Sponsor Lock-Up Agreement or any lock-up agreement of any Common Stockholder and shall enforce any violations thereof
- 2.8 Equity Incentive Plan. Not less than five Business Days prior to the filing with the SEC of the Schedule TO, the board of directors of Buyer shall adopt, subject to approval as part of the Required Buyer Stockholder Approval, an equity incentive plan, in form and substance reasonably acceptable to the Company, that provides for the issuance of restricted stock units and other forms of equity incentive awards by Buyer subject to an aggregate cap on such awards equal to 12% of the number of shares of Buyer Common Stock on a fully-diluted basis (the "Post-Closing Equity Incentive Plan").

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as disclosed in the disclosure schedule delivered by the Company concurrently with the execution and delivery of this Agreement (the "Disclosure Schedule"), the Company hereby represents and warrants to Buyer, Merger Sub and Merger Sub 2 as follows as of the date of this Agreement:

3.1 Organization and Good Standing. The Company is a duly organized and validly existing corporation in good standing under the laws of the jurisdiction in which it was organized. The Company is duly qualified and in good standing to do business as a foreign corporation in each jurisdiction in which the ownership or leasing of its properties or assets or the conduct of its business requires such qualification, except where the failure to be so qualified or in good standing has not had a Material Adverse Effect and will not materially and adversely affect the ability of the Company or the Stockholders to consummate the transactions contemplated herein. The Company has full corporate power and authority to own and lease its properties and assets and conduct its business as now conducted. The Company does not hold any equity interest, directly or indirectly, of any other Person. The Company

Page 13

has made available to Buyer a true, correct and complete copy of the Organizational Documents of the Company.

Shares owned by each Stockholder. Each Stockholder has good and valid title to all Shares owned by it, free and clear of any Encumbrance (other than restrictions imposed by securities laws applicable to securities generally and rights of Buyer under this Agreement). The Shares constitute all of the outstanding equity interests of the Company. All of the equity interests listed in Section 3.2 of the Disclosure Schedule are, or will be when issued, duly authorized, validly issued, and, if capital stock, fully paid and non-assessable. No equity interest of the Company was issued in violation of any Organizational Document of the Company, any Applicable Law or any preemptive right (or other similar right) of any Person. There is no: (i) preemptive right, option, warrant, put, call, purchase right, subscription right, conversion right, convertible instrument, exchange right or other security, Contract or commitment of any nature whereby any Person has, or has a right to receive, any equity interest of, or right or obligation to acquire any equity interest of, the Company; (ii) equity appreciation, phantom stock, profit participation or similar right with respect to the Company; or (iii) voting trust, proxy or other Contract with respect to any equity interest of the Company. The Company does not have any Subsidiaries or own any equity interests in any other Person.

3.3 Authority and Authorization; Conflicts; Consents.

- (a) Authority and Authorization. The Company has all requisite corporate power to execute and deliver this Agreement and each Ancillary Document to which the Company is a party and the execution, delivery and performance of this Agreement and each Ancillary Document of the Company have been duly authorized and approved by the Board of Directors of the Company and, upon execution of the Written Consent, no other corporate or stockholder action on the part of the Company or the Stockholders is necessary to authorize such execution, delivery and performance. Assuming due authorization, execution and delivery by Buyer of this Agreement and each Ancillary Document of Buyer, this Agreement is, and each Ancillary Document of the Company will be, the legal, valid and binding obligation of the Company enforceable against the Company in accordance with its terms, except to the extent enforceability may be limited by any Enforcement Limitation.
- Conflicts: Consents. Subject to the receipt of the Consents and the making of filings and submissions referenced in the next sentence, neither the execution nor delivery by the Company of this Agreement or of any Ancillary Document nor consummation by Company of the transactions contemplated herein or therein does or will (with or without the passage of time or giving of notice): (i) constitute a breach of, violate, conflict with or give rise to or create any right or obligation under any Organizational Document of the Company; (ii) violate any Applicable Law; (iii) constitute a breach or violation of or a default under, conflict with or give rise to or create any right of any Person other than the Company to accelerate, increase, terminate, modify or cancel any material right or obligation in a manner adverse to the Company or result in the creation of any Encumbrance, other than a Permitted Encumbrance, of the Company under any Major Contract; or (iv) result in the creation or imposition of any Encumbrance on the Shares, except where such a breach, violation, default, conflict or right under clause (ii) or (iii) above has not had a Material Adverse Effect and will not materially and adversely affect the ability of the Company or the Stockholders to consummate the transactions contemplated herein. No consent or approval by, notification to or filing with any Person is required under any Major Contract in connection with the Company's execution, delivery or performance of this Agreement or any Ancillary Document of the Company or the consummation of the transactions contemplated herein or therein by the Company or the Stockholders, except (1) the filing of the Certificates of Merger, (2) as required by Antitrust Laws, and (3) for any consent, approval, notice or filing, the absence of which has not had a Material Adverse Effect and will not materially and adversely affect the ability of the Company or the Stockholders to consummate the

Business Combination Agreement Page 14

transactions contemplated herein. "Consent' means each consent, approval, notice or filing listed in Section 3.3(b) of the Disclosure Schedule.

3.4 <u>Financial Statements and Undisclosed Liabilities.</u>

- (a) Financial Statements Defined. Section 3.4(a) of the Disclosure Schedule contains a true, correct and complete copy of the following:
 - (i) the Company's consolidated audited balance sheets as of, and audited income statements for each of the fiscal years ended on, December 31, 2019 (the "Annual Balance Sheet Date"), December 31, 2018, and December 31, 2017, together with the notes to the foregoing and the reports thereon of the Company's independent auditors; and
 - (ii) the unaudited (1) balance sheet of the Company, on a consolidated basis, as of June 30, 2020 (such date is the "Interim Balance Sheet Date" and such balance sheets are the "Interim Balance Sheets"); and (2) income statement of the Company, on a consolidated basis, for the 6-month period ended on the Interim Balance Sheet Date (collectively, the "Interim Financial Statements" and, together with the annual financial statements described in Section 3.4(a)(i), the "Financial Statements").
- (b) Financial Statements. The Financial Statements were prepared in accordance with GAAP and fairly present, in all material respects, the consolidated financial condition of the Company at their respective dates and the consolidated results of operations of the Company for the respective periods covered thereby, except (i) as otherwise disclosed in any note to the Financial Statements and (ii) that the Interim Financial Statements are subject to normal year-end adjustments and do not include notes.
- (c) Undisclosed Liabilities. The Company does not have any Liabilities, except for Liabilities (i) disclosed in, provided for, adequately reflected in, reserved against or otherwise described in the Financial Statements (including in any note thereto) or that are or will be included in the calculations regarding the determination of the Aggregate Purchase Price under Section 2.5, (ii) that have arisen in the Ordinary Course of Business of the Company since the Interim Balance Sheet Date, (iii) under this Agreement or any Ancillary Document or otherwise in connection with the transactions contemplated herein or therein, (iv) disclosed in or contemplated by this Agreement, including the Disclosure Schedule, or (v) that would not have been required to be provided for, reflected in, or reserved against in the Financial Statements (including in any notes thereto) in accordance with GAAP.
- (d) Internal Controls. The Company maintains a system of internal accounting controls with respect to its business that are designed to provide reasonable assurances regarding the reliability of financial reporting and the preparation of the consolidated financial statements in accordance with GAAP. In connection with the routine audit practices of the Company and its independent auditors, the Company has not identified for the Company's independent auditors any material weaknesses in internal controls nor to the Company's Knowledge, has there been any incident of fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal controls over financial reporting. To the Company's Knowledge, there are no significant deficiencies in the Company's internal controls which could adversely affect the ability of the Company to record, process, summarize and report financial data.

3.5 Taxes.

(a) All material Tax Returns that were required to be filed by, on behalf of or with respect to the Company were timely filed on or before the applicable due date (taking into account any extension of such due date) and were correct and complete in all material respects. All Taxes owed by or with respect to the Company with respect to any Pre-Closing Tax Period

were paid, or, in the case of Taxes not yet due, will be paid in full before the First Effective Time or will be properly included in Closing Net Working Capital.

- (b) The Company has not received a written claim since January 1, 2018 from any Governmental Authority in a jurisdiction where the Company does not file any Tax Return that the Company is or may be subject to Taxation by such jurisdiction.
- (c) All material Taxes that the Company is or was required to withhold or collect in connection with any amount paid or owing to any employee, independent contractor, stockholder, nonresident, creditor or other third party have been duly withheld or collected and have been paid, to the extent required, to the proper Governmental Authority or other Person.
- (d) No Tax audit or other Tax Proceeding is pending or being conducted or is, to the Company's Knowledge, Threatened against or otherwise with respect to the Company.
- (e) The Company does not have in effect any waiver of any statute of limitations regarding any Tax nor has it agreed to any extension of time regarding the assessment of any Tax deficiency.
- (f) There is no Encumbrance for any Tax upon any asset of the Company, except any Encumbrance for any current Tax not yet due.
- (g) The Company is not aware of the existence of any fact, or has taken or agreed to take any action, that would reasonably be expected to prevent or impede the transactions contemplated by this Agreement from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.
- (h) The Company is not a party to, not bound by or has no obligation under any Tax sharing Contract, Tax indemnification Contract, Tax allocation Contract or similar Contract (including any Contract providing for the sharing or ceding of Tax credits or Tax losses) and does not have a Liability to any Person as a result of or pursuant to any such Contract, in each case other than a Contract the primary purpose of which does not relate to Taxes (an "Ordinary Commercial Agreement").
- (i) The Company will not be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any Tax period (or portion thereof) ending after the Closing Date as a result of any (i) change in method of accounting made prior the Closing under Code Section 481(c) (or any corresponding or similar provision of state, local or non-U.S. income Tax law); (ii) "closing agreement" as described in Code Section 7121 (or any corresponding or similar provision of state, local or non-U.S. income Tax law) executed prior to the Closing; (iii) installment sale or open transaction disposition made prior to the Closing; (iv) intercompany transaction or any excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding or similar provision of state, local or non-U.S. Tax law) entered into or created prior to the Closing; or (v) prepaid amount received prior to the Closing outside the Ordinary Course of Business.
- (j) The Company has not been a member of an affiliated group filing a consolidated, combined or unitary U.S. federal, state, local or non-U.S. income Tax Return (other than a group of which the Company was the common parent).
- (k) The Company does not have any material Liability for the Taxes of any Person (other than the Company) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or non-U.S. law), as a transferee or successor, or, except pursuant to an Ordinary Commercial Agreement, by Contract or otherwise.
- (I) The Company does not have any request for a material closing agreement, private letter ruling or similar ruling in respect of Taxes pending between the Company, on the one hand, and any Tax authority, on the other hand.

- (m) The Company has made available to Buyer true, correct and complete copies of the U.S. federal income Tax Returns filed by the Company for tax years since 2017.
- (n) The Company has not in any year for which the applicable statute of limitations remains open distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 or Section 361 of the Code.
- (o) The Company has not engaged in or entered into a "listed transaction" within the meaning of Treasury Regulation Section 1,6011-4(b)(2).
- (p) Neither the IRS nor any other U.S. or non-U.S. taxing authority or agency has asserted in writing against the Company any deficiency or claim for any material Taxes or interest thereon or penalties in connection therewith.
- (q) The Company has not been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code. The Company has not received written notice from a non-United States Tax authority that it has a permanent establishment (within the meaning of an applicable Tax treaty) or otherwise has an office or fixed place of business in a country other than the country in which it is organized.
- (r) For U.S. federal income tax purposes, the Company is, and has been since its formation, classified as a corporation.
- 3.6 <u>Litigation and Orders</u>. There is no Proceeding pending or, to the Company's Knowledge, Threatened against the Company as of the date of this Agreement (other than workers' compensation claims), in each case that, if decided adversely to the Company, will have a Material Adverse Effect or will materially and adversely affect the ability of the Company or the Stockholders to consummate the transactions contemplated herein. As of the date of this Agreement, the Company is not specifically identified as a party that is subject to any material restriction or limitation on any of its operations under any Order.
- 3.7 <u>Compliance with Law; Permits.</u> The Company is, and has been since January 1, 2018, in compliance with all Applicable Laws, except where the failure to be in compliance would not have a Material Adverse Effect. No written notice has been received by the Company from January 1, 2018 to the date of this Agreement from any Governmental Authority alleging that the Company is not or was not in compliance with any Applicable Law that has not been remedied or where such non-compliance would not have a Material Adverse Effect. The Company has in effect all Permits that are required for the Company to conduct the Business as currently conducted in all material respects, all such Permits are valid and in full force and effect, and the Company is in compliance in all material respects with the terms of such Permits.

3.8 Contracts.

- (a) Section 3.8(a) of the Disclosure Schedule lists the following Contracts, in effect as of the date of this Agreement, to which the Company is a party (each a "Major Contract"):
 - (i) each Contract providing for the payment of any cash or other compensation or benefits upon the sale of all or a portion of the Company or the business of the Company;
 - (ii) each Contract for the employment of an individual or engagement of a consultant, in each case earning an annual base salary or annual consulting fee of \$200,000 or more per year (other than offer letters entered into in the Ordinary Course of Business for employees or agreements with consultants where such relationship is terminable at-will);

- (iii) collective bargaining agreements;
- (iv) each Contract that is reasonably expected to involve payments to the Company of more than \$100,000 in the 12 months following the Closing Date (other than those Contracts that on the Closing Date will be terminable at will or upon not more than 90 days' notice by the Company without any Liability to the Company, other than Liability with respect to services rendered before the termination thereof that is unrelated to such termination);
- (v) each Contract that is reasonably expected to involve payments by the Company of more than \$100,000 in the 12 months following the Closing Date (other than those Contracts that on the Closing Date will be terminable at will or upon not more than 90 days' notice by the Company without any Liability to the Company, other than Liability with respect to services rendered before the termination thereof that is unrelated to such termination):
- (vi) each Contract that restricts in any material respect the ability of the Company to compete with or conduct any line of business or employ any Person;
- (vii) each Contract (other than any Contract that is terminable at will or upon not more than 90 days' notice without any Liability to the Company, other than Liability with respect to periods prior to the termination thereof that is unrelated to such termination) that (1) requires any Person to purchase its total requirements of any product or service from any other Person or contains "take or pay" or similar provisions, or (2) contains a "most-favored nation" clause or similar term that provides preferential pricing or treatment or grants exclusive rights of any type or scope to any Person, in each case under clauses (1) and (2), that is material to the Company;
- (viii) each Contract concerning the occupancy, management, or operation of any Real Property, including each Real Property Lease and each lease or sublease from the Company to any other Person with respect to any Real Property;
- (ix) Company IP Agreements (other than a non-exclusive license for the use of any commercially available off-the-shelf software with an aggregate annual license or maintenance fee of less than \$100,000);
- (x) each Contract under which the Company is obligated to repay or has guaranteed any outstanding indebtedness for borrowed money or remains obligated to lend to or make any investment in (in the form of a loan, capital contribution or otherwise) any other Person with outstanding principal amounts in the aggregate exceeding \$200,000;
- (xi) each Contract involving the acquisition or disposition of any business (whether by merger, sale of stock, sale of assets, or otherwise), including any such Contract under which the Company has ongoing obligations for deferred purchase price, contingent, or earn-out payments; and
- (xii) each Contract involving a partnership, joint venture, or otherwise with respect to any equity interests of or owned by the Company or involving a share of profits, losses, costs, or Liabilities by the Company with any other Person, or the appointment of officers, directors, managers, or governors of the Company.
- (b) The Company has made available to Buyer a true, correct and complete copy of each Major Contract, or a form thereof (or, in the case of any oral Major Contract, an accurate summary of the material terms thereof). With respect to each Major Contract (and with the following assuming that each Consent is obtained, which, for any Consent that is a filing or notice, means the making of such filing or notice),

- (i) such Major Contract is in full force and effect and is enforceable in accordance with its terms against the Company and, to the Company's Knowledge, against each other party thereto, except in each case to the extent enforceability may be limited by any Enforcement Limitation, (ii) the Company is not and, to the Company's Knowledge, no other party thereto is in material breach under such Major Contract, and (iii) no party to such Major Contract has terminated, modified, accelerated, or canceled such Major Contract or any material right or Liability thereunder or communicated in writing such party's desire or intent to do so.
- 3.9 <u>Title to and Condition of Certain Assets</u>. The Company has good title to, or a valid leasehold interest in or valid license for, each material tangible asset used by it, shown on the Interim Balance Sheet or acquired by it after the Interim Balance Sheet Date, free and clear of any Encumbrance other than any Permitted Encumbrance, except for any asset disposed of in its Ordinary Course of Business since the Interim Balance Sheet Date. The tangible properties and assets of the Company are in all material respects in satisfactory operating condition for the uses for which they are currently being put, subject to ordinary wear and tear and subject to any maintenance or repairs performed in the Ordinary Course of Business.
- 3.10 <u>Certain Accounts</u>. Section 3.10 of the Disclosure Schedule lists each bank account, cash account, brokerage account and other similar account in which the Company has any interest.

3.11 Real Property.

- (a) Section 3.11(a) of the Disclosure Schedule lists the street address of (i) all real property owned by the Company (each being "Owned Real Property"); and (2) all real property leased or otherwise occupied by the Company pursuant to a real property lease, occupancy agreement or other Contract (each being "Leased Real Property," and any Contract pursuant to which the Company leases or occupies any Leased Real Property, a "Real Property Lease"). The Owned Real Property and the Leased Real Property are, collectively, the "Real Property."
- (b) As to the Real Property, except as would not reasonably be expected to have a Material Adverse Effect:
 - (i) All of the land, buildings, structures and other improvements located thereon or used by the Company in the conduct of its business are included in the Real Property. No improvements constituting a part of the Real Property encroach on real property owned or leased by a Person other than the Company. The use and operation of the Real Property in the conduct of the Company's business do not violate any Law, covenant, condition, restriction, easement, license, permit or agreement.
 - (ii) Except for the Real Property Leases, there is no lease (including sublease) or occupancy agreement or other instrument or agreement granting to any other Person any right to the possession, lease, occupancy or enjoyment of the Real Property in effect with respect to any Owned Real Property or Leased Real Property. There is no pending or Threatened Proceeding regarding condemnation or other eminent domain Proceeding affecting any Real Property or any sale or other disposition of any Real Property in lieu of condemnation.
 - (iii) The Company has a good and valid leasehold interest under each Real Property Lease. No Real Property Lease imposes any restriction that prevents or impairs the operation of the Business conducted at the relevant Leased Real Property. The Company is not in default or otherwise in breach under any Real Property Lease and no other party is in default or otherwise in breach thereof. The Company has made available to Buyer a true, correct and complete copy of each Real Property Lease.
 - (iv) The Company has no Owned Real Property.

(v) The Company has made available to Buyer a true, correct and complete copy of all surveys (including ALTA land title surveys), title insurance commitments, title opinion letters, abstracts of title, recorded deeds, and title insurance policies covering any Real Property that are in the Company's possession or control.

3.12 Environmental Matters.

- (a) The Company has made available to Buyer a true, correct and complete copy of all Phase I and Phase II environmental site assessments and all other material reports related to the environmental condition of any of the Real Property that are in the Company's possession or control.
- Except as would not reasonably be expected to have a Material Adverse Effect. (b) (i) the Company has obtained or filed and is in material compliance with all Permits that are required under any Environmental Law for the operation of the Business as presently conducted, including without limitation those activities relating to the generation, use, storage, treatment, disposal, release or remediation of Hazardous Substances and relating to the compliance of the Real Property with Environmental Law ("Environmental Permits"), (ii) all Environmental Permits remain in full force and effect and any required renewals have been timely filed, (iii) the Company is in compliance with all Environmental Laws and the terms and conditions of all Environmental Permits, and (iv) to the Company's Knowledge, there are no Hazardous Substances present on or in the environmental at any Real Property, including any Hazardous Substances contained in drums, above or underground storage tanks, landfills, land deposits, dumps, equipment (whether moveable or fixed) or other containers, either temporary or permanent, and deposited or located in land, water, sumps, or any other part of any Real Property, or incorporated into any structure therein or thereon, except for those Hazardous Substances used or handled in the Ordinary Course of Business and in compliance with Environmental Laws. Neither the Company, nor any Stockholder, nor any other Person for whose conduct they are or may be held responsible, or any other Person, has permitted or conducted, or is aware of, any Hazardous Activity conducted with respect to the Real Property or any other properties or assets in which the Company has or had an interest except for those Hazardous Activities conducted in the Ordinary Course of Business and in compliance with Environmental Laws. As used herein, "Hazardous Activity" means the distribution, generation, handling, importing, management, manufacturing, processing, production, refinement, Release, storage, transfer, transportation, treatment, or use (including any withdrawal or other use of groundwater) of Hazardous Substances into the environment in a manner so as to result in Liability for the Company under any Environmental Law, and any other act, omission, business, operation or thing that would reasonably be expected to result in Liability for the Company under any Environmental Law.
- There is no Environmental Claim pending or, to the Company's Knowledge, Threatened as of the date of this Agreement against the Company that, if decided adversely to the Company, will have a Material Adverse Effect or will materially and adversely affect the ability of the Company or the Stockholders to consummate the transactions contemplated herein. Except as would not reasonably be expected to have a Material Adverse Effect, neither the Company nor any Stockholder has any basis to expect, nor has any of them (or any other Person for whose conduct they are or may be held responsible) received, any actual or Threatened order, written notice, or other written communication from (i) any Governmental Authority or third party, or (ii) the current or prior owner or operator of any Real Property, of any actual or potential violation or failure to comply with any Environmental Law, or of any actual or Threatened obligation to undertake or bear the cost of any Liability or remedial action or recovery action with respect to the Real Property, or with respect to any property or facility at or to which Hazardous Substances were generated, manufactured, refined, transferred, imported, used, or processed by the Company, or the Stockholders, or any other Person for whose conduct they are or may be held

responsible, or from which Hazardous Substances have been transported, treated, stored, handled, transferred, disposed, recycled, or received.

(d) Except as would not reasonably be expected to have a Material Adverse Effect, the Company has not installed, used, generated, treated, disposed of or arranged for the disposal of any Hazardous Substance in a manner so as to create any Liability under any Environmental Law for the Company. To the Company's Knowledge, there has been no disposal by the Company of any Hazardous Substances on or in any site listed or formally proposed to be listed on the National Priorities List promulgated pursuant to CERCLA or any foreign or state remedial priority list promulgated or maintained pursuant to any comparable foreign or state law, except in any such case in a manner that would have a Material Adverse Effect.

3.13 Intellectual Property.

(a) Section 3.13(a) of the Disclosure Schedule lists all Intellectual Property owned or purported to be owned by the Company that is (i) registered with any Governmental Authority (or with any Person that maintains domain name registrations) and (ii) all applications for any such registration (clauses (i) and (ii) collectively, the "Registered IP"), specifying as to each, as applicable: the title, mark or design; the record owner and inventor(s), if any; the juris diction by or in which it has been issued, registered, or filed; the patent, registration, or application serial number; the issue, registration, or filing date; and the current status. The Company has used commercially reasonable efforts to protect all Registered IP and all unregistered Company Intellectual Property used or held for use in the Company's business which is material to the conduct of such business and listed on Section 3.13(a) of the Disclosure Schedule ("Material Unregistered IP").

To the Company's Knowledge, each item of Registered IP under clause (i) is valid, in full force and effect, and enforceable. The Company owns all right, title, and interest in and to each item of Registered IP and each item of Material Unregistered IP. The Company is the sole and exclusive legal and beneficial owner of all right, title, and interest in and to the Company Intellectual Property (free and clear of all Encumbrances, other than any Permitted Encumbrance), and has the valid and enforceable right to use all other Intellectual Property used in or necessary for the conduct of the Company's business and without payment of a royalty, license fee or similar fee (other than pursuant to a Major Contract or a Contract not required to be disclosed in the Disclosure Schedule to avoid a breach under Section 3.8).

- (b) The Company has not received written notice that any Registered IP or Material Unregistered IP has been declared unenforceable or otherwise invalid by any Governmental Authority. There are no (i) actions (including any opposition, cancellation, revocation, review, or other proceeding), whether settled, pending, or Threatened (including in the form of written offers to obtain a license) challenging the validity, enforceability, registrability, patentability, or ownership of any Company Intellectual Property, (ii) to the Company's Knowledge, actions (including any opposition, cancellation, revocation, review, or other proceeding), whether settled, pending, or Threatened (including in the form of written offers to obtain a license) against the Company challenging the validity, enforceability, registrability, patentability, or ownership of any Licensed Intellectual Property which is material to the conduct of the Company's business, (iii) the Company's right, title, or interest in or to any Company Intellectual Property, or (iv) the Company's right to use any Licensed Intellectual Property.
- (c) (i) The Company has not received any written charge, complaint, claim, demand or notice, alleging that the conduct of Company's business or any use, sale or offer to sell any good or service of the Company or any Company Intellectual Property infringes upon, misappropriates or violates any Intellectual Property right of any other Person, including any claim that the Company must license or refrain from using any Intellectual Property right of any

other Person or any offer by any other Person to license any Intellectual Property right of any other Person (but excluding offers to license solely as part of *bona fide* offers to sell commercial software).

- (ii) To the Company's Knowledge, neither the Company nor the Company Intellectual Property is infringing upon, misappropriating or violating, or has infringed upon, misappropriated or violated, the Intellectual Property of any other Person.
- (iii) Except as set forth in Section 3.13(c)(iii) of the Disclosure Schedule, to the Company's Knowledge, no other Person is infringing upon, misappropriating or violating, has infringed upon, misappropriated or violated, the Company Intellectual Property.
- (d) The Company has entered into binding, valid and enforceable, written Contracts with each current and former employee and independent contractor who is or was involved in or has contributed to the invention, creation, or development of any Intellectual Property during the course of employment or engagement with the Company whereby such employee or independent contractor (i) acknowledges the Company's exclusive ownership of all Intellectual Property invented, created, or developed by such employee or independent contractor within the scope of his or her employment or engagement with the Company; (ii) grants to the Company a present, irrevocable assignment of any ownership interest such employee or independent contractor may have in or to such Intellectual Property, to the extent such Intellectual Property does not constitute a "work made for hire" under Applicable Law; and (iii) irrevocably waives any right or interest, including any moral rights, regarding any such Intellectual Property, to the extent permitted by Applicable Law. All assignments and other instruments necessary to establish, record, and perfect the Company's ownership interest in the Registered IP have been validly executed, delivered, and filed with the relevant Governmental Authorities and authorized registrars.
- (e) The Company has taken reasonable steps to protect, preserve and maintain the confidentiality of all of its confidential information and trade secrets, including requiring its employees and independent contractors having access thereto to execute written nondisclosure agreements. To the Company's Knowledge, no trade secret owned by the Company has been disclosed or authorized to be disclosed to any third party other than pursuant to a nondisclosure agreement or other appropriate instrument that adequately protects the Company's proprietary interests in and to such trade secrets.
- (f) The Company has complied in all material respects with all written terms of use, terms of service, and other Contracts and all associated policies and guidelines relating to its use of any social media platforms, sites, or services (collectively, "Platform Agreements"). There are no actions, whether settled, pending, or Threatened, alleging any material breach or other violation of any Platform Agreement by the Company.
- (g) All Company IT Systems are in good working condition and are sufficient for the operation of the Company's business as currently conducted. In the past five years, there has been no material malfunction, failure, continued substandard performance, denial-of-service, or other cyber incident, including any cyberattack, or other impairment of the Company IT Systems. The Company has taken commercially reasonable steps to safeguard the confidentiality, availability, security, and integrity of the Company IT Systems, including implementing and maintaining commercially reasonable backup, disaster recovery, and software and hardware support arrangements.
- (h) Company is in actual possession of a complete and correct copy of the source code for all proprietary components of the software that is part of the Company Intellectual

Property (the "Company Owned Software'). To the Company's Knowledge, no third party possesses any copy or a material portion of any source code for any Company Owned Software. Furthermore, the Company has (x) implemented reasonable security protocols to provide appropriate notice to applicable department supervisors regarding unauthorized access to the Company Owned Software, and (y) the Company has not received written notice of any breach of such security protocols.

- Except for application programming interfaces and other interface code that is generally available to customers. Company has not disclosed, delivered, licensed, or otherwise made available, and does not have a duty or obligation (whether present, contingent, or otherwise) to disclose, deliver, license, or otherwise make available, any source code for any Company Owned Software to any escrow agent or any other Person, other than an employee, independent contractor, or consultant of Company pursuant to a valid and enforceable written agreement prohibiting use or disclosure except in the performance of services for the Business. Without limiting the foregoing, neither the execution of this Agreement nor the consummation of any of the transactions contemplated by this Agreement will, or would reasonably be expected to, result in the release from escrow or other delivery to any Person of any source code for any Company Owned Software. As of the date hereof, to the Company's Knowledge, there has been no unauthorized theft, reverse engineering, decompiling, disassembling, or other unauthorized disclosure of or access to any source code (including any material portion) for any Company Owned Software. To the Knowledge of the Company, the Company Owned Software is free of any computer instructions, devices or techniques that are designed to infect, disrupt, damage, disable or alter such software or its processing environment, including other programs, equipment and data. The Company has taken reasonable steps to prevent the introduction of such instructions, devices or techniques into the Company Owned Software.
- (j) No Open Source Code has been or is being used or distributed, in whole or in part, in a manner that would require any Company Owned Software (other than any Open Source Code portions thereof that were obtained from third parties and any portions thereof which are distributed as source code by the Company in the Ordinary Course of Business) to be licensed, disclosed or distributed without charge or under terms that require disclosure of source code. For purposes of this Agreement, "Open Source Code" means any software code (including, without limitation, any source code components, applications, plug-ins or libraries) that is distributed as "open source software" or is otherwise distributed publicly or made generally available in source code form under terms that permit modification and redistribution of such software. Open Source Code includes, without limitation, software code that is licensed under the GNU General Public License, GNU Lesser General Public License, Mozilla License, Common Public License, Apache License, BSD License, Artistic License, or Sun Community Source License.
- (k) The Company has in place a privacy policy regarding the collection and use of customer information in accordance with Applicable Law in all material respects. The Company has complied, in all material respects, with all Applicable Law and all Company publicly posted policies, notices, and statements concerning the Company's collection, use, processing, storage, transfer, and security of personal information in the conduct of the Company's business. In the past three years the Company has not (i) to the Company's Knowledge, experienced any actual, alleged, or suspected data breach or other security incident involving personal information in its possession or control or (ii) received any written notice of any audit, investigation, complaint, or other action by any Governmental Authority or other Person concerning the Company's collection, use, processing, storage, transfer, or protection of personal information or actual, alleged, or suspected violation of any Applicable Law concerning privacy, data security, or data breach notification. The transfer by Company to Purchaser of data subject to such privacy policy does not violate such privacy policy or the privacy rights of such customers.

- (l) Neither the execution, delivery or performance of this Agreement, nor the consummation of the transactions contemplated hereunder, will result in the loss or impairment of, or require the consent of any other Person in respect of, the Company's right to own or use any Company Intellectual Property or Licensed Intellectual Property.
- 3.14 <u>Insurance</u>. The Company maintains in full force and effect insurance (including self-insurance) for the Business in amounts and coverages that are reasonable and customary for Persons engaged in businesses similar to that engaged in by the Company and sufficient for material compliance with all Applicable Laws.
- 3.15 <u>Absence of Certain Events</u>. Since the Interim Balance Sheet Date, (a) the Company has conducted the Business in the Ordinary Course of Business and (b) there has not been any event that has had a Material Adverse Effect. Since the Interim Balance Sheet Date to the date of this Agreement, the Company has not taken any action that, if taken after the date of this Agreement, would constitute a breach of, or require the consent of Buyer under, Section 7.1.

3.16 Employee Benefits.

- (a) Section 3.16(a) of the Disclosure Schedule lists all Company Plans and Plans sponsored, maintained or contributed to by an ERISA Affiliate. The Disclosure Schedule separately identifies each Company Plan that contains a change in control provision and each Company Plan that is maintained, sponsored, contributed to, or required to be contributed to by the Company primarily for the benefit of employees outside of the United States.
- (b) Each Company Plan is, and has been for the last six years, administered and operated in compliance in all material respects with its terms, Applicable Laws and any applicable collective bargaining agreements and other similar Contracts.
- (c) Each Company Plan that is intended to be qualified under Section 401 of the Code has either received a favorable determination letter from the IRS with respect to the qualification and tax-exempt status of the Company Plan or is maintained on a preapproved plan document and is entitled to rely upon a favorable opinion letter issued by the IRS with respect to its qualified plan and tax-exempt status, and, to the Company's Knowledge, nothing has occurred (or failed to occur) since the receipt of such determination or opinion letter to cause the loss of the Company Plan's qualification and tax-exempt status, and such Company Plan has been amended in form for compliance with current law. The Company has not maintained, sponsored or contributed to a voluntary employees' beneficiary association under Section 501(c)(9) of the Code.
- (d) For the last six years, each Company Plan has in all material respects appropriately and timely filed and distributed all required reports and descriptions of the Company Plans (including IRS Form 5500 Annual Reports and Summary Plan Descriptions), and all notices required by ERISA, the Code or any other state or federal law, ruling or regulation with respect to the Company Plans. All contributions to the Company Plans for all periods ending on or before Closing (including periods from the first day of the current plan year to Closing) have been properly accrued.
- (e) To the Company's Knowledge, no prohibited transactions (as defined in Section 406 of ERISA or Section 4975 of the Code) have occurred with respect to the Company Plans.
- (f) The minimum funding requirements of Section 412 of the Code and Section 302 of ERISA have always been satisfied, as applicable.
- (g) Neither the Company nor any ERISA Affiliate maintains, sponsors or contributes, or has ever maintained, sponsored or contributed, to any Plan that is (i) a "defined benefit plan" (as defined in ERISA Section 3(35)) or (ii) a Multiemployer Plan and do not reasonably expect to incur any withdrawal liability relating to a Multiemployer Plan. The

Business Combination Agreement

Company and its ERISA Affiliates have not incurred nor reasonably expect to incur any liability under Title IV of ERISA. No Company Plan has assets that include securities issued by the Company.

- (h) There is no material Proceeding pending or, to the Company's Knowledge, Threatened against or with respect to the Company or Company Plan or fiduciary of a Company Plan as of the date of this Agreement regarding any Company Plan, in each case that was brought by or on behalf of any current or former employee of the Company, by any beneficiary or dependent of any such employee, by any Governmental Authority with jurisdiction over any Company Plan or by any fiduciary or co-fiduciary, in each case other than routine claims for benefits.
- (i) None of the payments contemplated by any Company Plan would, in the aggregate, constitute excess parachute payments (as defined in Section 280G of the Code) and neither the execution, delivery or performance of this Agreement would accelerate the payment, vesting or funding of benefits under any of the Company Plans.
- (j) For the last six years, the Company has complied with COBRA in all material respects. The Company has no retiree health plan obligations (other than COBRA).
- (k) The Company has not attempted to maintain the grandfathered health plan status under the Affordable Care Act of any Company Plan that is a welfare benefit plan under Section 3(1) of ERISA.

3.17 Employees and Labor Relations.

- (a) All employees of the Company are "at will" employees whose employment may be terminated without notice and without material Liability therefor. Each Company employee has entered into a non-competition and non-solicitation agreement with the Company. To the Company's Knowledge, the Company has not hired any employee or retained any independent contractor in violation of any restrictive covenant, non-compete agreement, non-solicitation or confidentiality agreement to which such employee or retained independent contractor is a party. No Person has made or Threatened a claim or provided written notice to the Company of any claim that the Company has hired any employee or retained any independent contractor in violation of any such restrictive covenant, non-compete agreement, non-solicitation or confidentiality agreement. To the Company's Knowledge, no current or former employee or independent contractor is in breach of any restrictive covenant, non-compete agreement, non-solicitation or confidentiality agreement between such Person and the Company.
- (b) (i) The Company is not a party to any collective bargaining agreement or other Contract with any union or labor organization; (ii) no union or labor organization or employee group has filed any representation petition or made any written demand to the Company for recognition; (iii) to the Company's Knowledge, since January 1, 2018, no employees of the Company have engaged in any efforts with any union, labor organization, or employee group to organize such employees and no such efforts are Threatened; (iv) no labor strike, work stoppage, picketing, slowdown or other material labor dispute has occurred since January 1, 2018 or, to the Company's Knowledge, is Threatened; and (v) there is no material employment-related Proceeding pending or, to the Company's Knowledge, Threatened, including any such pending or Threatened Proceeding regarding an alleged violation or breach by the Company (or any of its managers, officers, directors, or governors) of any Applicable Law or Contract.
- (c) Since January 1, 2015 the Company has not implemented any plant closing or layoff of employees subject to the requirements of the WARN Act or any similar Applicable Law.
- (d) The Company is in compliance with all Applicable Laws relating to employment and employment practices to the extent the same relate to employees of the Company,

independent contractors or individuals employed through third party staffing agencies, except where the failure to be in compliance would not have a Material Adverse Effect. The Company has not received any written notice from any Governmental Authority or other Person regarding any violation or alleged violation of any Applicable Law relating to hiring, recruiting, employing (or continuing to employ) anyone not authorized to work in the United States, except where such violation or alleged violation would not have a Material Adverse Effect.

- (e) The Company has, based upon its reasonable determination, properly classified all employees, leased employees, consultants and independent contractors that perform services for the Company for all purposes (including for all Tax purposes), in compliance with Applicable Law except where the failure to be in compliance would not have a Material Adverse Effect. All independent contractors and consultants engaged by the Company are properly registered with the applicable Governmental Authority, where required by Applicable Law. All independent contractors have entered into individual entrepreneur services agreements with the Company, where required under Applicable Law. The Company has not retained the services of any unpaid volunteers or unpaid interns to perform services for the Business in the past three years. There is no relationship between the Company and a third party.
- 3.18 <u>Certain Business Relationships</u>. The Company is not a party to any Contract or material transaction with any of its Affiliates, except (a) employment Contracts and Company Plans, (b) any Contract or transaction on terms comparable in all material respects to terms customary in an arm's-length arrangement of such type where the termination of such Contract or transaction will not materially and adversely affect the operation of the Business or (c) as otherwise expressly contemplated herein.
- 3.19 <u>Vendors and Subcontractors</u>. To the Company's Knowledge, its vendors and subcontractors, including subcontractors providing distribution center services, comply with all Applicable Laws in all material respects, including relating to bribery and corruption. To the Company's Knowledge, none of the Company's vendors or subcontractors have been investigated by any Governmental Authority or have been found by any Governmental Authority to be in violation of any Applicable Laws relating to the provision of services to the Company or the sale of the Company's products, or have engaged in conduct that would reasonably be expected to have a Material Adverse Effect on the Company. Section 3.19 of the Disclosure Schedule lists the top twenty vendors to the Company and the Company's top ten subcontractors providing distribution center services for calendar year 2019 based on total transaction value.
- 3.20 <u>Brokers</u>. Except as provided in Section 3.20 of the Disclosure Schedule, the Company does not have any Liability to any broker, finder or similar intermediary in connection with the transactions contemplated herein that would cause Buyer or the Company to become liable for payment of any fee or expense with respect thereto.
- 3.21 No Other Representations or Warranties. None of the company, the stockholders, stockholder representative or any other person makes any representations or warranties to buyer or any other person except as contained in article 3 of this agreement, in each case as modified by the disclosure schedule, or in any other ancillary document executed and delivered by or on behalf of the company, any stockholder or stockholder representative pursuant to this agreement and the company, stockholders, and stockholder representative hereby disclaim any other express or implied representations or warranties, including regarding any pro forma financial information, financial projections or similar

OTHER FORWARD-LOOKING STATEMENTS PROVIDED BY OR ON BEHALF OF THE COMPANY, THE STOCKHOLDERS, OR STOCKHOLDER REPRESENTATIVE.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES OF MERGER SUB AND MERGER SUB 2

Merger Sub and Merger Sub 2 each represent and warrant to the Company as follows as of the date of this Agreement:

4.1 <u>Organization and Good Standing</u>. Merger Sub is a corporation duly organized and validly existing and in good standing under the laws of the state of Delaware. Merger Sub 2 is a limited liability company duly organized, validly existing and in good standing under the laws of the state of Delaware.

4.2 Authority and Authorization; Conflicts; Consents.

- (a) Authority and Authorization. The execution, delivery and performance of this Agreement and each Ancillary Document of Merger Sub and Merger Sub 2 has been duly authorized and approved by all necessary corporate or limited liability company action with respect to Merger Sub or Merger Sub 2, as applicable, and such authorization and approval remains in full force and effect. Assuming due authorization, execution and delivery by the Company of this Agreement and each Ancillary Document to which any is a party, this Agreement is, and each Ancillary Document of Merger Sub or Merger Sub 2 at Closing will be, the legal, valid and binding obligation of Merger Sub or Merger Sub 2, as applicable, enforceable against it in accordance with its terms, except to the extent enforceability may be limited by any Enforcement Limitation.
- Conflicts; Consents. Subject to the receipt of the Consents and the making of filings and submissions referenced in the next sentence, neither the execution nor delivery by Merger Sub or Merger Sub 2 of this Agreement or of any Ancillary Document nor consummation by Merger Sub or Merger Sub 2 of the transactions contemplated herein or therein does or will (with or without the passage of time or giving of notice): (i) constitute a breach of, violate, conflict with or give rise to or create any right or obligation under any Organizational Document of Merger Sub or Merger Sub 2, as applicable; (ii) violate any Applicable Law; or (iii) constitute a breach or violation of or a default under, conflict with or give rise to or create any right of any Person other than Merger Sub or Merger Sub 2, as applicable, to accelerate, increase, terminate, modify or cancel any material right or obligation in a manner adverse to it under any Contract to which it is a party, except where such a breach, violation, default, conflict or right under clause (ii) or (iii) above will not materially and adversely affect its ability to consummate the transactions contemplated in this Agreement. No consent or approval by, notification to or filing with any Person is required in connection with the execution, delivery or performance of this Agreement or any Ancillary Document by Merger Sub or Merger Sub 2 or the consummation by Merger Sub or Merger Sub 2 of the transactions contemplated herein or therein, except (1) the filing of the Certificates of Merger, (2) as required by Antitrust Laws, or (3) for any consent, approval, notice or filing, the absence of which will not materially and adversely affect the ability of Merger Sub or Merger Sub 2 to consummate the transactions contemplated by this Agreement.
- 4.3 <u>Capitalization</u>. The authorized capital stock of Merger Sub consists solely of 1,000 shares of common stock, par value \$0.01 per share, all of which are validly issued and outstanding. All of the issued and outstanding capital stock of Merger Sub is and immediately prior to the First Effective Time will be, owned by Merger Sub 2 (free and clear of all Encumbrances). All of the issued and outstanding equity interests of Merger Sub 2 are, and at the First Effective Time and Second Effective Time will be, owned by Buyer (free and clear of all Encumbrances). Since their respective dates of incorporation or organization, Merger Sub and Merger Sub 2 have not carried on any business nor conducted any operations other than the execution of this Agreement and the performance of their respective obligations hereunder and matters ancillary thereto.

Business Combination Agreement Page 27

- 4.4 <u>Litigation and Orders</u>. There is no Proceeding pending or, to the Knowledge of Buyer, Threatened against Merger Sub or Merger Sub 2 as of the date of this Agreement that, if decided adversely to Merger Sub or Merger Sub 2, will materially and adversely affect the ability of Merger Sub or Merger Sub 2 to consummate the transactions contemplated in this Agreement. As of the date of this Agreement, neither Merger Sub nor Merger Sub 2 is subject to an Order that will materially and adversely affect its ability to consummate the transactions contemplated in this Agreement.
- 4.5 <u>Brokers.</u> Neither Merger Sub nor Merger Sub 2 has any Liability to any broker, finder or similar intermediary in connection with the transactions contemplated herein that would cause Merger Sub, Merger Sub 2 or the Company to become liable for payment of any fee or expense with respect thereto.

ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF BUYER

Except as disclosed in the disclosure schedule delivered by Buyer concurrently with the execution and delivery of this Agreement (the "Buyer Disclosure Schedule"), Buyer hereby represents and warrants to the Company and the Stockholders as follows as of the date of this Agreement:

5.1 Organization and Good Standing. Buyer (a) is a duly organized and validly existing corporation in good standing under the laws of the state of Delaware and (b) has full corporate power and authority to own and lease its properties and assets and conduct its business. Buyer is duly qualified and in good standing to do business as a foreign corporation in each jurisdiction in which the ownership and leasing of its properties and assets or the conduct of its business requires such qualification, except where the failure to be so qualified or in good standing will not materially and adversely affect Buyer's ability to consummate the transactions contemplated in this Agreement.

5.2 Authority and Authorization; Conflicts; Consents.

(a) Authority and Authorization.

- (i) The execution, delivery and performance of this Agreement and each Ancillary Document of Buyer has been duly authorized and approved by all necessary corporate action with respect to Buyer, and such authorization and approval remains in full force and effect. Assuming due authorization, execution and delivery by the Company and the Stockholders of this Agreement and each Ancillary Document to which any is a party, this Agreement is, and each Ancillary Document of Buyer at Closing will be, the legal, valid and binding obligation of Buyer, enforceable against it in accordance with its terms, except to the extent enforceability may be limited by any Enforcement Limitation.
- (ii) The affirmative vote of a majority of the votes cast by the holders of outstanding shares of Buyer Common Stock and Buyer Class F Common Stock, voting as a single class, in person or represented by proxy and entitled to vote thereon, is required to (i) approve this Agreement, the Business Combination and any related transactions contemplated hereby; (ii) approve the adoption of the Post-Closing Equity Incentive Plan; (iii) approve the issuance of the Closing Share Consideration and shares of Buyer Common Stock to be held in reserve by Buyer as part of the Adjustment Reserve Amount as described herein; (iv) approve the Required Buyer Stockholder Approval; and (v) adjourn the Buyer stockholder meeting called for purposes of approving these matters, if necessary, to permit further solicitation of proxies because there are not sufficient votes to approve and adopt these matters. Other than the matters described in this paragraph or any actions consented to by Buyer under Section 8.4(b)(i) that require consent of the holders of Buyer Common Stock and Buyer Class F Common Stock or as set forth in Section 5.2(a)(ii) of the Buyer Disclosure Schedule, there are no other votes

Business Combination Agreement

of the holders of the Buyer's capital stock with respect to the Business Combination, the matters set forth herein or any related matters.

- Conflicts; Consents. Subject to the receipt of the Consents and the making of filings and submissions referenced in the next sentence, neither the execution nor delivery by Buyer of this Agreement or of any Ancillary Document nor consummation by Buyer of the transactions contemplated herein or therein does or will (with or without the passage of time or giving of notice): (i) constitute a breach of, violate, conflict with or give rise to or create any right or obligation under any Organizational Document of Buyer; (ii) violate any Applicable Law; or (iii) constitute a breach or violation of or a default under, conflict with or give rise to or create any right of any Person other than Buyer to accelerate, increase, terminate, modify or cancel any material right or obligation in a manner adverse to Buyer under any Contract to which Buyer is a party, except where such a breach, violation, default, conflict or right under clause (ii) or (iii) above will not materially and adversely affect Buyer's ability to consummate the transactions contemplated in this Agreement. Except as set forth in Section 5.2(b) of the Buyer Disclosure Schedule, no consent or approval by, notification to or filing with any Person is required in connection with Buyer's execution, delivery or performance of this Agreement or any Ancillary Document of Buyer or the consummation by Buyer of the transactions contemplated herein or therein, except (1) the filing of the Certificates of Merger, (2) as required by Antitrust Laws, (3) the approvals of Buyer's Stockholders of the matters described in Section 5.2(a)(ii), (4) the approvals of the holders of the Buyer Public Warrants as described in Section 8.16, or (5) for any consent, approval, notice or filing, the absence of which will not materially and adversely affect Buyer's ability to consummate the transactions contemplated by this Agreement.
- 5.3 <u>Board Approval</u>. The Board of Directors of Buyer has, as of the date of this Agreement, at a meeting duly called and held or by unanimous written consent, adopted resolutions (a) approving this Agreement and the consummation of the transactions contemplated by this Agreement upon the terms and subject to the conditions set forth in this Agreement, (ii) determining that the terms of the Agreement and the transactions contemplated by this Agreement are fair to, and in the best interests of, Buyer and its stockholders, (iii) directing that this Agreement be submitted to the stockholders of Buyer for adoption, (iv) recommending that its stockholders adopt this Agreement and approve the transactions contemplated by this Agreement and (v) declaring that this Agreement is advisable. Such resolutions have not been amended or rescinded by the Board of Directors of Buyer prior to the date of this Agreement.

5.4 Capitalization.

- (a) The authorized capital stock of Buyer as of the date of this Agreement consists of 111,000,000 shares, which includes (i) 100,000,000 shares of Buyer Common Stock, (ii) 10,000,000 shares of Buyer Class F Common Stock, and 1,000,000 shares of Buyer Preferred Stock. The issued and outstanding Buyer Securities as of the date of this Agreement are set forth in Section 5.4(a) of the Buyer Disclosure Schedule. All outstanding Buyer Securities are duly authorized, validly issued, fully paid and non-assessable and not subject to or issued in violation of any purchase option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the DGCL, Applicable Law or the Buyer Organizational Documents or any Contract to which Buyer is a party. None of the outstanding Buyer Securities has been issued in violation of any applicable securities laws. Buyer does not have any Subsidiaries or own any equity interests in any other Person.
- (b) Except as set forth in Section 5.4(b) of the Buyer Disclosure Schedule, there is no (i) preemptive right, option, warrant, put, call, purchase right, subscription right, conversion right, convertible instrument, exchange right or other security, Contract or commitment of any nature whereby any Person has, or has a right to receive, any equity interest of, or right or obligation to acquire any equity interest of, Buyer; (ii) equity appreciation, phantom stock, profit participation or similar right with respect to Buyer; or (iii) voting trust, proxy or other Contract

Business Combination Agreement

with respect to any equity interest of Buyer. Other than the Redemption, as set forth in **Section 5.4(b)** of the **Buyer Disclosure Schedule**, or as expressly set forth in this Agreement, there are no outstanding obligations of Buyer to repurchase, redeem or otherwise acquire any shares of capital stock of Buyer or any other securities of Buyer or to provide funds to make any investment (in the form of a loan, capital contribution or otherwise) in any Person. Except as set forth in **Section 5.4(b)** of the **Buyer Disclosure Schedule**, there are no shareholders agreements, voting trusts or other agreements or understandings to which Buyer is a party with respect to the voting of any shares of capital stock or other securities of Buyer.

- (c) All Indebtedness of Buyer is disclosed in Section 5.4(c) of the Buyer Disclosure Schedule. No such Indebtedness of Buyer contains any restriction upon: (i) the prepayment of any such Indebtedness, (ii) the incurrence of Indebtedness by Buyer or (iii) the ability of Buyer to grant any Encumbrance on its properties or assets.
- (d) Since its date of formation, except as expressly contemplated by this Agreement or described in **Section 5.4(d) of the Buyer Disclosure Schedule** or as disclosed in the Buyer's SEC Reports filed prior to the date of this Agreement, Buyer has not declared or paid any distribution or dividend in respect of its shares of capital stock and has not repurchased, redeemed or otherwise acquired any of its shares of capital stock, and neither Buyer's Board of Directors has authorized any of the foregoing.
- (e) Buyer has no plan or intention to reacquire Buyer Common Stock issued to the Stockholders or to liquidate, merge, or otherwise cause the disposition of any of the assets of the Company after the Mergers, other than dispositions in the Ordinary Course of Business. Following the Mergers, Buyer and its Subsidiaries plan to continue the historic business of the Company or use a significant portion of the Company's business assets in a business.

5.5 SEC Filings and Buyer Financials.

(a) Buyer, since its formation, has filed all SEC Documents with the SEC under the Securities Act and the Exchange Act, together with any amendments, restatements or supplements thereto. Except to the extent available on the SEC's web site through EDGAR, Buyer has delivered to the Company copies in the form filed with the SEC of all of the following: (i) Buyer's annual reports on Form 10-K for each fiscal year of Buyer beginning with the first year Buyer was required to file such a form; (ii) Buyer's quarterly reports on Form 10-Q for each fiscal quarter that Buyer filed such reports to disclose its quarterly financial results and (iii) all other forms, reports, registration statements, prospectuses and other documents (other than preliminary materials) filed by Buyer with the SEC (the forms, reports, registration statements, prospectuses and other documents referred to in clauses (i), (ii) and (iii) above, whether or not available through EDGAR, are collectively, the "SEC Reports") and (iv) all certifications and statements required by (A) Rules 13a-14 or 15d-14 under the Exchange Act and (B) 18 U.S.C. \$1350 (Section 906 of SOX) with respect to any report referred to in clause (i) above (collectively, the "Public Certifications"). The SEC Reports (x) were prepared in all material respects in accordance with the requirements of the Securities Act and the Exchange Act, as the case may be, and the rules and regulations thereunder to the extent in effect at the time of filing and (y) except to the extent an SEC Report has been revised, amended, supplemented or superseded by a later-filed SEC Report that has been filed prior to the date of this Agreement, and did not, as of their respective effective dates (in the case of SEC Reports that are registration statements filed pursuant to the requirements of the Securities Act) and at their respective filing dates with the SEC (in the case of all other SEC Reports) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. The Public Certifications are each true as of their respective dates of filing with the SEC. As used in this Section 5.5, the term "file" shall be broadly construed to include any

manner permitted by SEC rules and regulations in which a document or information is furnished, supplied or otherwise made available to the SEC. (I) The Buyer Public Units, the Buyer Common Stock, and the Buyer Public Warrants are listed on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock may then be listed), (II) Buyer has not received any written deficiency notice from the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed) relating to the continued listing requirements of such Buyer Securities, (III) there are no Proceedings pending or, to the Knowledge of Buyer, Threatened against Buyer by the Financial Industry Regulatory Authority or any other Person with respect to the continued listing of such Buyer Securities on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed), including any intention by such entity to suspend, prohibit or terminate the quoting of such Buyer Securities on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed) and (IV) such Buyer Securities are in compliance with all of the applicable listing and corporate governance rules of the NYSE (or such other nationally recognized stock exchange).

- (b) None of (i) any current report on Form 8-K, and any exhibits thereto or any other report, form, registration or other filing made with any Governmental Authority with respect to the transactions contemplated by this Agreement; (ii) the Information Statement; or (iii) the mailings or other distributions to Buyer's stockholders or prospective investors with respect to the consummation of the transactions contemplated by this Agreement or in any amendment to any of documents identified in (i) through (iii), will, when filed, made available, mailed or distributed, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.
- (c) The financial statements and notes contained or incorporated by reference in the SEC Reports (the "Buyer Financials"), fairly present in all material respects the financial position and the results of operations, changes in shareholders' equity, and cash flows of Buyer at the respective dates of and for the periods referred to in such financial statements, all in accordance with (i) GAAP methodologies applied on a consistent basis throughout the periods involved and (ii) Regulation S-X or Regulation S-K promulgated by the SEC, as applicable (except as may be indicated in the notes thereto and for the omission of notes and audit adjustments in the case of unaudited quarterly financial statements to the extent permitted by Regulation S-X or Regulation S-K, as applicable).
- Buyer has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 and paragraph (e) of Rule 15d-15 under the Exchange Act) as required by Rules 13a-15 and 15d-15 under the Exchange Act. Buyer's disclosure controls and procedures are designed to ensure that all information (both financial and non-financial) required to be disclosed by Buyer in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such information is accumulated and communicated to Buyer's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of SOX. Buyer's management has completed an assessment of the effectiveness of Buyer's disclosure controls and procedures and, to the extent required by Applicable Law, presented in any applicable SEC Report, or any amendment thereto, its conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by such report or amendment based on such evaluation. Based on Buyer's management's most recently completed evaluation of Buyer's internal control over financial reporting, (i) Buyer had no significant deficiencies or material weaknesses in the design or operation of its internal control over financial reporting that would

reasonably be expected to adversely affect Buyer's ability to record, process, summarize and report financial information and (ii) Buyer does not have Knowledge of any fraud, whether or not material, that involves management or other employees who have a significant role in Buyer's internal control over financial reporting.

(e) Except as and to the extent reflected or reserved against in the Buyer Financials, Buyer has not incurred any liabilities or obligations of the type required to be reflected on a balance sheet in accordance with GAAP that is not adequately reflected or reserved on or provided for in the Buyer Financials, other than liabilities of the type required to be reflected on a balance sheet in accordance with GAAP that have been incurred since Buyer's formation in the Ordinary Course of Business.

5.6 Taxes.

- (a) All material Tax Returns that were required to be filed by, on behalf of or with respect to Buyer were timely filed on or before the applicable due date (taking into account any extension of such due date) and were correct and complete in all material respects. All Taxes owed by or with respect to Buyer with respect to any Pre-Closing Tax Period were paid, or, in the case of Taxes not yet due, will be timely paid.
- (b) Buyer has not, since its formation, received a written claim from any Governmental Authority in a jurisdiction where Buyer does not file any Tax Return that Buyer is or may be subject to Taxation by such jurisdiction.
- (c) All material Taxes that Buyer is or was required to withhold or collect in connection with any amount paid or owing to any employee, independent contractor, stockholder, nonresident, creditor or other third party have been duly withheld or collected and have been paid, to the extent required, to the proper Governmental Authority or other Person.
- (d) No Tax audit or other Tax Proceeding is pending or being conducted or is, to Buyer's Knowledge, Threatened against or otherwise with respect to Buyer.
- (e) Buyer does not have in effect any waiver of any statute of limitations regarding any Tax nor has it agreed to any extension of time regarding the assessment of any Tax deficiency.
- (f) There is no Encumbrance for any Tax upon any asset of Buyer, except any Encumbrance for any current Tax not yet due.
- (g) Buyer is not aware of the existence of any fact, or has taken or agreed to take any action, that would reasonably be expected to prevent or impede the transactions contemplated by this Agreement from qualifying as a "reorganization" within the means of Section 368 of the Code.
- (h) Buyer is not a party to, not bound by or has no obligation under any Tax sharing Contract, Tax indemnification Contract, Tax allocation Contract or similar Contract (including any Contract providing for the sharing or ceding of Tax credits or Tax losses) and does not have a Liability to any Person as a result of or pursuant to any such Contract, in each case other than an Ordinary Commercial Agreement.
- (i) Buyer will not be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any Tax period (or portion thereof) ending after the Closing Date as a result of any (i) change in method of accounting made prior the Closing under Code Section 481(c) (or any corresponding or similar provision of state, local or non-U.S. income Tax law); (ii) "closing agreement" as described in Code Section 7121 (or any corresponding or similar provision of state, local or non-U.S. income Tax law) executed prior to the Closing; (iii) installment sale or open transaction disposition made prior to the Closing; (iv) intercompany transaction or any excess loss account described in Treasury Regulations under

Section 1502 of the Code (or any corresponding or similar provision of state, local or non-U.S. Tax law) entered into or created prior to the Closing; or (v) prepaid amount received prior to the Closing outside the Ordinary Course of Business.

- (j) Buyer has not been a member of an affiliated group filing a consolidated, combined or unitary U.S. federal, state, local or non-U.S. income Tax Return (other than a group of which the Buyer was the common parent).
- (k) Buyer does not have any material Liability for the Taxes of any Person (other than the Buyer) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or non-U.S. law), as a transferee or successor, or, except pursuant to an Ordinary Commercial Agreement, by Contract or otherwise.
- (l) Buyer does not have any request for a material closing agreement, private letter ruling or similar ruling in respect of Taxes pending between the Buyer, on the one hand, and any Tax authority, on the other hand.
- (m) Buyer has made available to Company true, correct and complete copies of the U.S. federal income Tax Returns filed by the Buyer.
- (n) Buyer has not in any year for which the applicable statute of limitations remains open distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 or Section 361 of the Code.
- (o) Buyer has not engaged in or entered into a "listed transaction" within the meaning of Treasury Regulation Section 1.6011-4(b)(2).
- (p) Neither the IRS nor any other U.S. or non-U.S. taxing authority or agency has asserted in writing against the Buyer any deficiency or claim for any material Taxes or interest thereon or penalties in connection therewith.
- (q) Buyer has not been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code. Buyer has not received written notice from a non-United States Tax authority that it has a permanent establishment (within the meaning of an applicable Tax treaty) or otherwise has an office or fixed place of business in a country other than the country in which it is organized.
- (r) For U.S. federal income tax purposes, the Buyer is, and has been since its formation, classified as a corporation.
- 5.7 <u>Litigation and Orders</u>. There is no Proceeding pending or, to the Knowledge of Buyer, Threatened against Buyer as of the date of this Agreement that, if decided adversely to Buyer, will materially and adversely affect Buyer's ability to consummate the transactions contemplated in this Agreement. As of the date of this Agreement, Buyer is not subject to an Order that will materially and adversely affect Buyer's ability to consummate the transactions contemplated in this Agreement.
- 5.8 <u>Compliance with Law; Permits.</u> Buyer is, and has been since its formation, in compliance with all Applicable Laws, except where the failure to be in compliance would not have a Material Adverse Effect. No written notice has been received by Buyer since its formation to the date of this Agreement from any Governmental Authority alleging that Buyer is not or was not in compliance with any Applicable Law that has not been remedied or where such non-compliance would not have a Material Adverse Effect. Buyer has in effect all Permits that are required for Buyer to conduct the Business as currently conducted in all material respects, all such Permits are valid and in full force and effect, and Buyer is in compliance in all material respects with the terms of such Permits.

5.9 Buyer Major Contracts.

- (a) Except as set forth in Section 5.9(a) of the Buyer Disclosure Schedule, other than this Agreement and the Ancillary Documents (including the Warrant Amendments) to which Buyer is a party, there are no Contracts to which Buyer is a party which (i) creates or imposes a Liability greater than \$100,000, (ii) is not terminable at will or upon not more than 90 days' notice by Buyer without any Liability to Buyer, other than Liability with respect to services rendered before the termination thereof that is unrelated to such termination, or (iii) prohibits, prevents, restricts or impairs in any material respect any business practice of Buyer as its business is currently conducted, any acquisition of material property by Buyer, or restricts in any material respect the ability of Buyer from engaging in business as currently conducted by it or from competing with any other Person (each, a "Buyer Major Contract"). All Buyer Major Contracts have been made available to the Company other than those that are exhibits to the SEC Reports.
- (b) Buyer has made available to the Company a true, correct and complete copy of each Buyer Major Contract, or a form thereof (or, in the case of any oral Buyer Major Contract, an accurate summary of the material terms thereof). With respect to each Buyer Major Contract (and with the following assuming that each Consent is obtained, which, for any Consent that is a filing or notice, means the making of such filing or notice), (i) such Buyer Major Contract is in full force and effect and is enforceable in accordance with its terms against Buyer and, to Buyer's Knowledge, against each other party thereto, except in each case to the extent enforceability may be limited by any Enforcement Limitation, (ii) Buyer is not and, to Buyer's Knowledge, no other party thereto is in material breach under such Buyer Major Contract, and (iii) no party to such Buyer Major Contract has terminated, modified, accelerated, or canceled such Buyer Major Contract or any material right or Liability thereunder or communicated in writing such party's desire or intent to do so.
- 5.10 <u>Properties</u>. Since its formation, Buyer has not owned, licensed or otherwise has the right to use any material real property or personal property. Since its formation, Buyer has not owned, licensed or otherwise had any right, title or interest in any material Intellectual Property.
- 5.11 <u>Insurance</u>. Buyer maintains in full force and effect insurance (including self-insurance) in amounts and coverages that are reasonable and customary for Persons engaged in business similar to that engaged in by Buyer and sufficient for material compliance with all Applicable Laws.
- 5.12 Absence of Certain Events. Buyer, since its formation, has (a) conducted no business other than its formation, the public offering of its securities (and the related private offerings), public reporting and its search for an initial business combination as described in the final prospectus for Buyer's initial public offering dated November 16, 2017 (the "IPO Prospectus") (including the investigation of the Company and the negotiation and execution of this Agreement) and related activities and (b) has not been subject to a Material Adverse Effect.
- 5.13 Employees and Employee Benefit Plans. Since its formation, Buyer has not (a) had any employees or (b) maintained, sponsored, contributed to or otherwise had any Liability (including by virtue of membership in a "controlled group" of companies) under, any Plan.
- 5.14 <u>Certain Business Relationships</u>. Except as set forth in Section 5.14 of the Buyer Disclosure Schedule or except as disclosed in the Buyer's SEC Reports filed prior to the date of this Agreement, Buyer is not a party to any Contract or material transaction with any of its Affiliates.
- 5.15 <u>Investment Company Act</u>. Buyer is not an "investment company" or a Person directly or indirectly "controlled" by or acting on behalf of an "investment company," in each case within the meaning of the Investment Company Act.
- 5.16 Ownership of Transaction Consideration. Subject to obtaining the Required Buyer Stockholder Approval and the filing of the Amended Buyer Charter with the Secretary of State of the State of Delaware, all Closing Share Consideration be issued and delivered to the Stockholders and all shares of Buyer Common Stock to be reserved in an amount equal to the Adjustment Reserve Amount, in

each case in accordance with **Article 2**, shall be, upon issuance and delivery of such consideration free and clear of all Encumbrances, other than the restrictions under applicable securities laws, the restrictions set forth in **Section 2.7** and any Encumbrances incurred by any Stockholder, and fully paid and non-assessable, and the issuance and sale of Buyer Common Stock pursuant to this Agreement will not be subject to or give rise to any preemptive rights or rights of first refusal.

5.17 Trust Account.

- (a) The Trust Agreement is valid, binding and in full force and effect and enforceable in accordance with its terms and has not been amended or modified (other than as it may be amended or modified during prior to the Closing Date with the consent of the Company not to be unreasonably withheld, delayed or conditioned). Except as set forth in Section 5.17(a) of the Buyer Disclosure Schedule, there are no separate agreements, side letters, or other agreements or understandings that would cause the description of the Trust Agreement in the SEC Reports to be inaccurate in any material respect or that would entitle any Person to any portion of the funds in the Trust Account prior to the Closing other than the matters described in Section 11.1.
- (b) As of the date of this Agreement, the Trust Account consists of no less than \$63.7 million invested in U.S. "government securities" within the meaning of Section 2(a)(16) of the Investment Company Act having a maturity of 180 days or less, or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act. Prior to the Closing, none of the funds held in the Trust Account may be released except for the matters described in Section 11.1.
- 5.18 <u>Lock-Up Agreements</u>. Except as provided in Section 5.18 of the Buyer Disclosure Schedule or in Buyer's SEC Reports, Buyer has not entered into a lock-up agreement with any of Buyer's stockholders, founders or sponsor (as such parties are disclosed in the IPO Prospectus).
- 5.19 <u>Securities</u>. Buyer is acquiring the Shares hereunder for investment, solely for Buyer's own account and not with a view to, or for resale in connection with, any distribution or other disposition thereof in violation of the Securities Act or any applicable state securities law. Buyer acknowledges that none of the Shares may be resold in the absence of registration, or the availability of an exemption from such registration, under the Securities Act or any applicable state securities law. Buyer is an "accredited investor" as defined in Rule 501 promulgated under the Securities Act and has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of an investment in the Shares.
- 5.20 No Discussions. Other than discussions and negotiations relating to the Business Combination, including potential financing that may be required in connection therewith, and discussions with the Company, the Stockholders and their respective Affiliates and Representatives with respect to the Business Combination, Buyer is not actively pursuing with any other Person (each a "Third Party Target"): (a) a sale or exclusive license of all or substantially all of any Third Party Target's assets to Buyer or its' Affiliates; (b) any merger, consolidation or other business combination transaction with respect to any Third Party Target; or (c) the direct or indirect acquisition (including by way of a tender or exchange offer) by Buyer or its Affiliates of beneficial ownership or a right to acquire beneficial ownership of shares representing a majority of the voting power of the then outstanding shares of capital stock of any Third Party Target.
- 5.21 <u>Independent Investigation</u>; <u>No Reliance</u>. The consummation of the transactions contemplated by this Agreement by Buyer is not done in reliance upon any representation or warranty by, or information from, the Company, any Stockholder, or any of their respective Affiliates, employees, or representatives, whether oral or written, express or implied, including any implied warranty of merchantability or of fitness for a particular purpose, except for the representations and warranties specifically and expressly set forth in **Article 3** (as modified by the Disclosure Schedule), and Buyer acknowledges that the Stockholders and the Company expressly disclaim any other representations and

warranties. The consummation of the transaction contemplated by this Agreement by Buyer is instead done entirely on the basis of Buyer's own investigation, analysis, judgment, and assessment of the present and potential value and earning power of the Company, as well as those representations and warranties by the Company and the Stockholders specifically and expressly set forth in Article 3 (as modified by the Disclosure Schedule). Buyer acknowledges that neither the Company nor any Stockholder has made any representations or warranties to Buyer regarding the probable success or profitability of the Company or the Business. Buyer further acknowledges that none of the Company, the Stockholders, nor any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any information regarding the Company or its business or the transactions contemplated by this Agreement not specifically and expressly set forth in Article 3 (as modified by the Disclosure Schedule), and none of the Company, the Stockholders, or any other Person will have or be subject to any Liability to Buyer or any other Person resulting from the distribution to Buyer or its representatives or Buyer's use of any such information, including any confidential information memoranda distributed on behalf of the Company relating to the Business or other publications or data room (including any electronic or "virtual" data room) information provided or made available to Buyer or its representatives, or any other document or information in any form provided or made available to Buyer or its representatives, including any management presentations, in connection with the transactions contemplated by this Agreement.

5.22 <u>Brokers</u>. Except as provided in Section 5.22 of the Buyer Disclosure Schedule, Buyer does not have any Liability to any broker, finder or similar intermediary in connection with the transactions contemplated herein that would cause the Company or any Stockholder to become liable for payment of any fee or expense with respect thereto.

ARTICLE 6 CERTAIN COVENANTS

6.1 Certain Actions to Close Transactions.

Consents and Notifications Generally. On the terms and subject to the conditions of this Agreement (including Sections 6.1(b) and 6.1(c)), the Company and Buyer will use its commercially reasonable efforts to (i) take, or cause to be taken, all actions, and do, or cause to be done, all things, necessary, proper or advisable to cause the conditions to Closing to be satisfied as promptly as practicable (and in any event no later than the Outside Date) and to consummate and make effective, in the most expeditious manner practicable, consummation of the transactions contemplated by this Agreement, including preparing and filing promptly and fully all documentation to effect all necessary filings, notifications, notices, petitions, statements, registrations, submissions of information, applications and other documents (including any required or recommended filings under applicable Antitrust Laws), (ii) obtain as promptly as practicable (and in any event no later than the Outside Date) all approvals, consents, clearances, expirations or terminations of waiting periods, registrations, permits, authorizations and other confirmations from any Governmental Authority or third party necessary, proper or advisable to consummate the transactions contemplated by this Agreement, (iii) defend any Proceedings, whether judicial or administrative, challenging this Agreement or the consummation of the transactions contemplated by this Agreement, and (iv) obtain as promptly as practicable all necessary consents, approvals or waivers from third parties. For purposes of this Agreement, "Antitrust Laws" means the Sherman Act, as amended, the Clayton Act, as amended, the HSR Act, the Federal Trade Commission Act, as amended, and all other Applicable Laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening of competition through merger or acquisition. None of the Company, Stockholders, or Buyer will, nor will they permit any of their respective Affiliates to, take any actions that would, or that could reasonably be expected to, result in any of the conditions set forth in Article 6 either being delayed or not being satisfied.

Business Combination Agreement

- (b) Antitrust Notification. In furtherance and not in limitation of the foregoing, the Company and Buyer will (i) as promptly as practicable, but in no event later than 10 Business Days after the date of this Agreement, file or cause to be filed with the United States Federal Trade Commission (the "FTC") and the Antitrust Division of the United States Department of Justice (the "DOJ") a Notification and Report Form pursuant to the HSR Act with respect to the transactions contemplated by this Agreement and (ii) supply as promptly as practicable any additional information and documentary material that may be requested by the FTC, the DOJ or any other Governmental Authority pursuant to the HSR Act or any other Antitrust Law and use its commercially reasonable efforts to take, or cause to be taken (including by their respective Subsidiaries), all other actions consistent with this Section 6.1 necessary to cause the expiration or termination of the applicable waiting period under the HSR Act as soon as practicable (and in any event no later than the Outside Date). The Company and Buyer will furnish to the other such necessary information and reasonable assistance as the other may request in connection with its preparation of any filing or submission under the HSR Act or such other Antitrust Law.
- Antitrust Actions. Buyer agrees to take, or cause to be taken (including by its Affiliates), any and all steps and to make, or cause to be made, any and all undertakings necessary to resolve such objections, if any, that a Governmental Authority may assert under any Antitrust Law with respect to the transactions contemplated by this Agreement, and to avoid or eliminate each and every impediment under any Antitrust Law that may be asserted by any Governmental Authority with respect to the transactions contemplated by this Agreement, in each case, so as to enable the Closing to occur as promptly as practicable and in any event no later than the Outside Date, including (i) proposing, negotiating, committing to and effecting, by consent decree, hold separate order, or otherwise, the sale, divestiture or disposition of any businesses, assets, equity interests, product lines or properties of Buyer (or its Affiliates), (ii) creating, terminating, or divesting relationships, ventures, contractual rights, or obligations of Buyer or its Affiliates and (iii) otherwise taking or committing to take any action that would limit Buyer's freedom of action with respect to, or its ability to retain or hold, directly or indirectly, any businesses, assets, equity interests, product lines, or properties of Buyer or its Affiliates, in each case as may be required in order to obtain all approvals, consents, clearances, expirations, or terminations of waiting periods, registrations, permits, authorizations, and other confirmations required directly or indirectly under any Antitrust Law or to avoid the commencement of any action to prohibit the transactions contemplated by this Agreement under any Antitrust Law, or, in the alternative, to avoid the entry of, or to effect the dissolution of, any injunction, temporary restraining order or other order in any Proceeding seeking to prohibit the transactions contemplated by this Agreement or delay the Closing beyond the Outside Date. For the avoidance of doubt, nothing in this Agreement requires the Company or any Stockholder to, and Buyer may not and may not agree to, divest, hold separate, or otherwise take any action that limits Buyer's, the Company's, any Stockholder's or their respective Affiliates' freedom of action, ownership or control with respect to, any of the businesses, assets, equity interests, product lines or properties of the Company, any Stockholder. or any of their respective Affiliates.
- (d) Antitrust Proceedings. In furtherance and not in limitation of the covenants of the Parties contained in this Section 6.1, if any Proceeding, including any Proceeding by a private party, is instituted (or Threatened to be instituted) challenging the transactions contemplated by this Agreement or the Ancillary Documents as violative of any Antitrust Law, the Company and Buyer will use commercially reasonable efforts to contest and resist any such Proceeding and to have vacated, lifted, reversed or overturned any Order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the transactions contemplated by this Agreement.
- (e) Cooperation. The Company and Buyer will use (and will cause their respective Affiliates to use) its commercially reasonable efforts to (i) cooperate in all respects with each

other in connection with any filing or submission with a Governmental Authority in connection with the transactions contemplated by this Agreement and in connection with any investigation or other inquiry by or before a Governmental Authority relating to the transactions contemplated by this Agreement, including any Proceeding initiated by a private party, (ii) promptly inform the other Party of (and supply to the other Party) any communication received by such Party from, or given by such Party to, the FTC, the DOJ, or any other Governmental Authority and of any material communication received or given in connection with any Proceeding by a private party, in each case regarding the transactions contemplated by this Agreement, (iii) permit the other Party to review in advance and incorporate the other Party's reasonable comments in any communication to be given by it to any Governmental Authority with respect to obtaining any clearances required under any Antitrust Law in connection with the transactions contemplated by this Agreement and (iv) consult with the other Party in advance of any meeting or teleconference with any Governmental Authority or, in connection with any Proceeding by a private party, with any other Person, and, to the extent not prohibited by the Governmental Authority or other Person, give the other Party the opportunity to attend and participate in such meetings and teleconferences. The Parties will take reasonable efforts to share information protected from disclosure under the attorney-client privilege, work product doctrine, joint defense privilege or any other privilege pursuant to this Section in a manner so as to preserve the applicable privilege.

(f) No Liability for Unobtained Consents and Permits. Except the obligation to use commercially reasonable efforts set forth in Section 6.1 (or as a result of a breach of any representation or warranty in Article 3), none of the Company or any Stockholder will have any obligation whatsoever to Buyer arising out of, relating to, or resulting from the failure to obtain any Consent or Permit (provided that the Company and the Stockholders have performed their obligations under this Agreement). No term of this Agreement will be breached, and no condition will be deemed not satisfied as a result of (i) the failure to obtain any Consent or Permit or (ii) any Proceeding commenced or Threatened by or on behalf of any Person arising out of, relating to, or resulting from the failure to obtain any Consent or Permit.

6.2 Access to Information.

- Pre-Closing Access for Buyer. From the date hereof through Closing, subject to the Nondisclosure Agreement and the reasonable confidentiality precautions of the Company, the Company will (i) cause Buyer and its representatives to have reasonable access during normal business hours and upon prior consent of the Company (not to be unreasonably withheld, conditioned or delayed), to the facilities, books and records of the Company for the opportunity to investigate the Business, and (ii) furnish to Buyer and its representatives such additional financial and operating data and other information relating to the Business as Buyer reasonably requests. Buyer and its representatives will conduct such investigation in a manner that does not unreasonably interfere with the operations of the Company and will not have access to any personnel without the written consent of the Chief Executive Officer of the Company. All nonpublic or otherwise confidential information regarding the Company obtained by Buyer or any of its representatives pursuant to this Section 6.2(a) will be kept confidential in accordance with the Nondisclosure Agreement. Nothing contained in this Section 6.2 obligates the Company or any of its Affiliates to (1) violate any Applicable Law, (2) breach any duty of confidentiality owed to any Person whether such duty arises contractually, statutorily or otherwise (provided that the Company will, and will cause its Affiliates to, use commercially reasonable efforts to obtain any third-party consents or waivers that would permit the supply of such information), or (3) jeopardize the protection of any attorney-client or attorney work product privilege or similar privilege.
- (b) Post-Closing Access for the Stockholders and Buyer Spons or. Throughout the seven-year period after Closing, subject to Buyer's reasonable confidentiality precautions, Buyer will, during normal business hours and upon reasonable notice from any Stockholder or the

Business Combination Agreement Page 38

Buyer Sponsor: (i) cause the Stockholder, the Stockholder Representative, the Buyer Sponsor and their respective representatives to have reasonable access to the books and records (including financial and Tax records, Tax Returns, files, papers and related items) of the Company, and to the personnel responsible for preparing and maintaining such books and records, in each case to the extent necessary or reasonably desirable to (1) defend or pursue any Proceeding, (2) prepare or audit financial statements, (3) prepare or file Tax Returns, or (4) address Tax, accounting, financial or legal matters or respond to any investigation or other inquiry by or under the control of any Governmental Authority; and (ii) permit the Stockholders, Stockholder Representative, the Buyer Sponsor and their respective representatives to make copies of such books and records for the foregoing purposes, at their expense. If requested by Buyer, the applicable Stockholder or the Buyer Sponsor will provide reasonable substantiation of the purpose for such access to show that such access is for any of the foregoing purposes. Nothing contained in this Section 6.2 obligates Buyer or any of its Affiliates to (1) violate any Applicable Law, (2) breach any duty of confidentiality owed to any Person whether such duty arises contractually, statutorily or otherwise (provided that Buyer will, and will cause its Affiliates to, use commercially reasonable efforts to obtain any third-party consents or waivers that would permit the supply of such information), or (3) jeopardize the protection of any attorney-client or attorney work product privilege or similar privilege.

- Sponsor. At and after Closing, the Stockholders and the Buyer Sponsor may retain in their (or any of its Affiliates' or representatives') respective possession, subject to Buyer's reasonable confidentiality precautions, any copies of any books or records of such types described in Section 6.2(b) that are in the possession or control of the Stockholders, Stockholder Representative or the Buyer Sponsor (or the possession or control of any of their respective Affiliates or representatives), in each case for any purpose described in Section 6.2(b).
- 6.3 <u>Further Assurances</u>. If after Closing any further action is necessary, proper or desirable to carry out any purpose of this Agreement, then each Party will take such further action (including the execution and delivery of further documents) as the other Party reasonably requests to carry out such purpose. The foregoing will be at the expense of such requesting Party, except to the extent this Agreement otherwise allocates such expense to any other Party.

6.4 Confidentiality and Publicity.

- (a) Nondisclosure Agreement. Subject to the other terms of this Section 6.4, each Party acknowledges that the information being furnished or made available to it in connection with the transactions contemplated under this Agreement and the Ancillary Documents is subject to the terms of the Confidentiality Agreement between the Company and Buyer, dated July 24, 2020 (the "Nondisclosure Agreement"), the terms of which are incorporated herein by reference. Effective upon, and only upon, the Closing, the Nondisclosure Agreement will terminate with respect to information relating exclusively to the Company.
- (b) Confidentiality. From and after the Closing, each Stockholder shall keep confidential any and all non-public information relating to the Company or the Business; provided, however, that no Stockholder shall be liable hereunder with respect to any disclosure to the extent such disclosure is determined by that Stockholder (with the advice of counsel) to be required or requested in connection with any Applicable Law or Order and provided further that a Stockholder may disclose any such information as reasonably necessary to perform its obligations or enforce its rights under this Agreement and the Ancillary Documents. If a Stockholder is required or requested in connection with any Applicable Law or Order to disclose any non-public information, that Stockholder shall (i) to the extent permissible by such Applicable Law or Order, provide Buyer with prompt written notice of such requirement, (ii) disclose only that information that the Stockholder determines (with advice of counsel) is

required or requested to be disclosed, and (iii) use reasonable efforts to preserve the confidentiality of such non-public information, including by, at Buyer's request, reasonably cooperating with Buyer (at Buyer's sole cost and expense) to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded such non-public information. Notwithstanding the foregoing, such non-public information shall not include information that is or becomes available to the public after the Closing other than as a result of a disclosure by a Stockholder in breach of this Section 6.4(b).

- c) Publicity. Except as stated in this Agreement (including Section 8.6) or as required by Applicable Law or the rules or regulations of the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed), each Party will not, and each Party will cause each of its Affiliates not to, make any public release or announcement regarding this Agreement or any of the transactions contemplated herein without the prior written approval thereof of each Party provided, however, that each Party may make any such announcement which it in good faith believes, based on advice of legal counsel, is necessary or advisable in connection with any requirement of Applicable Law or the rules or regulations of the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed), it being understood and agreed that each Party will, to the extent reasonably practicable, confer with the other Parties concerning the timing and content of such press release or public announcement before the same is made and will consider in good faith any comments made by the other Parties.
- 6.5 Notification of Changes. From the date of this Agreement until the earlier to occur of (a) the Closing or (b) termination of this Agreement in accordance with the provisions of Section 9.3 (the "Interim Period"), each Party shall promptly advise the others in writing of (i) any event occurring subsequent to the date of this Agreement that would render the representations or warranties of that Party contained in this Agreement untrue or inaccurate such that the condition set forth in Section 10.2(a) or Section 10.3(a), as applicable, would not be satisfied, (ii) any breach of the covenants or obligations of that Party under this Agreement or any Ancillary Document to which it is a party such that the condition set forth in Section 10.2(b) or Section 10.3(b), as applicable, would not be satisfied (except to the extent such covenant or obligation shall by its terms be complied with or satisfied at the Closing), (iii) any Material Adverse Effect on that Party. In no event shall the giving of any such notice be deemed to cure any breach of this Agreement or to modify the Disclosure Schedule or Buyer Disclosure Schedule, as applicable, in each case, except if the Closing occurs.
- 6.6 <u>Post-Closing Board of Directors</u>. The Parties shall take all necessary action so that the Post-Closing Buyer Directors are the board of directors of Buyer.

ARTICLE 7 CERTAIN COVENANTS OF THE COMPANY AND STOCKHOLDERS

- 7.1 Pre-Closing Conduct of Business by the Company. Except as expressly contemplated in this Agreement (including in Schedule 7.1), as required under Applicable Law or any Contract in existence on the date of this Agreement to which the Company is a party, or as otherwise consented to in writing by Buyer (which consent will not be unreasonably withheld, conditioned, or delayed), from the date of this Agreement through Closing:
 - (a) the Company will conduct its business in its Ordinary Course of Business; provided, however, that:
 - (i) no action or failure to take action by the Company with respect to the matters addressed in **Section 7.1(b)** shall constitute a breach of this **Section 7.1(a)** unless such action or failure to take such action would also constitute a breach of **Section 7.1(b)**; and

Business Combination Agreement

- (ii) nothing in this **Section 7.1(a)** requires that the Company make any capital expenditure or otherwise pay any amounts or incur any expense, nor does it prohibit the Company from making any distribution of Cash to its equity holders; and
- (b) the Company will not do any of the following:
- (i) issue or otherwise allow to become outstanding or redeem or otherwise acquire any equity interest of the Company or right to any such equity interest, other than in the ordinary course administration of any Company Plan;
 - (ii) split, combine, or reclassify any equity interests of the Company;
- (iii) make any purchase, sale, or disposition of any asset (including by merging or consolidating with, or by purchasing a substantial portion of the stock or assets of, any other Person) of more than \$500,000 individually or \$2,000,000 in the aggregate, except in its Ordinary Course of Business;
- (iv) grant any Encumbrance on any material asset, other than (i) pursuant to a Major Contract (or a Contract not required to be disclosed in the Disclosure Schedule to avoid a breach under **Section 3.8**) or (ii) any Permitted Encumbrance;
- (v) sell, assign, convey, transfer, abandon, allow to lapse, fail to maintain or dispose of any Company Intellectual Property or Company IP Agreements (except nonexclusive licenses or sublicenses granted in the Ordinary Course of Business), or fail to take reasonable measures to protect the confidentiality or value of any material Trade Secrets included in the Company Intellectual Property;
- (vi) become a guarantor with respect to any obligation of any other Person or assume any obligation of any such Person for borrowed money;
- (vii) incur any indebtedness for borrowed money that cannot be prepaid at any time without penalty;
- (viii) except in its Ordinary Course of Business, enter into any Contract that would, if existing on the date hereof, be a "Major Contract," or amend or terminate any Major Contract in any respect that is material and adverse to the Company;
- (ix) (i) fail to prepare and timely file all material Tax Returns with respect to Tax Returns required to be filed by the Company before Closing or timely withhold and remit any Taxes or (ii) make or change any material election with respect to Taxes;
- (x) (i) adopt or change any material accounting method or principle, except as required under GAAP or (ii) change any annual accounting period;
- (xi) except in its Ordinary Course of Business or as may be required under any existing Contract or Applicable Law, (i) enter into any employment agreement or grant any bonus or otherwise materially increase the compensation payable to or to become payable to, or benefits of, any director, officer or management-level employee with an annual salary of more than \$200,000 other than annual cost of living or merit increases of not more than 5% (as determined by reference to the employees of the Company in the aggregate), or (ii) amend in any material respect or terminate any Company Plan or enter into any new arrangements that would, if existing on the date hereof, constitute a Company Plan;
 - (xii) amend any of its Organizational Documents;
- (xiii) pay, discharge, settle or satisfy any material Liability, except (i) in its Ordinary Course of Business or (ii) to the extent reflected or reserved for on the Interim Balance Sheet;

- (xiv) commence or settle any litigation claim that is pending or Threatened against, relating to or involving the Company or the Stockholders, other than in the Ordinary Course of Business, if the settlement amount would not be covered by insurance or reserves reflected in the Financial Statements and would be in excess of \$500,000 individually or \$2,000,000 in the aggregate or that would materially and adversely affect the ability of the Company or the Stockholders to consummate the transactions contemplated herein;
- (xv) file a petition in bankruptcy under any provisions of federal or state bankruptcy law or consent to the filing of any bankruptcy petition against such entity under any other similar Applicable Law;
- (xvi) abandon or discontinue any material existing aspect of the Business or enter into any material new line of business; or
 - (xvii) agree or commit to do any of the foregoing.
- (e) the Stockholders will not do any of the following:
- (i) commence or settle any litigation claim that is pending or Threatened against, relating to or involving the Company or the Stockholders that would materially and adversely affect the ability of the Company or the Stockholders to consummate the transactions contemplated herein; or
- (ii) directly or indirectly Transfer, or permit any Encumbrance to arise or be placed on, any of the Stockholder's shares of Company Common Stock or Company Preferred Stock.
- No Shop. Throughout the period that begins on the date hereof and ends upon the earlier of the First Effective Time or the termination of this Agreement pursuant to Section 9.3, neither the Company nor any Stockholder will, and the Company will cause each authorized representative or agent of Stockholders or the Company not to, directly or indirectly, solicit, initiate, seek, or encourage any inquiry, proposal, or offer from or furnish any information to any Person (other than Buyer or any Person on its behalf) regarding any transaction that would constitute a Company Takeover Proposal. Stockholders and the Company will, and will cause each authorized representative or agent of Stockholders or the Company to, immediately terminate all such discussions or negotiations that may be in progress on the date hereof regarding a Company Takeover Proposal. "Company Takeover Proposal" means any inquiry, proposal or offer from any Person or group relating to, in a single transaction or series of related transactions, any (a) direct or indirect acquisition or exclusive license of 20% or more of the assets of the Company (based on the fair market value thereof) or assets comprising 20% or more of the revenues, net income or EBITDA (net income, before interest, Taxes, depreciation and amortization) of the Company, (b) direct or indirect acquisition of 20% or more of the outstanding Company Common Stock or the outstanding voting power of the Company (or any other equity interests of the Company representing such voting power giving effect to any right of conversion or exchange thereof), (c) tender offer or exchange offer that if consummated would result directly or indirectly in any Person or group (or the stockholders of any Person or group) beneficially owning 20% or more of the outstanding Company Common Stock or the outstanding voting power of the Company (or any other equity interests of the Company representing such voting power giving effect to any right of conversion or exchange thereof), (d) merger, consolidation, share exchange, business combination, recapitalization, liquidation, dissolution or other transaction involving the Company which would result in any Person or group (or the stockholders of any Person or group) beneficially owning, directly or indirectly, 20% or more of the outstanding Company Common Stock or the outstanding voting power of the Company or of the surviving entity in a merger involving the Company or the resulting direct or indirect parent of the Company or such surviving entity (or any equity interests of the Company representing such voting power giving effect to any right of conversion or exchange thereof) or (e) any combination of the foregoing. For the avoidance of doubt, the sales and purchases of inventory and fixed assets in the

Business Combination Agreement

Ordinary Course of Business of the Company, the Business Combination and the transactions contemplated by this Agreement shall not be deemed a Company Takeover Proposal.

7.3 Cooperation with Information Statement and SEC Filings.

- (a) Prior to the Closing and in connection with Buyer's preparation of the Information Statement, any other filing required to be made by Buyer with the SEC under the Exchange Act or any responses to any comments from the SEC relating to the Information Statement, the Schedule TO or other required filings, the Company shall use its commercially reasonable efforts to provide to Buyer, and shall use its commercially reasonable efforts to cause its Representatives, including legal and accounting representatives, to provide all cooperation reasonably requested by Buyer that is customary in connection with the preparation of the Information Statement, Schedule TO and such other filings or responses to SEC comments, which may include, among other things, obtaining the consents of any auditor to the inclusion of the financial statements of Company in the Information Statement and other filings with the SEC. The Company agrees that it will use commercially reasonable efforts to provide all information reasonably necessary for Buyer to prepare and file the Information Statement and Schedule TO no later than October 2, 2020 (or as soon as reasonably practicable thereafter).
- (b) The Company and Buyer shall ensure that none of the information supplied by or on its behalf for inclusion or incorporation by reference in the Information Statement will, at the date it is first mailed to the stockholders of Buyer, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. If at any time prior to the Closing any information relating to the Company or Buyer, or any of their respective Affiliates, directors or officers is discovered by the Company or Buyer that is required to be set forth in an amendment or supplement to the Information Statement so that such document would not include any misstatement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the Party that discovers such information shall promptly notify the other Party and an appropriate amendment or supplement describing such information shall be promptly filed with the SEC and, to the extent required by Law, disseminated to the stockholders of Buyer.
- 7.4 <u>Written Consent.</u> No later than one Business Days following the execution and delivery of this Agreement by all Parties, the Company shall deliver to Buyer a copy of the executed Written Consent.
- 7.5 <u>Stockholders Agreement</u>. The Company and the Stockholders shall take all necessary action to cause the Stockholders Agreement to be terminated effective immediately prior to the Closing,

ARTICLE 8 CERTAIN COVENANTS OF BUYER

8.1 Employee Matters.

(a) Compensation and Severance. Throughout the 12-month period after Closing, Buyer will, or will cause one or more of Buyer's Affiliates (including the Company) to, provide to each employee of the Company (i) base salary or hourly wages which are no less than the base salary or hourly wages provided by the Company immediately prior to the Closing and (ii) target bonus and incentive opportunities which are no less than the target bonus and incentive opportunities, if any, provided by the Company immediately prior to the Closing. If any employee of the Company is terminated by the Company during the 12-month period after Closing in circumstances under which such employee would have been entitled to receive severance benefits immediately prior to the Closing, the Company will provide severance benefits

Business Combination Agreement

to such individuals which are no less than those provided by the Company immediately prior to the Closing.

- Employee Benefits. Throughout the 12-month period after Closing, Buyer will, or will cause one or more of Buyer's Affiliates (including the Company) to, provide to each employee of the Company (and their dependents) employee benefits that as a whole are at least comparable in all material respects to the employee benefits received by the employees (and their dependents) as employees of the Company immediately before Closing. Without limiting any obligation already existing, for purposes of eligibility and vesting for all benefits and for purposes of accruals or entitlement to vacation, sick leave, paid time off, severance and similar benefits, Buyer will cause each employee of the Company to be given or retain (as applicable) full credit under the employee benefits of Buyer and its Affiliates (including the Company) after Closing for such employee's service with the Company or Affiliate before Closing (including the full carryover to such benefits or retention, as applicable, of any earned, but unused, time or pay with respect to the foregoing). Each such employee (and such dependent) will be allowed to participate in such employee benefits without regard to preexisting conditions, waiting periods, evidence of insurability or other exclusions or limitations not imposed on such individual by, or which were otherwise satisfied under, the Company's employee benefit arrangements before Closing. Buyer will apply or credit (or cause to be applied or credited) toward any deductible requirements under all welfare benefit Plans all amounts paid by employees (or dependents) under the welfare benefit Plans of the Company during the Company Plan year in which the Closing Date occurs.
- (c) Certain COBRA Obligations. At all times from and after Closing, Buyer will, or will cause an Affiliate of Buyer to, make available all COBRA Coverage, with respect to current or former employees of the Company or their dependents who are classified as "M&A Qualified Beneficiaries" within the meaning of Treasury Regulation Section 54.4980B-9. "COBRA Coverage" means the continuation of health coverage requirements of Part 6 of Subtitle B of Title I of ERISA as added by COBRA and subsequent amendments or similar Applicable Law.
- (d) No Amendment of Benefits nor Third-Party Beneficiary. Without limiting Sections 11.6 and 11.9, (i) no Company Plan or other employee benefit is or will be deemed to be amended by any term hereof and (ii) no Person, including any employee (or dependent thereof) of the Company, is a third-party beneficiary of any term of this Agreement.
- Exculpation and Indemnification of Directors and Officers. Throughout the six-year period after Closing, Buyer will prevent the amendment, repeal or modification of any provision in any Organizational Document of the Company (including, as result of the Mergers, the Organizational Documents of the Surviving Company) relating to exculpation or indemnification and hereby guaranties, and agrees to promptly pay or cause to be paid, all amounts required to be indemnified by the Surviving Company for acts or omissions prior to the First Effective Time of the Company's former officers, directors, governors, members, partners, or managers holding office before the First Effective Time (except to the extent required by Applicable Law), it being the intent of the Parties that such officers, directors, governors, members, partners, or managers will continue to be entitled to such exculpation and indemnification for such acts or omissions to the greatest extent permitted under Applicable Law. Throughout the six-year period after Closing, Buyer will maintain (or cause to be maintained) in effect directors' and officers' liability insurance covering acts or omissions occurring before the First Effective Time of such officers, directors, governors, members, partners, or managers on terms no less favorable than those of the current directors' and officers' liability insurance; provided, however, that in no event will Buyer be required to expend (or cause to be expended) in any one year an amount exceeding 250% of the annual premium currently paid by the Company for such current coverage; and, provided, further, that if the annual premium for such coverage exceeds such amount, then Buyer will be obligated to maintain (or cause to be

maintained) a policy with the greatest coverage available for a cost not exceeding such amount. Notwithstanding the foregoing, at any time prior to the First Effective Time, the Company may cause the Company to purchase "tail" directors' and officers' liability insurance policies, covering the same persons and providing the same terms with respect to coverage and amount as aforesaid, and the terms of which will provide coverage until the sixth anniversary of the Closing, and upon the purchase of such insurance, Buyer's obligations under the immediately preceding sentence will be deemed satisfied. Buyer's obligations under this Section will not be terminated or modified in a manner that adversely affects any Person to whom this Section relates with respect to acts or omissions prior to the First Effective Time without the consent of such affected Person (it being expressly agreed that each such Person to whom this Section relates will be a third-party beneficiary of this Section).

- 8.3 No Shop. Throughout the period that begins on the date hereof and ends upon the earlier of the First Effective Time or the termination of this Agreement pursuant to Section 9.3, Buyer will not, and will cause each authorized representative or agent of Buyer not to, directly or indirectly, solicit, initiate, seek, or encourage any inquiry, proposal, or offer from, furnish any information to, or participate in any discussion or negotiation with any Person (other than the Company, the Stockholders or any Person on their behalf) regarding any transaction (other than the transactions contemplated by this Agreement) that would constitute a Business Combination.
- 8.4 <u>Pre-Closing Conduct of Business by Buyer</u>. Except as expressly contemplated in this Agreement, as required under Applicable Law or any Contract in existence on the date of this Agreement to which Buyer is a party, or as otherwise consented to in writing by the Company (which consent will not be unreasonably withheld, conditioned, or delayed), from the date of this Agreement through Closing:
 - (a) Buyer will conduct its business in its Ordinary Course of Business; provided, however, that no action or failure to take action by Buyer with respect to the matters addressed in Section 8.4(b) shall constitute a breach of this Section 8.4(a) unless such action or failure to take such action would also constitute a breach of Section 8.4(b); and
 - (b) the Company will not do any of the following:
 - (i) except as mutually agreed in writing by Buyer and the Company, issue or otherwise allow to become outstanding or redeem or otherwise acquire any equity interest of Buyer or right to any such equity interest;
 - (ii) split, combine, or reclassify any equity interests of Buyer;
 - (iii) make any purchase, sale, or disposition of any material asset (including by merging, consolidating or entering into a business combination with, or by purchasing a substantial portion of the stock or assets of, any other Person);
 - (iv) grant any Encumbrance on any material asset, other than (i) pursuant to a Buyer Major Contract (or a Contract not required to be disclosed in the Disclosure Schedule to avoid a breach under **Section 5.9**) or (ii) any Permitted Encumbrance;
 - (v) become a guarantor with respect to any obligation of any other Person or assume any obligation of any such Person for borrowed money;
 - (vi) incur any indebtedness for borrowed money that cannot be prepaid at any time without penalty;
 - (vii) except in its Ordinary Course of Business, enter into any Contract that would, if existing on the date hereof, be a "Buyer Major Contract," or amend or terminate any Buyer Major Contract in any respect that is material and adverse to Buyer;

Business Combination Agreement

- (viii) (i) fail to prepare and timely file all material Tax Returns with respect to Tax Returns required to be filed by Buyer before Closing or timely withhold and remit any Taxes or (ii) make or change any material election with respect to Taxes;
- (ix) (i) adopt or change any material accounting method or principle, except as required under GAAP or (ii) change any annual accounting period;
 - (x) amend any of its Organizational Documents;
- (xi) pay, discharge, settle or satisfy any material Liability, except in its Ordinary Course of Business;
- (xii) commence or settle any litigation claim that is pending or Threatened against, relating to or involving Buyer;
- (xiii) file a petition in bankruptcy under any provisions of federal or state bankruptcy law or consent to the filing of any bankruptcy petition against such entity under any other similar Applicable Law;
 - (xiv) establish any Subsidiary or enter into any new line of business;
 - (xv) enter into any Contract with respect to the voting of Buyer Securities;
 - (xvi) make any capital expenditure;
- (xvii) take any action that would reasonably be expected to significantly delay or impair (A) the timely filing of any of its public filings with the SEC (giving effect to permitted extensions), or (B) its compliance in all material respects with Applicable Laws relating to securities; or
 - (xviii) agree or commit to do any of the foregoing.

8.5 Use of Trust Account Proceeds.

Prior to the Closing, none of the funds held in the Trust Account may be used or released except (i) for the withdrawal of interest to pay franchise and income taxes, (ii) up to \$187,500 to fund working capital requirements, and (iii) to effectuate the Tender Offer. The Parties agree that at the Closing, the funds in the Trust Account, after taking into account payments for the Redemption, shall first be used (a) to pay the Preferred Payment and (b) up to \$16,400,000 to pay all Buyer Transaction Expenses. Any remaining proceeds shall be distributed to the Company and used for working capital and general corporate purposes. Thereafter, the Trust Account and Trust Agreement shall be terminated.

8.6 Tender Offer.

Tender Offer. As promptly as practicable, and in any event not more than the (a) later of 10 days from the date hereof or seven Business Days following delivery of the Company Historical Financials (the "TO Commencement Date"), Buyer shall commence (under the meaning of Rule 14d-2 under the Exchange Act) an offer to purchase all outstanding shares of Buyer Common Stock (the "Tender Offer") for cash equal to their pro rata share of the aggregate amount then on deposit in the Trust Account, less Taxes and interest (the "Tender Consideration"). Buyer agrees that no shares of Buyer Common Stock held by Buyer will be tendered in the Tender Offer. Unless otherwise agreed to by the Parties, Buyer shall use its commercially reasonable efforts (subject to market conditions) to conduct the Tender Offer without stockholder vote pursuant to Rule 13e-4 and Regulation 14E of the Exchange Act (as modified, waived or otherwise agreed to with the SEC) which regulates issuer tender offers and in compliance with the requirements of Buyer's certificate of incorporation, and will file Tender Offer documents with the SEC. The obligation of Buyer to accept for payment shares of Buyer Common Stock validly tendered and not validly withdrawn pursuant to the Tender Offer shall be subject to the satisfaction of each of the conditions set forth in the Tender Offer. Unless agreed to

by the Parties or required by the SEC, no material change (including changing the amount per share offered to the stockholders) may be made to the Tender Offer which imposes additional conditions to the Tender Offer or is inconsistent with this Section. The Parties hereby agree to negotiate in good faith to amend the conditions set forth in the Tender Offer to reflect any changes that may be reasonably required as a result of discussions with the SEC or its staff. Unless an extension of the tender offer period is agreed to by the Parties or is required as a result of a change required by the SEC, the Tender Offer shall expire on the date that is 20 Business Days following the commencement of the Tender Offer (the "Initial Expiration Date"). Notwithstanding the foregoing, and subject to the provisions of Section 9.3, if, at any scheduled expiration of the Tender Offer, the conditions set forth in the Tender Offer, have not been satisfied or waived, Buyer may extend the Tender Offer for one or more consecutive periods beyond the Initial Expiration Date (the Initial Expiration Date as extended, the "Expiration Time"). Notwithstanding the foregoing, Buyer, without the consent of the Company, may extend the Tender Offer for any period required by any Applicable Law or any change requested or required by the SEC, or the staff thereof, applicable to the Tender Offer.

- (b) **Payment Obligations.** Subject to the terms of the Tender Offer and this Agreement and the satisfaction or waiver of all of the conditions to the Tender Offer, Buyer shall accept for payment all shares of Buyer Common Stock validly tendered and not validly withdrawn pursuant to the Tender Offer promptly after the Expiration Time and pay for such shares as soon as practicable (and, in any event, no more than three Business Days) after the Expiration Time.
- c) Payments to Persons Other than Registered Holders. If the payment to a registered holder under the Tender Offer is to be made to a Person other than the Person in whose name the surrendered certificate formerly evidencing the shares of Buyer Common Stock is registered on the transfer books of Buyer, it shall be a condition of payment that the certificate so surrendered shall be endorsed properly or otherwise be in proper form for transfer and that the Person requesting such payment shall have paid all transfer and other Taxes required by reason of the payment of such consideration to a Person other than the registered holder of the certificate surrendered, or shall have established to the reasonable satisfaction of Buyers that such Taxes either have been paid or are not applicable.
- Offer Documents. As promptly as reasonably practicable on the date of the commencement of the Tender Offer, Buyer shall file with the SEC a Tender Offer Statement on Schedule TO (together with all amendments and supplements thereto, the "Schedule TO") with respect to the Tender Offer which shall contain or shall incorporate by reference an offer to purchase (an "Offer to Purchase") and forms of the related letter of transmittal and any related summary advertisement (such Schedule TOs, Offers to Purchase and such other documents, together with all supplements and amendments thereto, being referred to herein collectively as the "Offer Documents"). The Offer Documents will comply as to form in all material respects with the requirements of the Exchange Act and the rules and regulations promulgated thereunder and in addition shall contain substantially the same financial and other information about the transactions contemplated by this Agreement and the redemption rights as is required under Regulation 14A of the Exchange Act which regulates the solicitation of proxies. Each Party agrees to correct promptly any information provided by it for use in the Offer Documents that shall have become false or misleading in any material respect, and Buyer further agrees to take all steps necessary to cause the Schedule TO, as so corrected, to be filed with the SEC, and the other Offer Documents, as so corrected, to be disseminated to holders of shares of Buyer Common Stock, in each case as and to the extent required by applicable federal securities Laws. No filing of, or amendment or supplement to, the Offer Documents shall be made by Buyer without the prior consent (which shall not be unreasonably withheld, delayed or conditioned) of the Company. Buyer shall give the Company and its counsel a reasonable opportunity to review and

Business Combination Agreement Page 47

comment on the Offer Documents prior to such documents being filed with the SEC or disseminated to holders of the shares of Buyer Common Stock and shall give due consideration to all reasonable additions, deletions or changes suggested thereby by the Company and its counsel. Buyer shall provide the Company and its counsel with any comments (whether written or oral) that Buyer or its counsel may receive from the SEC or its staff with respect to the Offer Documents promptly after the receipt of such comments, shall provide the Company and its counsel with a reasonable opportunity to review and comment on the responses of Buyer to such comments, shall allow the Company and its counsel to participate in any discussions with the SEC or its staff, and shall give due consideration to all reasonable additions, deletions or changes suggested thereby by the Company and its counsel.

- Company Cooperation. The Company acknowledges that a substantial portion of the filings with the SEC and mailings to holders of shares of Buyer Common Stock with respect to the Tender Offer shall include disclosure regarding the Company and its management, operations and financial condition. Accordingly, the Company agrees to (i) deliver the Company Historical Financials not later than October 2, 2020 (or as soon as reasonably practicable thereafter) and (ii) as promptly as reasonably practicable but in no event later than seven Business Days prior to the TO Commencement Date, provide Buyer with such information as shall be reasonably requested by Buyer for inclusion in or attachment to the Offer Documents to be filed or mailed as of and following the commencement of the Tender Offer, that is accurate in all material respects and complies as to form in all material respects with the requirements of the Exchange Act and the rules and regulations promulgated thereunder and in addition shall contain substantially the same financial and other information about the Company as is required under Regulation 14A promulgated under the Exchange Act regulating the solicitation of proxies even if such information is not required under the tender offer rules. The Company understands that such information shall be included in the Offer Documents or responses to comments from the SEC or its staff in connection therewith and mailings. The Company shall make its officers and employees reasonably available to Buyer and its counsel in connection with the drafting of such filings and mailings and responding in a timely manner to comments from the SEC.
- (f) **Termination of Tender Offer**. Buyer shall not terminate the Tender Offer prior to any scheduled Expiration Time except in the event this Agreement is terminated pursuant to the terms hereof.
 - (g) Other Actions. Notwithstanding anything to the contrary in Section 6.4(c):
 - (i) as promptly as practicable after the execution of this Agreement (and in any event before the opening of trading on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed) on the next Business Day thereafter), Buyer and the Company shall mutually agree on and issue a press release announcing the execution of this Agreement (the "Signing Press Release"). Immediately after the issuance of the Signing Press Release (and in any event within four (4) Business Days after the execution of this Agreement), Buyer shall prepare and file a pre-commencement Schedule TO-C or Report on Form 8-K pursuant to the Exchange Act to report the execution of this Agreement, attaching this Agreement, and describing the terms thereof as required by applicable federal securities laws, and the Signing Press Release thereto (the "Signing Filing"), which the Company shall review, comment upon and approve (which approval shall not be unreasonably withheld, conditioned or delayed) prior to filing; and
 - (ii) as promptly as practicable after the completion of the Tender Offer, Buyer shall prepare a draft amendment to Schedule TO or Form 8-K announcing the completion of the Tender Offer, if applicable, together with, or incorporating by reference such other information that may be required to be disclosed with respect to such

Business Combination Agreement Page 48

results, including the Closing of the Business Combination, if applicable, in any report or form to be filed with the SEC, which the Company shall review, comment upon and approve (which approval shall not be unreasonably withheld, conditioned or delayed) prior to filing. As promptly as practicable after the completion of the Tender Offer, Buyer and the Company shall mutually agree on and issue a press release announcing the results of the Tender Offer and, if applicable, the consummation of the Business Combination ("Closing Press Release") and if applicable, Buyer shall prepare a draft Form 8-K announcing the completion of the Tender Offer, which the Company shall review, comment upon and approve (which approval shall not be unreasonably withheld, conditioned or delayed) prior to filing ("Closing Filing"). Concurrently with the Closing, if not included in the same filings after completion of the Tender Offer, Buyer and the Company shall mutually agree on, and Buyer shall distribute, the Closing Press Release and Buyer shall file the Closing Filing with the SEC as soon as reasonably practicable thereafter.

- 8.7 <u>Registration Rights Agreement</u>. Buyer agrees to execute and deliver a Registration Rights Agreement in the form attached hereto as **Exhibit 8.7** (the "**Registration Rights Agreement**") with the other parties thereto at Closing, which shall include all of the Stockholders.
- 8.8 <u>Sponsor Lock-Up Restriction</u>. Buyer shall, within one day following the date of this Agreement, enter into an amended and restated lock-up agreement with the Buyer Sponsor on the same terms as those provided in Section 2.7 (the "Sponsor Lock-Up Agreement").
- 8.9 <u>Blue Sky Laws</u>. Buyer shall take such steps as may be necessary to comply with the securities and blue sky laws of all jurisdictions which are applicable to the issuance of the Buyer Common Stock in connection with the transactions contemplated by this Agreement and the registration and resale of the Buyer Common Stock pursuant to the Registration Rights Agreement. The Company shall reasonably cooperate with Buyer, at Buyer's reasonable request, to assist Buyer to comply with the securities and blue sky laws of all jurisdictions which are applicable in connection with the issuance of Buyer Common Stock in connection with the transactions contemplated by this Agreement and the registration and re-sale of the Buyer Common Stock pursuant to the Registration Rights Agreement.

8.10 Stockholder Consent and Information Statement.

- As promptly as reasonably practicable following the execution of this Agreement and in lieu of calling a meeting of the Buyer's stockholders, the Buyer shall submit to certain of the Buyer's stockholders holding, in the aggregate, a number of shares of Buyer Securities sufficient to approve the Buyer Stockholder Approval Matters by the Required Buyer Stockholder Approval, (i) within one Business Day after the date hereof, the form of irrevocable written consent attached hereto as Exhibit 8.10(a)(i) of this Agreement, and (ii) no later than one Business Day prior to the date of the Information Statement, the form of irrevocable written consent attached hereto as Exhibit 8.10(a)(ii), (such written consents, as duly executed and delivered by such Stockholders, and taken together, the "Buyer Stockholders' Written Consent"). As soon as practicable upon receipt of the Buyer Stockholders' Written Consent by the corporate secretary of the Buyer, the Buyer will provide the Company with a copy of such Buyer Stockholders' Written Consent, certified as true and complete by the corporate secretary of the Company. In connection with the Buyer Stockholders' Written Consent, the Buyer shall take all actions necessary to comply, and shall comply in all respects, with the requirements of Section 228 of the DGCL and the applicable provisions of the Organizational Documents of the Company.
- (b) Buyer shall, promptly following the delivery of the Buyer Stockholders' Written Consent provide Buyer's stockholders an opportunity in accordance with the Buyer Organizational Documents and the IPO Prospectus to have their shares of Buyer Common Stock redeemed (the "Redemption"). The Buyer Stockholders' Written Consent shall include

Business Combination Agreement

resolutions in favor of approving (i) the approval of this Agreement and the transactions contemplated hereby by Buyer's stockholders in accordance with the Buyer Organizational Documents, the DGCL and the rules and regulations of the SEC and NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed), (ii) the approval of the issuance of the Closing Share Consideration and the shares of Buyer Common Stock representing the Adjustment Reserve Amount in accordance with the rules of the NYSE (or of such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed), (iii) the adoption and approval of an amended and restated certificate of incorporation of Buyer (the "Amended Buyer Charter") in form and substance reasonably acceptable to Buyer and the Company, which among other things will (A) change Buyer's name to PARTS iD, Inc., (B) designate the classes of the members of the Post-Closing Buyer Board, and (C) eliminate provisions allowing the stockholders to act by written consent in lieu of a stockholders meeting, (iv) the adoption of the Post-Closing Equity Incentive Plan, (v) the adoption of an employee stock purchase plan, in form and substance reasonably acceptable to the Company, that provides for the issuance of Buyer Common Stock by Buyer subject to an aggregate cap on such awards equal to 5% of the number of shares of Buyer Common Stock on a fully diluted basis, and (vi) such other matters as the Company and Buyer shall hereafter mutually determine to be necessary or appropriate in order to effect the transactions contemplated by this Agreement (the approvals described in foregoing clauses (i) through (vi), collectively, the "Buyer Stockholder Approval Matters"). As promptly as practicable following the execution of this Agreement (but in any event not later than the TO Commencement Date), Buyer shall prepare and file with the SEC a written information statement of the type contemplated by Rule 14c-2 of the Exchange Act (such information statement, including any amendment or supplement thereto, the "Information Statement") containing (i) the information specified in Schedule 14C under the Exchange Act concerning the Buyer Stockholders' Written Consent and the Buyer Stockholder Approval, and (ii) a notice to every Buyer stockholder that did not execute the Buyer Stockholder Written Consent in accordance with the provisions of this Section 8,10(b) of this Agreement, the Exchange Act, the DGCL and the Organizational Documents of the Buyer. Each of the Company and the Stockholders shall promptly furnish to the Buyer all information concerning such Person as may be reasonably requested in connection with the preparation, filing and distribution of the Information Statement. The Information Statement shall comply as to form in all material respects with the applicable requirements of the Exchange Act, the DGCL and the rules of the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed). If, at any time prior to the date that is twenty (20) days after the Information Statement is first mailed to the Stockholders, any information relating to the Buyer, the Company, the Stockholders or any of their respective affiliates, officers or directors should be discovered by the Buyer, the Company, or the Stockholders which is required to be set forth in an amendment or supplement to the Information Statement, so that the Information Statement shall not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading, the Party that discovers such information shall promptly notify the other Parties hereto, and an appropriate amendment or supplement describing such information shall be filed with the SEC and, to the extent required by Applicable Law, disseminated to the Stockholders. Prior to filing with the SEC, Buyer will make available to the Company drafts of the Information Statement and any other related documents to be filed with the SEC, both preliminary and final, and any amendment or supplement to the Proxy Statement or such other related document and will provide the Company with a reasonable opportunity to comment on such drafts and shall consider such comments in good faith. Buyer shall not file any such documents with the SEC without the prior written consent of the Company (such consent not to be unreasonably withheld, conditioned or delayed). The notice to every Buyer stockholder that did not that did not execute the Buyer Stockholder Written Consent included in the Information Statement shall (i) be a statement to the effect that the Buyer board of directors

unanimously determined that the Business Combination is advisable in accordance with the DGCL and in the best interests of Buyer's stockholders and unanimously approved and adopted this Agreement and the Business Combination and (ii) provide Buyer's stockholders to whom it is sent with notice of the actions taken in the Required Buyer Stockholder Approval, including the approval and adoption of this Agreement and the Business Combination in accordance with the DGCL and the certificate of incorporation and bylaws of Buyer. If applicable, the notice will include therewith a copy of Section 262 of the DGCL, and shall be sufficient in form and substance as contemplated by the DGCL.

- (c) Buyer shall take any and all actions required to satisfy the requirements of the Exchange Act and other Applicable Laws in connection with the Information Statement and the Redemption.
- 8.11 <u>Listing of Buyer Common Stock</u>. Buyer shall cause the Buyer Common Stock to be approved for listing on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed), subject to official notice of issuance, prior to the Closing. From and after the Closing, Buyer shall be listed on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed) as "ID" or such other available trading symbol as Buyer and the Company may mutually agree upon.
- 8.12 <u>Buver Public Filings</u>. During the period prior to the Closing Date, Buyer will file all SEC Documents required to be filed by Buyer subsequent to the date of this Agreement and will keep current and timely fill all of its SEC Documents and otherwise comply in all material respects with applicable securities Laws and shall use its commercially reasonable efforts to maintain the listing of the Buyer Public Units, the Buyer Common Stock and the Buyer Public Warrants on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed); provided, that the Parties acknowledge and agree that from and after the Closing, Buyer intends to list on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are listed immediately prior to the Closing) only the Buyer Common Stock and the Buyer Public Warrants.
- 8.13 <u>Documents and Information</u>. After the Closing Date, Buyer shall, and shall cause its Subsidiaries to, until the seventh anniversary of the Closing Date, retain all books, records and other documents pertaining to the business of the Company in existence on the Closing Date and make the same available for inspection and copying by the Stockholder Representative during normal business hours of Buyer and its Subsidiaries, as applicable, upon reasonable request and upon reasonable notice. No such books, records or documents shall be destroyed after the seventh anniversary of the Closing Date by Buyer or its Subsidiaries without first advising the Stockholder Representative in writing and giving the Stockholder Representative a reasonable opportunity to obtain possession thereof.
- 8.14 Post-Closing Board of Directors and Executive Officers. Buyer shall take all necessary action, including causing directors of Buyer to resign or otherwise be removed from their positions as directors of Buyer, so that the Post-Closing Buyer Directors are the board of directors of the Buyer. Buyer shall take all necessary action, including causing executive officers of Buyer to resign or otherwise be removed from their positions effective as of the Closing so that Antonino Ciappina will be the chief executive officer of Buyer and Kailas Agrawal will be the Chief Financial Officer of Buyer immediately after the Closing.
- 8.15 Tax Status of Merger Sub 2. Buyer has not and will not elect under Treasury Regulation § 301.7701-3 to cause Merger Sub 2 to be treated as a corporation for federal income tax purposes effective for any period on or before the date of the Second Merger or otherwise causing Merger Sub 2 not to be treated as a disregarded entity wholly owned by Buyer for the same period.

8.16 Warrant Amendment.

Business Combination Agreement Page 51

- Buyer shall use its commercially reasonable best efforts to obtain the vote or consent of the holders of at least 65% of the outstanding Buyer Public Warrants that were initially issued to investors in the IPO to: (i) amend that certain Warrant Agreement between Buyer and Continental Stock Transfer & Trust Company, dated as of November 16, 2017 (as amended from time to time, the "Warrant Agreement"), to provide, among other things, that each outstanding Buyer Public Warrant shall no longer be exercisable to purchase one-half of a share of Buyer Common Stock for \$5.75 per half-share (subject to adjustment as provided in the Warrant Agreement) and instead shall be converted solely into the right to receive: if, at the Closing, the aggregate gross cash in the Trust Account, plus the aggregate gross proceeds received by Buyer pursuant to any private offering conducted in accordance with Section 8.4. (A) is at least \$60,000,000, \$0.35 in cash and 0.065 of a share of Buyer Common Stock, (B) is less than \$60,000,000, but at least equal to \$44,000,000, \$0.25 in cash and 0.075 of a share of Buyer Common Stock, or (C) is less than \$44,000,000, \$0.18 in cash and 0.082 of a share of Buyer Common Stock (the "Public Warrant Amendment"), and (ii) to amend the Warrant Agreement to provide, among other things, that each of the 2,912,230 outstanding warrants issued pursuant to that certain Sponsor Warrants Purchase Agreement dated as of October 24, 2017 between Buyer and Buyer Sponsor (the "Buyer Private Placement Warrants") and not owned by Buyer Sponsor shall no longer be exercisable to purchase one-half of a share of Buyer Common Stock for \$5.75 per half-share (subject to adjustment as provided in the Warrant Agreement) and instead shall be converted solely into the right to receive; if, at the Closing, the aggregate gross cash in the Trust Account, plus the aggregate gross proceeds received by Buyer pursuant to any private offering conducted in accordance with Section 8.4. (A) is at least \$60,000,000, \$0.35 in cash and 0.065 of a share of Buyer Common Stock, (B) is less than \$60,000,000, but at least equal to \$44,000,000, \$0.25 in cash and 0.075 of a share of Buyer Common Stock, or (C) is less than \$44,000,000, \$0.18 in cash and 0.082 of a share of Buyer Common Stock (the "Private Warrant Amendment" and together with the Public Warrant Amendment, the "Warrant Amendments").
- (b) As of the date of this Agreement, Buyer has entered into agreements with holders of at least 65% of the outstanding Buyer Public Warrants to vote in favor of the Warrant Amendments (each such agreement, a "Warrant Holder Support Agreement"). Each Warrant Holder Support Agreement (i) has been made available to the Company, and (ii) with respect to Buyer and, to the Buyer's Knowledge, with respect to each holder of Buyer Public Warrants party thereto, is valid and enforceable in accordance with its terms.
- In connection with the Warrant Amendments, as promptly as practicable following the execution of this Agreement (but in any event not later than the TO Commencement Date), Buyer shall prepare and file with the SEC a written consent solicitation in preliminary form seeking the approval of the holders of the Buyer Public Warrants of the Warrant Amendments contemplated hereby (the "Warrantholder Consent Solicitation Statement"), in accordance with and as required by Buyer's Organizational Documents, the Warrant Agreement, Applicable Law and any applicable rules and regulations of the SEC and the New York Stock Exchange (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed) and, as soon as practicable following the resolution of any comments from the SEC relating to the Warrantholder Consent Solicitation Statement or a reasonable determination by Buyer that there are none, shall file with the SEC the Warrantholder Consent Solicitation Statement in definitive form and mail such Warrantholder Consent Solicitation Statement to all holders of the Buyer Public Warrants as soon as possible thereafter. Each of the Company and the Stockholders shall promptly furnish to the Buyer all information concerning such Person as may be reasonably requested in connection with the preparation, filing and distribution of the Warrantholder Consent Solicitation Statement. The Warrantholder Consent Solicitation Statement shall comply as to form in all material respects with the applicable requirements of the Exchange Act, the DGCL and the rules of the NYSE (or such other nationally

Business Combination Agreement

recognized stock exchange on which shares of Buyer Common Stock are then listed). If, at any time prior to the date that is twenty (20) days after the Warrantholder Consent Solicitation Statement is first mailed to the holders of the Buyer Public Warrants, any information relating to the Buyer, the Company, the Stockholders or any of their respective affiliates, officers or directors should be discovered by the Buyer, the Company, or the Stockholders which is required to be set forth in an amendment or supplement to the Warrantholder Consent Solicitation Statement, so that the Warrantholder Consent Solicitation Statement shall not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading, the Party that discovers such information shall promptly notify the other Parties hereto, and an appropriate amendment or supplement describing such information shall be filed with the SEC and, to the extent required by Applicable Law, disseminated to the holders of the Buyer Public Warrants. Prior to filing with the SEC, Buyer will make available to the Company drafts of the Warrantholder Consent Solicitation Statement and any other related documents to be filed with the SEC, both preliminary and definitive, and any amendment or supplement to the Warrantholder Consent Solicitation Statement or such other related document and will provide the Company with a reasonable opportunity to comment on such drafts and shall consider such comments in good faith.

(d) Buyer shall use commercially reasonable efforts to satisfy the requirements of the Exchange Act and other Applicable Laws in connection with the Warrantholder Consent Solicitation Statement and the Warrant Amendments.

ARTICLE 9 CLOSING DELIVERIES AND TERMINATION

- 9.1 <u>Closing Deliveries by the Company</u>. At Closing, the Company will deliver, or cause to be delivered, to Buyer (or as Buyer or this Agreement otherwise directs), the following:
 - (a) the written resignation (or documentation reasonably satisfactory to Buyer showing the removal) of each director, governor, officer, and manager of the Company, with each such resignation (or removal) effective no later than the First Effective Time;
 - (b) an officer's certificate of a duly authorized officer of the Company, dated the Closing Date and executed by such officer, certifying that attached thereto is a true, correct and complete copy of the resolutions of the Board of Directors of the Company, authorizing the execution, delivery and performance of this Agreement and each Ancillary Document of the Company, and the transactions contemplated herein and therein, in each case as are then in full force and effect;
 - (c) an officer's certificate of a duly authorized officer of the Company, dated the Closing Date and executed by such officer, certifying the items in Sections 10.2(a) and 10.2(b); and
 - (d) all other documents and items required by this Agreement to be delivered, or caused to be delivered, by the Company at Closing.
- 9.2 <u>Closing Deliveries by Buyer, Merger Sub and Merger Sub 2</u>. At Closing, Buyer will deliver, or cause to be delivered, to Stockholders (or as the Company or this Agreement otherwise directs), the following:
 - (a) the payments and deposits described under Section 2.4;
 - (b) an officer's certificate of a duly authorized officer of Buyer, dated the Closing Date and executed by such officer, certifying that attached thereto is a true, correct and complete copy of the resolutions of the Board of Directors of Buyer authorizing the execution, delivery and performance of this Agreement and each Ancillary Document of Buyer and the transactions

contemplated herein and therein (including authorization of the issuance of Buyer Common Stock in connection with this Agreement), in each case as are then in full force and effect;

- (c) an officer's certificate of a duly authorized officer of Buyer, dated the Closing Date and executed by such officer, certifying the items in Sections 10.3(a) and 10.3(b); and
- (d) a duly executed certificate from each of a secretary or assistant secretary of Merger Sub and the manager of Merger Sub 2, in each case, certifying that attached thereto is a true and correct copy of (i) its Organizational Documents as of immediately prior to the Mergers; (ii) a list of its officers, directors and managers as of immediately prior to the Mergers; (iii) resolutions of the board of directors or managers authorizing the execution, delivery and performance of this Agreement and each Ancillary Document to which it is a party and the transactions contemplated herein and therein, in each case as are then in full force and effect; and
- (e) all other documents and items required by this Agreement to be delivered, or caused to be delivered, by Buyer, Merger Sub or Merger Sub 2 at Closing.
- 9.3 <u>Termination of Agreement</u>. The sole and exclusive rights to terminate this Agreement before Closing (and the Party that has any such right) are as follows:
 - (a) by mutual written consent of Buyer and the Company;
 - (b) by either Buyer or the Company, if the terminating Party (including, with respect to the Company, the Stockholders, and with respect to Buyer, Merger Sub and Merger Sub 2) is not then in material breach of any provision of this Agreement or the failure to satisfy any condition to Closing or consummate the transactions contemplated herein results in any material respect from the breach by such Party of any of its representations, warranties, covenants, or agreements herein or in any Ancillary Document), if Closing has not occurred on or before November 20, 2020 (the "Outside Date"); provided, however, if this Agreement has not otherwise been terminated pursuant to this Section 9.3, and the Buyer stockholders have approved an extension of time beyond November 20, 2020 during which Buyer may consummate the transactions contemplated by this Agreement, then the Outside Date shall be automatically extended to December 31, 2020 (or such earlier date so approved by Buyer stockholders);
 - (c) by the Company, if the Buyer Common Stock is delisted from the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed) unless there is then-pending (or already accepted) application to list the Buyer Common Stock on another nationally recognized stock exchange;
 - (c) by Buyer, if any condition in **Section 10.1** or **Section 10.2** becomes incapable of fulfillment by the Outside Date, unless Buyer has waived such condition or unless such failure results in any material respect from the breach by Buyer (or Merger Sub or Merger Sub 2) of any of its representations, warranties, covenants, or agreements herein or in any Ancillary Document;
 - (d) by Buyer, if any Stockholder exercises dissenters' rights in accordance with Section 1.7;
 - (e) by Buyer, if any Stockholder or any of its Affiliates has initiated or Threatened a Proceeding that if decided adversely to Buyer or the Company would reasonably be expected to have a Material Adverse Effect on the Buyer or the Company, or materially adversely affect the ability of any Party to consummate the transactions contemplated herein;
 - (f) by the Company, if any condition in Section 10.1 or Section 10.3 becomes incapable of fulfillment by the Outside Date, unless the Company has waived such condition or unless such failure results in any material respect from the breach by the Company (or any Stockholder) of any of its representations, warranties, covenants, or agreements herein or in any Ancillary Document; or

(g) by the Company, if prior to the TO Commencement Date Buyer has not received consents meeting the requirements of the DGCL representing a sufficient number of shares to obtain the approval of the Buyer Stockholders' Written Consent, the Company may immediately terminate this Agreement.

A termination of this Agreement under any of the preceding clauses (b) through (f) will be effective two Business Days after the Party seeking termination gives to the other Party written notice of such termination.

9.4 <u>Effect of Termination</u>. Termination of this Agreement under Section 9.3 will terminate all obligations of the Parties under this Agreement, without Liability of any Party to any other Party, and, provided that the terms of Section 6.4 (Confidentiality and Publicity), Article 11 (Certain General Terms and Other Agreements), and this Section 9.4 shall survive such termination and nothing herein shall relieve any party from any Liability for any willful and material breach of the provisions of this Agreement prior to the termination of this Agreement, in which case the non-breaching party shall be entitled to all rights and remedies available at law or in equity.

ARTICLE 10 CONDITIONS TO OBLIGATIONS TO CLOSE

- 10.1 <u>Conditions to Obligations of Each Party to Close</u>. The obligations of each Party to effect the closing of the transactions contemplated herein is subject to the satisfaction at or before Closing of all of the following conditions, any one or more of which may be waived by Buyer and the Company:
 - (a) No Legal Actions. No Governmental Authority of competent jurisdiction will have instituted any Proceeding to restrain, prohibit or otherwise challenge the legality or validity of the transactions contemplated herein that has not been dismissed or otherwise resolved in a manner that does not materially and adversely affect the transactions contemplated herein and no injunction, order or decree of any Governmental Authority will be in effect that restrains or prohibits the purchase or sale of the Shares or the consummation of the other transactions contemplated herein.
 - (b) Antitrust Laws. The applicable waiting period, and any extension thereof, under the HSR Act and any other applicable Antitrust Law will have expired or been duly terminated.
 - (c) Written Consent. On the date of this Agreement, the Requisite Stockholders will have executed and delivered the Written Consent, a copy of which will have been provided to Buyer no later than the first Business Day thereafter.
 - (d) Required Buyer Stockholder Approval. On the date of this Agreement, Buyer shall have obtained the Buyer Stockholders' Written Consent described in Exhibit 8.10(a)(i) constituting the Required Buyer Stockholder Approval sufficient to approve the Buyer Stockholder Approval Matters described therein, a copy of which will have been provided to the Company no later than the first Business Day thereafter, and which Required Buyer Stockholder Approval shall remain in full force and effect.
 - (e) **Information Statement**. The Information Statement shall have been mailed to the Buyer's stockholders in accordance with **Section 8.10** and Regulation 14C of the Exchange Act at least twenty (20) days prior to the Closing Date.
 - (f) Appointment of Post-Closing Board of Directors. The Post-Closing Buyer Directors shall have been elected or appointed to the board of directors of Buyer as of the Closing.
- 10.2 <u>Conditions to Obligation of Buyer, Merger Sub and Merger Sub 2 to Close.</u> The obligation of each of Buyer, Merger Sub and Merger Sub 2 to effect the closing of the transactions

Business Combination Agreement Page 55

contemplated herein is subject to the satisfaction at or before Closing of all of the following conditions, any one or more of which may be waived by Buyer, in Buyer's sole discretion:

- (a) Accuracy of Representations and Warranties. Each representation and warranty of the Company in Article 3 (as modified by the Disclosure Schedule) will be true and correct (without regard to any qualifications as to material, materiality, material respects, Material Adverse Effect, or any similar term) as of the Closing Date as if made on the Closing Date (or, in each case, if any such representation and warranty is expressly stated to have been made as of a specific date, then, for such representation and warranty, as of such specific date), except for any inaccuracy in any representation or warranty that, individually or in the aggregate with any other such inaccuracy, has not had a Material Adverse Effect and does not materially and adversely affect the ability of the Company or the Stockholders to consummate the transactions contemplated herein.
- (b) Observance and Performance. The Company and the Stockholders will have performed or complied with, in all material respects, all covenants and agreements required by this Agreement to be performed or complied with by the Company or the Stockholders, as applicable, on or before the Closing Date.
- (c) **Payoff Letters**. On or before the Closing Date, the Company will have provided Buyer with customary pay-off letters from all holders of Closing Indebtedness to be paid under **Section 2.4**.
- (d) FIRPTA Affidavit. The Company will have delivered to Buyer a certificate, sworn under penalty of perjury and dated as of the Closing Date, stating that the Company is not and has not been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code and the applicable Treasury Regulations during the five-year period ending on the Closing Date.
- (e) **Termination of Certain Agreements.** The Company shall deliver evidence of termination of the Stockholder Agreement.
- (f) **Registration Rights Agreement**. Buyer and the Stockholders shall have delivered counterpart signatures to the Registration Rights Agreement.
- (g) Stockholder Support Agreement. On the date of this Agreement, the holder of the Company Preferred Stock and a majority of the Company Common Stock shall have duly executed the Stockholder Support Agreement, and the Company shall have delivered same to Buyer.
- (h) **Delivery of Other Items**. The Company and the Stockholders will have delivered (or caused to be delivered) to Buyer each of the other items contemplated to be so delivered by this Agreement, including each item listed in **Section 9.1**.
- 10.3 <u>Conditions to Obligation of the Company to Close</u>. The obligation of the Company to effect the closing of the transactions contemplated herein is subject to the satisfaction at or before Closing of all of the following conditions, any one or more of which may be waived by the Company, in its sole discretion:
 - (a) Accuracy of Representations and Warranties. Each representation and warranty of Merger Sub and Merger Sub 2 in Article 4 and of Buyer in Article 5 will be true and correct (without regard to any qualifications as to material, materiality, material respects, material adverse effect, or any similar term) as of the Closing Date as if made on the Closing Date (or, in each case, if any such representation and warranty is expressly stated to have been made as of a specific date, then, for such representation and warranty, as of such specific date), except for any inaccuracy in any representation or warranty that, individually or in the aggregate with any other such inaccuracy, has not had a material adverse effect on Buyer, Merger Sub or Merger Sub 2 and

does not materially and adversely affect the ability of Buyer, Merger Sub or Merger Sub 2 to consummate the transactions contemplated herein.

- (b) Observance and Performance. Each of Buyer, Merger Sub and Merger Sub 2 will have performed or complied with, in all material respects, all covenants and agreements required by this Agreement to be performed or complied with by it on or before the Closing Date.
- (c) Amended Buyer Charter. Buyer shall have delivered evidence to the Company that Buyer has filed the Amended Buyer Charter with the Secretary of State of Delaware.
- (d) **Listing**. The Buyer Common Stock issued as Closing Share Consideration and reserved for issuance in accordance with Section 2.4(b) shall have been approved for listing on the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are listed immediately prior to the Closing).
- (e) **Sponsor Support Agreement**. Buyer shall have delivered a copy of the Sponsor Support Agreement, duly executed by each party thereto.
- (f) Buyer Stockholders' Written Consent. Buyer shall have obtained and delivered to the Company the Buyer Stockholders' Written Consent described in Exhibit 8.10(a)(ii) constituting the Required Buyer Stockholder Approval sufficient to approve the Buyer Stockholder Approval Matters described therein, which Required Buyer Stockholder Approval shall remain in full force and effect.
- (g) **Delivery of Other Items.** Buyer will have delivered (or caused to be delivered) to the Company and the Stockholders each of the other items contemplated to be so delivered by this Agreement, including each item listed in **Section 9.2**.

ARTICLE 11 CERTAIN GENERAL TERMS AND OTHER AGREEMENTS

Waiver of Claims Against Trust Account. Reference is made to the IPO Prospectus. 11.1 The Company and each Stockholder represents and warrants that it has read the IPO Prospectus and understands that Buyer has established the Trust Account containing the proceeds of the IPO and overallotment shares acquired by Buyer's underwriters (including interest accrued from time to time thereon) for the benefit of Buyer's public stockholders (including overallotment shares acquired by Buyer's underwriters) (the "Public Stockholders") and that, except as otherwise described in the IPO Prospectus, Buyer may disburse monies from the Trust Account only: (a) to the Public Stockholders in the event they elect to redeem their shares of Buyer Common Stock in connection with the consummation of Buyer's initial business combination (as such term is used in the IPO Prospectus) ("Business Combination") or in connection with an extension of the deadline to consummate a Business Combination, (b) with respect to any interest earned on the amounts held in the Trust Account, as necessary to pay any franchise or income Taxes, and (c) to Buyer after or concurrently with the consummation of its Business Combination. For and in consideration of Buyer entering into this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Company and each Seller hereby agrees on behalf of itself and its Affiliates that, notwithstanding anything to the contrary in this Agreement, neither the Company nor any Stockholder does now or shall at any time prior to the Closing have any right, title, interest or claim of any kind in or to any monies in the Trust Account (or, or at any time hereafter, to distributions therefrom to Public Stockholders or otherwise occurring prior to the Closing in accordance with the terms of the Trust Agreement "Public Distributions"), or make any claim against the Trust Account (including any Public Distributions therefrom), regardless of whether such claim arises as a result of, in connection with or relating in any way to, this Agreement or any proposed or actual business relationship between Buyer or its Representatives, on the one hand, and the Company, any Stockholder or their respective Representatives, on the other hand, this Agreement or any other matter, and regardless of whether such claim arises based on contract, tort, equity or any other theory of legal liability (any and all such claims

Business Combination Agreement Page 5

are collectively referred to hereafter as the "Released Claims"). The Company and each Stockholder on behalf of itself and their respective Affiliates hereby irrevocably waives any Released Claims that the Company, such Stockholder or their respective Affiliates may have against the Trust Account now or in the future (in each case, however, prior to the consummation of a Business Combination) as a result of, or arising out of, any negotiations or Contracts with Buyer or its Representatives and will not seek recourse against the Trust Account (including any Public Distributions therefrom) for any reason whatsoever (including for an alleged breach of this Agreement or any other agreement with Buyer or its Affiliates). The Company and each Stockholder on behalf of itself and their respective Affiliates agrees and acknowledges that such irrevocable waiver is material to this Agreement and specifically relied upon by Buyer and its Affiliates to induce Buyer to enter in this Agreement, and the Company and each Stockholder further intends and understands such waiver to be valid, binding and enforceable against the Company, each Stockholder and their respective Affiliates. To the extent the Company, a Stockholder or any of their respective Affiliates commences any Proceeding based upon, in connection with, relating to or arising out of any matter relating to Buyer or its Representatives, which Proceeding seeks, in whole or in part, monetary relief against Buyer or its Representatives, the Company and each Stockholder hereby acknowledges and agrees that the Company's, each Stockholder's and their respective Affiliates' sole remedy shall be against funds held outside of the Trust Account and that such claim shall not permit any of the Company, any Stockholder or their respective Affiliates (or any Person claiming on any of their behalves or in lieu of them) to have any claim against the Trust Account (including any Public Distributions therefrom) or any amounts contained therein. Notwithstanding the foregoing, (i) nothing herein shall serve to limit or prohibit the Company's or the Stockholder Representative's right to pursue a claim against Buyer for legal relief against monies or other assets held outside the Trust Account, for specific performance or other equitable relief pursuant to Section 11.15 (including a claim for Buyer to specifically perform its obligations under this Agreement and cause the disbursement of the balance of the cash remaining in the Trust Account (after giving effect to the Redemption) to the Stockholders in accordance with the terms of this Agreement and the Trust Agreement, or for Buyer's fraud, and (ii) nothing herein shall serve to limit or prohibit any claims that the Company may have in the future against Buyer's assets or funds that are not held in the Trust Account (including any funds that have been released from the Trust Account to Buyer and any assets that have been purchased or acquired with any such funds). This Section 11.1 shall survive termination of this Agreement for any reason.

- Notices. All notices, requests, permissions, waivers, consents, and other communications hereunder must be in writing and will be deemed to have been given only (a) three Business Days following sending by registered or certified mail, postage prepaid, (b) when sent, if sent by facsimile transmission (provided that (i) the sender receives confirmation that the delivery was successful, (ii) such notice or communication is promptly thereafter delivered in accordance with clause (a), (c), or (d), and (iii) if such notice is received after 5:00 p.m. local time at the location of the recipient or is sent on a day other than a Business Day, such notice will be deemed given as of 9:00 a.m. local time at the location of the recipient on the next succeeding Business Day), (c) when delivered, if delivered personally to the intended recipient, or (d) one Business Day following sending by overnight delivery via a national courier service (receipt requested) and, in each case, addressed to a Party at the following address for such Party:
 - (i) if to the Stockholders (pre-Closing), Buyer (post-Closing), or the Company,

ONYX ENTERPRISES INT'L CORP. 1 Corporate Dr Ste C Cranbury, NJ 08512

Attention: Kailas Agrawal, Chief Financial Officer

Facsimile: (609) 964-1983

with a copy (which will not constitute notice) to:

FAEGRE DRINKER BIDDLE & REATH LLP 600 Campus Drive

Florham Park, NJ 07932 Attention: James M. Fischer Facsimile: (973) 360-9831

and

FAEGRE DRINKER BIDDLE & REATH LLP 2200 Wells Fargo Center 90 South Seventh Street Minneapolis, MN 55402 Attention: Jonathan R. Zimmerman and Kate Sherburne Facsimile: (612) 766-1600

(ii) if to Buyer (pre-Closing), Merger Sub, or Merger Sub 2,

LEGACY ACQUISITION CORP. 1308 Race Street, Suite 200 Cincinnati, OH 45202 Attention: Darryl McCall

with a copy (which will not constitute notice) to:

GRAYDON HEAD & RITCHEY LLP 312 Walnut Street, 18th Floor Cincinnati, OH 45202 Attention: Richard G. Schmalzl Facsimile: (513) 651-3836

(iii) if to Buyer Sponsor,

LEGACY ACQUISITION SPONSOR I, LLC 1308 Race Street, Suite 200 Cincinnati, OH 45202 Attention: Darryl McCall

with a copy (which will not constitute notice) to:

GRAYDON HEAD & RITCHEY LLP 312 Walnut Street, 18th Floor Cincinnati, OH 45202 Attention: Richard G. Schmalzl Facsimile: (513) 651-3836

(iv) if to the Stockholders (after the Closing) or to the Stockholder Representative:

Shareholder Representative Services LLC 950 17th Street, Suite 1400 Denver, CO 80202 Attention: Managing Director

Email: deals@srsacquiom.com Facsimile: (303) 623-0294 Telephone: (303) 648-4085

or to such other address, facsimile or email as is furnished in writing by any such Party to the other Party in accordance with the provisions of this Section 11.2.

Business Combination Agreement

- 11.3 No Survival. None of the representations and warranties contained in this Agreement, any Ancillary Document, or in any instrument delivered under this Agreement will survive the First Effective Time. From and after the Closing, the Stockholders, Buyer, Merger Sub, Merger Sub 2, the Company and their respective Representatives shall not have any further obligations, nor shall any claim be asserted or action be brought against the Stockholders, Buyer, Merger Sub, Merger Sub 2 or the Company or their respective Representatives with respect thereto. The covenants and agreements made by the Stockholders, Buyer, Merger Sub, Merger Sub 2, and the Company in this Agreement or in any Ancillary Document, including any rights arising out of any breach of such covenants or agreements, shall not survive the Closing, except for those covenants and agreements contained herein and therein that by their terms apply or are to be performed in whole or in part after the Closing (which such covenants shall survive the Closing and continue until fully performed in accordance with their terms).
- 11.4 Expenses. Whether or not the transactions contemplated by this Agreement are consummated, except as otherwise expressly provided herein, each of the Parties is responsible for the payment of its own respective costs and expenses incurred in connection with the negotiations leading up to and the performance of its respective obligations pursuant to this Agreement and the Ancillary Documents, including the fees of any attorneys, accountants, brokers or advisors employed or retained by or on behalf of such Party.

11.5 Interpretation; Construction. In this Agreement:

- (a) the table of contents and headings are for convenience of reference only and will not affect the meaning or interpretation of this Agreement;
- (b) the words "herein," "hereunder," "hereby" and similar words refer to this Agreement as a whole (and not to the particular sentence, paragraph or Section where they appear);
- (c) terms used in the plural include the singular, and vice versa, unless the context clearly requires otherwise;
- (d) unless expressly stated herein to the contrary, reference to any document means such document as amended or modified and as in effect from time to time in accordance with the terms thereof;
- (e) unless expressly stated herein to the contrary, reference to any Applicable Law means such Applicable Law as amended, modified, codified, replaced or reenacted, in whole or in part, and as in effect from time to time, including any rule or regulation promulgated thereunder;
- (f) the words "including," "include" and variations thereof are deemed to be followed by the words "without limitation";
- (g) "or" is used in the sense of "and/or"; "any" is used in the sense of "any or all"; and "with respect to" any item includes the concept "of," "under" or "regarding" such item or any similar relationship regarding such item;
- (h) unless expressly stated herein to the contrary, reference to a document, including this Agreement, will be deemed to also refer to each annex, addendum, exhibit, schedule or other attachment thereto;
- (i) unless expressly stated herein to the contrary, reference to an Article, Section, Schedule or Exhibit is to an article, section, schedule or exhibit, respectively, of this Agreement;
- (j) all dollar amounts are expressed in United States dollars and will be paid in cash (unless expressly stated herein to the contrary) in United States currency;
- (k) when calculating a period of time, the day that is the initial reference day in calculating such period will be excluded and, if the last day of such period is not a Business Day, such period will end on the next day that is a Business Day;

- (l) with respect to all dates and time periods in or referred to in this Agreement, time is of the essence;
- (m) the phrase "the date hereof" means the date of this Agreement, as stated in the first paragraph hereof; and
- (n) the Parties participated jointly in the negotiation and drafting of this Agreement and the documents relating hereto, and each Party was (or had ample opportunity to be) represented by legal counsel in connection with this Agreement and such other documents and each Party and each Party's counsel has reviewed and revised (or had ample opportunity to review and revise) this Agreement and such other documents; therefore, if an ambiguity or question of intent or interpretation arises, then this Agreement and such other documents will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the terms hereof or thereof.
- 11.6 No Third-Party Beneficiaries. Except as provided in Section 8.2 and Section 11.3, this Agreement is for the sole benefit of the Parties and their respective successors and permitted assigns and nothing herein expressed or implied gives or will be construed to give to any Person other than the Parties and such successors and permitted assigns, any legal or equitable rights, benefits, or remedy of any nature whatsoever under or by reason of this Agreement.
- 11.7 <u>Governing Law.</u> This Agreement will be construed and enforced in accordance with the substantive laws of the State of Delaware without reference to principles of conflicts of law to the extent such principles would require or permit the application of laws of another jurisdiction.
- Jurisdiction, Venue and Waiver of Jury Trial. EACH PARTY HEREBY 11.8 IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE COURT OF CHANCERY OF THE STATE OF DELAWARE OR IF JURISDICTION IS NOT PROPER IN SUCH COURT, IN ANY STATE OR FEDERAL COURT SITTING IN WILMINGTON, DELAWARE, IN ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY ANCILLARY DOCUMENT. AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY, AND TO THE RESPECTIVE COURT TO WHICH AN APPEAL OF THE DECISIONS OF ANY SUCH COURT MAY BE TAKEN, AND EACH PARTY AGREES NOT TO COMMENCE. OR COOPERATE IN OR ENCOURAGE THE COMMENCEMENT OF, ANY SUCH PROCEEDING, EXCEPT IN SUCH A COURT. EACH PARTY HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT IT MAY DO SO, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE THEREIN OF SUCH A PROCEEDING. EACH PARTY HEREBY IRREVOCABLY CONSENTS TO THE SERVICE OF A COPY OF THE SUMMONS AND COMPLAINT, AND ANY OTHER PROCESS WITH RESPECT TO ANY SUCH PROCEEDING THAT MAY BE SERVED IN ANY SUCH PROCEEDING, BY CERTIFIED MAIL, RETURN RECEIPT REQUESTED, BY DELIVERING A COPY THEREOF TO SUCH PARTY AT ITS RESPECTIVE ADDRESS, BY THE REQUIREMENTS OF SECTION 11.2 OR BY ANY OTHER METHOD PROVIDED BY APPLICABLE LAW. EACH PARTY HEREBY AGREES THAT A FINAL JUDGMENT IN ANY SUCH PROCEEDING WILL BE CONCLUSIVE AND MAY BE ENFORCED IN ANY JURISDICTION BY SUIT ON THE JUDGMENT OR BY ANY OTHER MANNER PROVIDED BY APPLICABLE LAW. EACH PARTY HEREBY EXPRESSLY WAIVES ANY RIGHT IT MAY HAVE TO A JURY TRIAL IN ANY SUCH PROCEEDING.
- 11.9 Entire Agreement; Amendment; Waiver. This Agreement, including the Disclosure Schedule, the Buyer Disclosure Schedule, other Schedules and Exhibits, constitutes the entire Agreement between the Parties pertaining to the subject matter herein and supersedes any other existing representation, warranty, covenant, agreement, or similar assurance (whether direct or indirect, written or oral, or statutory, express or implied) of any Party regarding such subject matter (and there is no other representation, warranty, covenant, agreement, or similar assurance of any Party regarding such subject matter). No supplement, modification or amendment hereof will be binding unless expressed as such and

Business Combination Agreement Page 61

executed in writing by each Party. Except to the extent as may otherwise be stated herein, no waiver of any term hereof will be binding unless expressed as such in a document executed by the Party making such waiver (and then only to the extent so expressed). No waiver of any term hereof will be a waiver of any other term hereof, whether or not similar, nor will any such waiver be a continuing waiver beyond its stated terms. Except to the extent as may otherwise be stated herein, failure to enforce strict compliance with any term hereof will not be a waiver of, or estoppel with respect to, any existing or subsequent failure to comply.

- 11.10 <u>Assignment; Binding Effect</u>. Neither this Agreement nor any right or obligation hereunder will be assigned, delegated or otherwise transferred (by operation of law or otherwise) by either Party without the prior written consent of the other Party, except as otherwise provided in this Agreement. This Agreement will be binding on and inure to the benefit of the respective permitted successors and assigns of the Parties. Any purported assignment, delegation or other transfer not permitted by this Section is void.
- 11.11 <u>Severability; Blue-Pencil</u>. If any term of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced, then all other terms of this Agreement will nevertheless remain in full force and effect, and such term automatically will be amended so that it is valid, legal and enforceable to the maximum extent permitted by Applicable Law, but as close to the Parties' original intent as is permissible.
- 11.12 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which will be considered one and the same agreement, and will become effective when one or more counterparts have been signed by each of the Parties and delivered, in person or by facsimile or electronic image scan, receipt acknowledged, to the other Party.
- 11.13 <u>Disclosure Schedules</u>. Certain information is contained in the Disclosure Schedule or the Buyer Disclosure Schedule solely for informational purposes, may not be required to be disclosed pursuant hereto (including disclosed items that do not meet or exceed stated dollar thresholds), and will not imply that such information or any other information is required to be disclosed. Inclusion of such information will not establish any level of materiality or similar threshold or be an admission that any of such information is material. Each matter disclosed in any section of the Disclosure Schedule or Buyer Disclosure Schedule in a manner that makes its relevance to one or more other sections of the Disclosure Schedule or Buyer Disclosure Schedule, respectively, reasonably apparent on the face of such disclosure will be deemed to have been appropriately included in each such other section of the Disclosure Schedule or Buyer Disclosure Schedule (notwithstanding the presence or absence of any reference in or to any section of the Disclosure Schedule or Buyer Disclosure Schedule).
- 11.14 Legal Representation; Conflicts Waiver. Recognizing that Faegre Drinker has acted as legal counsel to the Company and certain Stockholders prior to the Closing, and that Faegre Drinker may act as legal counsel to those Stockholders after the Closing, Buyer and the Company hereby waive on their own behalf, and agree to cause their Affiliates to waive, to the extent permitted by Applicable Law, any conflicts of interest that may arise in connection with Faegre Drinker representing the Stockholders after the Closing in connection with this Agreement and the transactions contemplated by this Agreement, based on the fact that Faegre Drinker may be deemed to have represented the Company prior to the Closing or in connection with the transactions contemplated in this Agreement, In addition, all communications involving attorney-client confidences between the Company, any Stockholder, and Faegre Drinker in the course of the negotiation and documentation of the transactions contemplated by this Agreement will be deemed to be attorney-client confidences that belong solely to the Stockholders (and not the Company) and may be controlled by the Stockholders. Without limiting the generality of the foregoing, upon and after the Closing, (a) the Stockholders and their post-Closing Affiliates (and not the Company) will be the sole holders of the attorney-client privilege with respect to the negotiation and documentation of the transactions contemplated hereby (collectively, the "Pre-Sale Communications"), and the Company will not be a holder of the Pre-Sale Communications, and (b) Faegre Drinker will have

Business Combination Agreement

no duty whatsoever to reveal or disclose the Pre-Sale Communications or files pertaining to the Pre-Sale Communications to the Company by reason of any attorney-client relationship between Faegre Drinker and the Company or otherwise. Any access by Buyer, Merger Sub, Merger Sub 2, or the Company to Pre-Sale Communications after the Closing (whether through the Company's computer systems or otherwise) will not amount to a waiver of the attorney-client privilege, nor will it adversely affect any rights of the Stockholders with respect to that attorney-client privilege.

11.15 Specific Enforcement.

- (a) Subject to Section 11.15(b), the Parties acknowledge and agree that, subject to the provisions of this Section 11.15, irreparable damage would occur if any of the provisions of this Agreement are not performed in accordance with their specific terms or are otherwise breached, and that monetary damages, even if available, would not be an adequate remedy therefor and therefore fully intend for specific performance to be an available remedy for breaches of this Agreement. It is accordingly agreed that, subject to Section 11.15(b), prior to the termination of this Agreement pursuant to Section 9.3, each Party shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the performance of terms and provisions of this Agreement to any court referred to in Section 11.8, without proof of actual damages, this being in addition to any other remedy to which they are entitled at law or in equity.
- (b) Each of the Parties agree that it will not oppose the granting of an injunction, specific performance and other equitable relief, on the terms set forth herein, on the basis that the other Parties have an adequate remedy at law or an award of specific performance is not an appropriate remedy for any reason at law or equity. Any party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement shall not be required to provide any bond or other security in connection with any such order or injunction.

11.16 Appointment of Stockholder Representative.

Each Stockholder hereby irrevocably appoints Shareholder Representative Services LLC as his, her or its representative ("Stockholder Representative"), as true and lawful agent and attorney-in-fact, with full power of substitution, with full power and authority to act for and on behalf of such Stockholder for all purposes of this Agreement and the Ancillary Documents, and with respect to the consummation of the transactions contemplated by this Agreement and the Ancillary Documents, agrees to be bound by the provisions of this Agreement or other such agreements and the terms of such transactions. Stockholder Representative hereby accepts such appointment. Stockholder Representative may take any and all actions that it believes are reasonably necessary or appropriate under this Agreement and the Ancillary Documents, including interpreting all of the terms and provisions of this Agreement and the Ancillary Documents. Each Stockholder acknowledges and agrees that Stockholder Representative, pursuant to this Agreement, has the exclusive authority to act on his, her or its behalf in connection with this Agreement, the Ancillary Documents and other transaction documents and related matters, including (i) authorization of payments to be made by any Stockholder or Stockholder Representative under this Agreement or any Ancillary Document, (ii) directing the distribution of all payments and amounts to any Stockholder, (iii) the giving and receiving of notices to be given or received by any Stockholder, (iv) the right to resolve, settle, defend, or dispute any claims made by Buyer under this Agreement or any Ancillary Document, including with respect to any adjustments to the Purchase Price pursuant to Section 2.5, or take any actions and exercise such other power, rights and authority as set forth herein, (v) the right to retain legal counsel and to engage accountants with respect to matters regarding this Agreement, any Ancillary Document, and related documents, (vi) the right to waive any condition or provision of this Agreement or any Ancillary Document and to take any other action under this

Business Combination Agreement Page 63

Agreement or any Ancillary Document, including amending this Agreement or any Ancillary Document in any respect, prior to or after the Closing on behalf of such Stockholder, and (vii) taking any and all other actions specified in or contemplated by this Agreement or any Ancillary Document. All decisions and actions by Stockholder Representative permitted by this Agreement shall be binding upon all Stockholders, and no Stockholder shall have any right to object, dissent, protest or otherwise contest the same and all Stockholders agree to be severally and not jointly (in accordance with their Pro Rata Shares) liable therefor. Buyer shall be entitled to deal with and rely conclusively on Stockholder Representative as provided herein as if, and with the same effect as if, Stockholder Representative constituted all of the Stockholders. The Stockholder Representative may resign at any time upon 30 day's advance written notice. Upon the death or resignation of Stockholder Representative appointed hereby, Stockholders shall appoint a replacement Stockholder Representative who shall possess and assume the rights, powers, duties and obligations of Stockholder Representative with the same effect as though such substitute representative had originally been Stockholder Representative under this Agreement.

- Stockholder Representative shall have no duties to the Stockholders or liability to the Stockholders, other than as expressly set forth in this Agreement, with respect to any action taken, omission, decision made or instruction given by Stockholder Representative in connection with this Agreement or any Ancillary Document while acting in good faith and without gross negligence or willful misconduct. The Stockholder Representative shall not be liable for any action or omission pursuant to the advice of counsel. The Stockholders will indemnify, defend and hold harmless the Stockholder Representative from and against any and all Liabilities. forfeitures, and actions (including the fees and expenses of counsel and experts and their staffs and all expense of document location, duplication and shipment) (collectively, "Representative Losses") arising out of or in connection with the Stockholder Representative's execution and performance of this Agreement and any Ancillary Documents, in each case as such Representative Loss is suffered or incurred; provided, that in the event that any such Representative Loss is finally adjudicated to have been directly caused by the gross negligence or willful misconduct of the Stockholder Representative, the Stockholder Representative will reimburse the Stockholders the amount of such indemnified Representative Loss to the extent attributable to such gross negligence or willful misconduct. If not paid directly to the Stockholder Representative by the Stockholders, any such Representative Losses may be recovered by the Stockholder Representative from (i) the funds in the Stockholder Representative Reserve Fund and (ii) any other funds that become payable to the Stockholders under this Agreement at such time as such amounts would otherwise be distributable to the Stockholders; provided, that while this Section allows the Stockholder Representative to be paid from the aforementioned sources of funds, this does not relieve the Stockholders from their obligation to promptly pay such Representative Losses as they are suffered or incurred, nor does it prevent the Stockholder Representative from seeking any remedies available to it at law or otherwise. In no event will the Stockholder Representative be required to advance its own funds on behalf of the Stockholders or otherwise. Notwithstanding anything in this Agreement to the contrary, any restrictions or limitations on liability or indemnification obligations of, or provisions limiting the recourse against non-parties otherwise applicable to, the Stockholders set forth elsewhere in this Agreement are not intended to be applicable to the indemnities provided to the Stockholder Representative under this Section. The foregoing indemnities will survive the Closing, the resignation or removal of the Stockholder Representative or the termination of this Agreement.
- (c) Upon the Closing, \$350,000 (the "Stockholder Representative Reserve Fund") will be wired to the Stockholder Representative pursuant to Section 2.4(a), which will be used for the purposes of paying directly, or reimbursing the Stockholder Representative for, any third party expenses pursuant to this Agreement and the agreements ancillary hereto. The Stockholders will not receive any interest or earnings on the Stockholder Representative Reserve Fund and irrevocably transfer and assign to the Stockholder Representative any ownership right that they

may otherwise have had in any such interest or earnings. The Stockholder Representative will not be liable for any loss of principal of the Stockholder Representative Reserve Fund other than as a result of its gross negligence or willful misconduct. The Stockholder Representative will hold these funds separate from its corporate funds, will not use these funds for its operating expenses or any other corporate purposes and will not voluntarily make these funds available to its creditors in the event of bankruptcy. As soon as practicable following the completion of the Stockholder Representative's responsibilities, the Stockholder Representative will deliver any remaining balance of the Stockholder Representative Reserve Fund to the Stockholders. For tax purposes, the Stockholder Representative Reserve Fund will be treated as having been received and voluntarily set aside by the Stockholders at the time of Closing. The parties agree that the Stockholder Representative is not acting as a withholding agent or in any similar capacity in connection with the Stockholder Representative Reserve Fund.

- (d) Any action taken by Stockholder Representative pursuant to the authority granted in this **Section 11.16** shall be effective and absolutely binding as the action of the Stockholders and Stockholder Representative under this Agreement and the Ancillary Documents.
- (e) Buyer shall be entitled to rely, and shall be fully protected in relying, upon any statements furnished to it by, and the actions of, or the failure to act by, Stockholder Representative. Buyer shall not be liable to any Stockholder for any losses sustained by any such Stockholder, arising out of or related to the performance of, or failure to perform by, Stockholder Representative of its obligations set forth in this Agreement, as applicable.

ARTICLE 12 CERTAIN DEFINITIONS

"Affiliate" means, with respect to any Person, any other Person that directly, or indirectly through one or more intermediaries, controls, is controlled by or is under common control with, such Person. For purposes of this definition, "control," "controlled by" and "under common control with," as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through the ownership of voting securities, by Contract or otherwise.

"Ancillary Document" means, with respect to a Person, any document executed and delivered by such Person, in connection with the execution and delivery of this Agreement or Closing, pursuant to the terms of this Agreement (but not including this Agreement).

"Applicable Law" means any applicable provision of any constitution, treaty, statute, law (including the common law), rule, regulation, ordinance, code or order enacted, adopted, issued or promulgated by any Governmental Authority.

"Business Day" means any day, other than a Saturday or Sunday and other than a day that banks in the state of New York are generally authorized or required by Applicable Law to be closed.

"Buyer Certificate of Incorporation" means the Certificate of Incorporation of Buyer, as amended and in effect as of the date of this Agreement.

"Buyer Class F Common Stock" means Buyer's Class F Common Stock, par value \$0.0001 per share.

"Buyer Common Stock" means Buyer's Class A Common Stock, par value \$0.0001 per share.

"Buyer Preferred Stock" means Buyer's Preferred Stock, par value \$0.0001 per share.

"Buyer Public Unit" means the units issued in Buyer's initial public offering described in the IPO Prospectus consisting of one share of Buyer Common Stock and one Buyer Public Warrant.

Business Combination Agreement

- "Buyer Public Warrants" means one whole warrant of a type that was included as part of the Buyer Public Units, entitling the holder thereof to purchase one-half share of Buyer Common Stock at a purchase price of \$5.75 per share of Buyer Common Stock.
- "Buyer Required Stockholder Approvals" means a valid stockholder consent effected pursuant to the DGCL, which remains in full force and effect as of the Closing Date, approving the Buyer Stockholder Approval Matters.
- "Buyer Securities" means the Buyer Common Stock, the Buyer Class F Common Stock, the Buyer Preferred Stock, the Buyer Public Units, the Buyer Public Warrants, and the Buyer Private Placement Warrants, collectively.
 - "Buyer Sponsor" means Legacy Acquisition Sponsor I, LLC.
- "Buyer Transaction Expenses" means all expenses of the Buyer incurred prior to the Closing in connection with the preparation, execution, and consummation of this Agreement, the transactions contemplated hereby to be consummated on or before the Closing Date, and the Closing, including fees and disbursements of attorneys, investment bankers, accountants, and other advisors and service providers (including the HSR filing fee, SEC registration fees, and fees of any proxy solicitation and investment relations firms engaged by Buyer as reasonably necessary to obtain all approvals required to consummate the transactions contemplated by this Agreement), the Deferred Underwriting Amount, termination fees payable to third parties and fees and disbursements of attorneys, investment bankers, accountants, and other advisors and service providers incurred in connection with business combinations previously considered by Buyer but not consummated, to the extent any such expenses have not been paid as of the Closing.
- "Cash" means the cash and cash equivalents of the Company (including marketable securities and short term investments), calculated in a manner consistent with the policies and procedures set forth on Schedule 12.1.
 - "Clayton Act" means the Clayton Antitrust Act of 1914, as amended.
- "Closing Net Working Capital" means the Net Working Capital as of 11:59 p.m. on the date immediately prior to the Closing Date.
 - "COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.
 - "Code" means the Internal Revenue Code of 1986, as amended.
- "Common Stockholders" means holders of Shares of Company Common Stock immediately prior to the First Effective Time.
 - "Company Common Stock" means the common stock, no par value per share, of the Company.
- "Company Intellectual Property" means all Intellectual Property that is owned or purported to be owned by the Company.
- "Company IP Agreements" means all licenses, sublicenses, consent to use agreements, settlements, coexistence agreements, covenants not to sue, waivers, releases, permissions, and other Contracts, whether written or oral, relating to the Intellectual Property to which the Company is a party, beneficiary, or otherwise bound.
- "Company IT Systems" means all software, computer hardware, servers, networks, platforms, peripherals, and similar or related items of automated, computerized, or other information technology (IT) networks and systems (including telecommunications networks and systems for voice, data, and video) owned, leased, licensed, or used (including through cloud-based or other third-party service providers) by the Company.

"Company Plan" means a Plan of which the Company is a Plan Sponsor, or to which the Company otherwise contributes or has contributed, or in which any employee of the Company otherwise participates, other than a Multiemployer Plan.

"Company Preferred Stock" means the preferred stock, no par value per share, of the Company.

"Company Transaction Expenses" means (a) all expenses of the Company and the Stockholders incurred prior to the Closing in connection with the preparation, execution, and consummation of this Agreement, the transactions contemplated hereby to be consummated on or before the Closing Date, and the Closing, including transaction-related bonuses and stock appreciation rights payable to employees of the Company upon Closing (other than as a result of Buyer's decision to terminate employees at or after Closing) and fees and disbursements of attorneys, investment bankers, accountants, the Stockholder Representative, and other advisors and service providers, payable by the Company (prior to the Closing) and which, in each case, have not been paid as of the Closing and (b) one-half of the filing fees under Section 6.1 and one-half of the SEC registration and New York Stock Exchange listing fees incurred in connection with, or the amount of which are known and will become due solely as a result of, consummation of the transactions contemplated by this Agreement, in an aggregate amount under this clause (b) not to exceed \$250,000. Company Transaction Expenses will not include any amounts that are accrued in the calculation of Closing Net Working Capital.

"Contract" means any contract, agreement, license, lease (whether for real estate, a capital lease, an operating lease or other), instrument or note, in each case that creates a legally binding obligation, and in each case whether oral or written.

"Deferred Underwriting Amount" means the portion of the underwriting discounts and commissions held in the Trust Account, which the IPO Underwriters are entitled to receive upon the Closing in accordance with the Trust Agreement.

"DGCL" means the Delaware General Corporation Law.

"Encumbrance" means any mortgage, pledge, security interest, charge, lien, option or other right to purchase, restriction or reservation or any other encumbrance whatsoever.

"Enforcement Limitation" means any applicable bankruptcy, reorganization, insolvency, moratorium or other similar Applicable Law affecting creditors' rights generally and principles governing the availability of equitable remedies.

"Environmental Claim" means any written notice alleging potential liability (including potential liability for investigatory costs, cleanup costs, governmental response costs, natural resources damages, property damages, personal injuries or penalties) arising out of or resulting from (a) the presence, or release into the environment, of any material or form of energy at any location, whether or not owned by the Company, or (b) any violation, or alleged violation, of any Environmental Law.

"Environmental Law" means any Applicable Law relating to pollution or protection of the environment or human health, including any Applicable Law relating to any emission, discharge, release or possible release of any pollutant, contaminant, hazardous or toxic material, substance or waste into air, surface water, groundwater or land, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of any pollutant, contaminant or hazardous or toxic material, substance or waste.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means, with respect to the Company, any corporation, trade or business which, together with the Company, is a member of a controlled group of corporations or a group of trades or businesses under common control within the meaning of Sections 414(b), (c), (m) or (o) of the Code.

"Federal Trade Commission Act" means the Federal Trade Commission Act of 1914, as amended.

"Founder Shares" means the shares of Buyer Class F Common Stock and Buyer Common Stock held by the Buyer Sponsor.

"Fully Diluted Shares" means the aggregate number of Shares of Company Common Stock outstanding immediately prior to the First Effective Time.

"GAAP" means United States generally accepted accounting principles, consistently applied.

"Governmental Authority" means any: (a) nation, state, county, city, district or other similar jurisdiction of any nature; (b) federal, state, local or foreign government; (c) governmental or quasi-governmental authority of any nature (including any governmental agency, branch, commission, bureau, instrumentality, department, official, entity, court or tribunal); (d) multi-national organization or body; or (e) body entitled by law to exercise any arbitrative, administrative, executive, judicial, legislative, police, regulatory or Taxing authority or power.

"Hazardous Substance" means any pollutant, contaminant, hazardous substance, hazardous waste or petroleum or fraction thereof, and any other chemical, waste, substance or material listed in or regulated by or identified in any Environmental Law.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Income Tax" means any Tax (other than sales, use, stamp, duty, value-added, business, goods and services, property, transfer, recording, documentary, conveyancing or similar Tax) based upon or measured by gross or net receipts of gross or net income (including any Tax in the nature of minimum taxes, tax preference items and alternative minimum taxes) and including any Liability arising pursuant to the application of Treasury Regulation Section 1.1502-6 or any similar provision of any Applicable Law regarding any Tax.

"Indebtedness" of any Person means, without duplication, any Liability under or for any of the following (excluding any trade payable incurred in the Ordinary Course of Business of such Person): (a) indebtedness for borrowed money (including if guaranteed or for which a Person is otherwise liable or responsible, including an obligation to assume indebtedness); (b) obligation evidenced by a note, bond, debenture or similar instrument; (c) any amount that has been called or drawn under a surety bond; (d) swap or hedging Contract; (e) banker acceptance; (f) purchase money mortgage, indenture, deed of trust or other purchase money lien or conditional sale or other title retention agreement; (g) indebtedness secured by any mortgage, indenture or deed of trust upon any asset; or (h) interest, fee or other expense regarding any of the foregoing. Indebtedness will not include any capital or operating leases or, with respect to the Company, any amounts with respect to Net Working Capital or Company Transaction Expenses.

"Intellectual Property" means any and all rights in, arising out of, or associated with any of the following in any jurisdiction throughout the world: (a) issued patents and patent applications (whether provisional or non-provisional), including divisionals, continuations, continuations-in-part, substitutions, reissues, reexaminations, extensions, or restorations of any of the foregoing, and other Governmental Authority-issued indicia of invention ownership (including certificates of invention, petty patents, and patent utility models) ("Patents"); (b) trademarks, service marks, brands, certification marks, logos, trade dress, trade names, and other similar indicia of source or origin, together with the goodwill connected with the use of and symbolized by, and all registrations, applications for registration, and renewals of, any of the foregoing ("Trademarks"); (c) copyrights and works of authorship, whether or not copyrightable, and all registrations, applications for registration, and renewals of any of the foregoing ("Copyrights"); (d) internet domain names and social media account or user names (including "handles"), whether or not Trademarks, all associated web addresses, URLs, websites and web pages, social media sites and pages, and all content and data thereon or relating thereto, whether or not Copyrights; (e) mask works, and all registrations, applications for registration, and renewals thereof; (f) industrial designs, and all Patents, registrations, applications for registration, and renewals thereof; (g) trade secrets, know-how, inventions (whether or not patentable), discoveries, improvements, technology, business and technical information,

databases, data compilations and collections, tools, methods, processes, techniques, and other confidential and proprietary information and all rights therein ("Trade Secrets"); (h) Software; (i) rights of publicity; and (j) all other intellectual or industrial property and proprietary rights.

"IPO" means the initial public offering of Buyer Public Units pursuant to the IPO Prospectus.

"IPO Underwriters" means Wells Fargo Securities, LLC, Cantor Fitzgerald & Co., Stifel, Nicolaus & Company, Incorporated, and Loop Capital Markets LLC.

"IRS" means the United States' Internal Revenue Service.

"Knowledge" has the following meaning: (a) an individual will have "Knowledge" of a particular fact or other matter if such individual is actually consciously aware of such fact or matter; (b) the Company will have "Knowledge" of a particular fact or other matter if Antonino Ciappina, Ajay Roy, or Kailas Agrawal is actually consciously aware of such fact or matter; and (b) a Person, other than an individual or the Company, will have "Knowledge" of a particular fact or other matter if any individual who is serving as a director, officer or trustee (or similar executive or position) of such Person currently has Knowledge, as stated in clause (a), of such fact or other matter.

"Liability" means any existing liability or obligation, including any obligation under any Contract).

"Licensed Intellectual Property" means all Intellectual Property owned by another Person and in which the Company holds any rights or interests granted by other Persons, including Seller or any of its Affiliates.

"Material Adverse Effect" means, with respect to a Person, any event or condition that, individually or in the aggregate, has had or is reasonably likely to have a materially adverse effect on the business of the Person, except that none of the following will be deemed to constitute, and none of the following will be taken into account in determining the occurrence (or possible occurrence) of a Material Adverse Effect: (a) the reaction (including subsequent actions) of any Person not a party to any transaction contemplated herein resulting from the public announcement of this Agreement, any transaction contemplated herein or otherwise; (b) any event or condition generally affecting any of the industries in which the Person operates, the United States economy as a whole or any foreign economy in any location where, or with respect to which, the Person has material operations, or financial, banking, or securities markets in general; (c) any national or international political or social event or condition. including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States or any of its territories, possessions or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States; (d) compliance with any term of, or the taking of any action required by, this Agreement; (e) any change in GAAP or other accounting requirement or principle or any change in Applicable Law or the interpretation thereof or effect resulting therefrom; (f) any action required to be taken under any Contract or Applicable Law; (g) the fact that the Person failed to meet its internal budgets, plans or forecasts, including regarding earnings, revenue or other financial performance measures (it being understood that the facts giving rise to such failure may be taken into account in determining whether there has been a Material Adverse Effect); or (h) any adverse event or condition regarding the Person that is, in all material respects, cured or otherwise mitigated before the earlier of (i) the Closing Date and (ii) the date on which this Agreement is terminated pursuant to Section 9.3; except, with respect to clauses (b), (c) and (e), to the extent (and only to the extent) that the Person is disproportionately impacted by such events in comparison to other Persons in the industry in which it operates.

"Multiemployer Plan" has the meaning given in Section 3(37) of ERISA.

"Net Working Capital" means the amount of (a) all current assets of the Company (consisting of Cash, accounts receivable, inventory, prepaid expenses, advances, outstanding claims receivables, deferred Tax assets, and advance Taxes paid) minus (b) all current liabilities of the Company (consisting

of accounts payable, credit card payables, and other current liabilities), in each case as of 12:01 a.m. Eastern on the date of the First Effective Time, which calculations will be made in accordance with the categories of current assets and current liabilities, and the policies and procedures, set forth on **Schedule 12.1**. Net Working Capital will not include any amounts with respect to Closing Indebtedness, or Company Transaction Expenses.

"Net Working Capital Target" means negative Eleven Million Five Hundred Thousand Dollars (-\$11,500,000).

"Offering Shares" has the meaning set forth in the Buyer Certificate of Incorporation.

"Order" means any order, writ, injunction, decree, judgment, award or determination of or from any Governmental Authority or similar binding decision of any arbitration (or similar Proceeding).

"Ordinary Course of Business" means any action (which includes, for this definition, any failure to take action), condition, circumstance or status of or regarding a Person that is: (a) consistent with the past practices of such Person and is taken or exists in the ordinary course of the normal operations of such Person; or (b) similar in nature and magnitude to actions customarily taken or not taken by such Person.

"Organizational Document" means, for any Person (in all cases, as amended through the date of this Agreement and as may be amended subsequently as expressly contemplated by this Agreement):
(a) the articles or certificate of incorporation, formation or organization (as applicable) and the by-laws or similar governing document of such Person; or (b) any limited liability company agreement, partnership agreement, operating agreement, Stockholder agreement, voting agreement, voting trust agreement or similar document of or regarding such Person.

"Party" means the Company, each Stockholder, or Buyer.

"Permit" means any license or permit from, or registration with, a Governmental Authority.

"Permitted Encumbrance" means any: (a) Encumbrance for any Tax, assessment or other governmental charge that is not yet due and payable; (b) Encumbrance arising under any original purchase price conditional sales contract or equipment lease entered into in the Ordinary Course of Business which are not, individually or in the aggregate, material to the business of the Company; (c) easement, covenant, condition or restriction of record which are not, individually or in the aggregate, material to the business of the Company; (d) easement, covenant, condition or restriction not of record which are not, individually or in the aggregate, material to the business of the Company and as to which no material violation or encroachment exists or, if such violation or encroachment exists, as to which the cure of such violation or encroachment would not materially interfere with the conduct of the business of the Company; (e) zoning or other governmentally established Encumbrance which are not, individually or in the aggregate, material to the business of the Company; (f) pledge or deposit to secure any obligation under any workers or unemployment compensation law or to secure any other public or statutory obligation; (g) mechanic's, materialmen's, landlord's, carrier's, supplier's or vendor's lien or similar Encumbrance arising or incurred in the Ordinary Course of Business of the applicable Person that secures any amount that is not overdue for a period of more than 90 days or that are being contested in good faith through (if then appropriate) appropriate proceedings; or (h) railroad trackage agreement, utility, slope or drainage easement, right-of-way easement or lease-which are of record and which are not, individually or in the aggregate, material to the business of the Company.

"Person" means any individual, partnership, corporation, limited liability company, association, joint stock company, trustee or trust, joint venture, unincorporated organization or any other business entity or association or any Governmental Authority.

"Plan" means an "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) and any other employee benefit plan, program, agreement or arrangement of any kind, including any: stock option or ownership plan; stock appreciation rights plan; stock purchase plan; phantom stock plan;

executive compensation plan; bonus, incentive compensation, deferred compensation or profit-sharing plan; or arrangement regarding any vacation, holiday, sick leave, fringe benefit, educational assistance, pre-Tax premium or flexible spending account plan or life insurance.

"Post-Closing Buyer Board" means Buyer's Board of Directors immediately after the Closing.

"Post-Closing Buyer Directors" means the following Persons: Edwin Rigaud, Darryl McCall, Richard White, and four additional directors designated by the Company, such that a majority of the Post-Closing Buyer Directors shall qualify as independent under the rules of the NYSE (or such other nationally recognized stock exchange on which shares of Buyer Common Stock are then listed).

"Pre-Closing Tax Period" means (a) any Tax period ending the day prior to the Closing Date and (b) with respect to a Straddle Period, any portion thereof ending on the day prior to the Closing Date.

"Proceeding" means any action, arbitration, audit, investigation, litigation, proceeding, or suit (whether civil, criminal or administrative), in each case that is commenced, brought, conducted or heard by or before, or otherwise involving, any Governmental Authority or arbitrator.

"Redeemed Share" means an Offering Share that has been redeemed prior to the Reference Time pursuant to the Redemption Rights (as defined in the Buyer Certificate of Incorporation) and Article IX of the Buyer Certificate of Incorporation.

"Redemption Percentage" means, as of the Reference Time, the percentage of the Offering Shares outstanding as of the date of this Agreement that have become Redeemed Shares.

"Reference Time" means immediately prior to the Closing on the Closing Date (but without giving effect to the transactions contemplated by this Agreement, including any payments by Buyer hereunder to occur as of the Closing).

"Representatives" means, as to any Person, such Person's Affiliates and the respective managers, directors, officers, employees, independent contractors, consultants, advisors (including financial advisors, counsel and accountants), agents and other legal representatives of such Person or its Affiliates.

"Requisite Stockholders" means the affirmative vote or consent required to approve the transactions contemplated by this Agreement under the Company Organizational Documents and Stockholders Agreement.

"Required Buyer Stockholder Approval" means the consents meeting the requirements of the DGCL representing a sufficient number of shares to obtain the approval of the Buyer Stockholder Approval Matters by the requisite vote of Buyer's stockholders in accordance with the DGCL.

"SEC" means the U.S. Securities and Exchange Commission.

"SEC Documents" means, with respect to any Person, each report, schedule, form, statement, registration statement, prospectus or other document filed or required to be filed or furnished with the SEC by such Person under the Securities Act or Exchange Act, together with any amendments, restatements or supplements thereto.

"Securities Act" means the Securities Act of 1933, as amended.

"Shares" means all of the outstanding shares of capital stock of the Company.

"Sherman Act" means the Sherman Antitrust Act of 1890, as amended.

"Software" means any and all computer software and code, including all new versions, updates, revisions, improvements, and modifications thereof, whether in source code, object code, or executable code format, including systems software, application software (including mobile apps), firmware, middleware, programming tools, scripts, routines, interfaces, libraries, and databases, and all related

Business Combination Agreement

specifications and documentation, including developer notes, comments and annotations, user manuals, and training materials relating to any of the foregoing.

"Stockholder" means a holder of Shares.

"Stockholders Agreement" means the Stockholders Agreement, dated July 17, 2015, the Investor Rights Agreement, dated July 17, 2015, and any voting agreement, voting trust agreement, or similar document of or regarding the Company as of the date of this Agreement.

"Stockholder Litigation" means the Proceeding in the Superior Court of New Jersey, Chancery Division, Monmouth County, Docket No. MON-C-45-18.

"Stockholder Support Agreement" means that certain Stockholder Support Agreement entered into on the date hereof among Buyer and the holder of the Company Preferred Stock and a majority of the Company Common Stock.

"Straddle Period" means any taxable period beginning on or before and ending after the Closing Date.

"Subsidiary" means, with respect to any Person, any other Person of which at least a majority of the securities or other interests, having by their terms ordinary voting power to elect a majority of the board of directors of such other Person (or others performing similar functions with respect to such other Person or to otherwise control such Person as the managing member, general partner, or similar position), is directly or indirectly owned or controlled by such first Person or by any one or more of such first Person's Subsidiaries.

"Tax" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Section 59A of the Code), customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated or other tax of any kind whatsoever, including any interest, fine, penalty or similar addition thereto, whether disputed or not.

"Tax Return" means any return, declaration, report, filing, claim for refund or information return or statement relating to any Tax, including any schedule or attachment thereto and including any amendment thereof.

"Threatened" means, with respect to any matter, that a demand, notice or statement has been made or given, in writing, that states that such matter is being or will be asserted, commenced, taken or otherwise pursued.

"Transfer" means, with respect to any Buyer Common Stock that is Closing Share Consideration or that is delivered to a Common Stockholder upon settlement of upon the final determination of the Aggregate Purchase Price pursuant to Section 2.5, (a) the sale of, offer to sell, Contract to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder with respect to such security, (b) the entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any such security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) the public announcement of any intention to effect any transaction specified in clause (a) or (b).

"Treasury Regulations" means the United States Treasury regulations issued pursuant to the Code.

"Trust Account" means the trust account established by Buyer with the proceeds of the IPO pursuant to the Trust Agreement in accordance with the IPO Prospectus.

"Trust Agreement" means that certain Investment Management Trust Agreement, effective as of November 16, 2017, by and between Legacy Acquisition Corp. and Continental Stock Transfer & Trust Company, as amended by Amendment No. 1 to Investment Management Trust Agreement, dated as of October 22, 2019, and Amendment No. 2 to Investment Management Trust Agreement, dated as of May 18, 2020.

"VWAP" means the volume weighted average per share price for the Buyer Common Stock on the New York Stock Exchange (or if the Buyer Common Stock is not then listed on the New York Stock Exchange, then on such other stock exchange or market on which such shares are then listed) from 9:30 a.m. to 4:00 p.m. Eastern Time for any 20-day trading period, as reported by Bloomberg Financial Markets.

"WARN Act" means the Worker Adjustment and Retraining Notification Act of 1988, as amended.

* * * * *

[Signature Page Follows]

IN WITNESS WHEREOF, each Party has executed this Business Combination Agreement effective as of the date first written above.

BUYER:

LEGACY ACQUISITION CORP.

Name: Edwin J. Rigaud

Its: Chief Executive Office

MERGER SUB:

EXCEL MERGER SUB I, INC.

Name: Edwin J. Rigaud

Its: Chairman and Chief Executive Officer

MERGER SUB 2:

EXCEL MERGER SUB II, LLQ

Name: Edwin J. Rigaud Its: Chairman and Chief Executive Officer

COMPANY:

ONYX ENTERPRISES INT'L CORP.

Name: Antonino Ciappina

Its: Interim General Manager

STOCKHOLDER REPRESENTATIVE:

SHAREHOLDER REPRESENTATIVE SERVICES LLC, solely in its capacity as the Stockholder Representative

By: Kinsele angilly Name: Kimberley Angilly

Its: Director

Business Combination Agreement

Signature Page

EXHIBIT A

Form of Sponsor Support Agreement

[attached]

TRADEMARK
REEL: 007280 FRAME: 0731

SPONSOR SUPPORT AGREEMENT

This SPONSOR SUPPORT AGREEMENT (this "Agreement"), dated as of September 18, 2020, is made and entered into by and among Legacy Acquisition Sponsor I LLC, a Delaware limited liability company (together with its successors, the "Sponsor"), Legacy Acquisition Corp., a Delaware corporation ("Legacy"), and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as the Stockholder Representative ("Stockholder Representative"), pursuant to the terms of the Business Combination Agreement, dated as of the date hereof, among Legacy, Excel Merger Sub I, Inc., Excel Merger Sub II, LLC, the Company, the Stockholder Representative, and each of the stockholders of the Company (the "Business Combination Agreement"). Sponsor, Legacy and Stockholder Representative shall be referred to herein from time to time collectively as the "Parties". Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Business Combination Agreement (as defined below).

WHEREAS, the Business Combination Agreement contemplates that the Parties will enter into this Agreement; and

WHEREAS, it is contemplated that pursuant to the terms and conditions of this Agreement, the Sponsor shall agree to forfeit certain Sponsor Shares and Sponsor Warrants in Legacy.

NOW, THEREFORE, in consideration of the premises and the mutual promises contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

- 1. <u>Representations and Warranties</u>. The Sponsor represents and warrants to Legacy and Stockholder Representative that the following statements are true and correct:
- (a) The Sponsor has the requisite limited liability company or other similar power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary limited liability company action on the part of the Sponsor. This Agreement has been duly and validly executed and delivered by the Sponsor and constitutes a valid, legal and binding agreement of the Sponsor, enforceable against the Sponsor in accordance with its terms.
- (b) The Sponsor is the record owner of all of the 7,500,000 outstanding shares of Legacy's Class F Common Stock (the "Sponsor Shares") and 17,500,000 warrants to purchase 8,750,000 shares of Legacy's Class A Common Stock at a price of \$11.50 per share (the "Sponsor Warrants") as of the date hereof, which constitutes all of the equity securities in Legacy held by Sponsor as of the date hereof. Immediately prior to the Closing, all of the Equity Reduction Shares (as defined herein) will be owned of record by the Sponsor, and all other Sponsor Shares and Sponsor Warrants will be owned of record by Sponsor or its direct or indirect equityholders, which Equity Reduction Shares and Equity Reduction Warrants, and such other Sponsor Shares and Sponsor Warrants owned of record by the Sponsor and any other

1

equity securities of Legacy acquired by the Sponsor in accordance with Section 4(c) hereof will constitute all of the equity securities in Legacy held by Sponsor as of immediately prior to the Closing. The Sponsor has, or will have as of the date hereof and immediately prior to giving effect to the transactions occurring on the Closing Date, as applicable, valid, good and marketable title to the Equity Reduction Shares and Equity Reduction Warrants free and clear of all Encumbrances (other than Encumbrances pursuant to this Agreement and transfer restrictions under Applicable Law or under the certificate of incorporation or bylaws of Legacy). Except for this Agreement, the Sponsor is not party to any option, warrant, purchase right, or other contract or commitment that could require the Sponsor to sell, transfer, or otherwise dispose of the Equity Reduction Shares or Equity Reduction Warrants. Except as provided in this Agreement, or the Business Combination Agreement, the Sponsor is not a party to any voting trust, proxy or other agreement or understanding with respect to the voting of the Sponsor Shares or the Sponsor Warrants. Neither the Sponsor, nor any transferees of any equity securities of Legacy initially held by the Sponsor, has asserted or perfected any rights to adjustment or other anti-dilution protections with respect to any equity securities of Legacy (including the Sponsor Shares and the Sponsor Warrants) (whether in connection with the transactions contemplated by the Business Combination Agreement or otherwise).

The execution, delivery and performance by it of this Agreement and the consummation by the Sponsor of the transactions contemplated hereby do not: (i) conflict with or result in any breach of any provision of the certificate of formation or limited liability company agreement of the Sponsor, (ii) result in a violation or breach of, or constitute (with or without due notice or lapse of time or both) a default or give rise to any right of termination, cancellation or acceleration under, any of the terms, conditions or provisions of any note, bond, mortgage, indenture, lease, license, contract, agreement or other instrument or obligation to which the Sponsor is a party or by which its properties or assets may be bound, (iii) violate any Applicable Law or Order applicable to the Sponsor or its Subsidiaries, or any of their respective properties or assets (including the Sponsor Shares and the Sponsor Warrants), as applicable, or (iv) result in the creation of any Encumbrance (other than Encumbrances pursuant to this Agreement to which it is subject or bound and transfer restrictions under Applicable Law or under the certificate of incorporation or bylaws of Legacy) upon its assets (including the Sponsor Shares and the Sponsor Warrants), except in the case of clauses (ii), (iii) and (iv) above, for violations which would not reasonably be expected to impair, delay or prevent the ability of the Sponsor to consummate the transactions contemplated by this Agreement or to otherwise perform its obligations hereunder.

2. Sponsor Equity Reduction Shares.

- (a) The Sponsor hereby agrees that, immediately prior to the Closing, the Sponsor shall automatically be deemed to irrevocably assign and transfer to Legacy, as partial consideration for the Sponsor Deferred Shares (as defined below), 3,000,000 shares of Class F Common Stock of Legacy (such shares, the "Forfeited Shares") and that from and after such time, such Forfeited Shares shall be cancelled and no longer outstanding.
- (b) The Sponsor agrees that, if after giving effect to the exercise of redemption rights by the redeeming stockholders of Legacy, the amount of funds available in the trust fund established by Legacy for the benefit of its public stockholders (the "Trust Fund")

immediately prior to the time of the consummation of the transactions contemplated in the Business Combination Agreement will be less than \$54,000,000, then immediately prior to the consummation of the transactions contemplated in the Business Combination Agreement, the Sponsor shall surrender and forfeit to Legacy and shall cease to have any rights with respect to any Equity Reduction Shares (as defined below) held by it, including not having any right to receive any Class A Common Stock or any other securities of Legacy or its affiliates in respect thereof. The number of such Equity Reduction Shares shall be calculated as follow: one (1) share of Class F common stock of Legacy for each \$16.621 shortfall in the amount of funds available in the Trust Fund below \$54,000,000 (such number of shares subject to forfeiture, in the aggregate, the "Equity Reduction Shares"), up to a maximum of 3,250,000 Equity Reduction Shares. For example, if after giving effect to the exercise of redemption rights by the redeeming stockholders of Legacy the amount of funds available in the Trust Fund immediately prior to the time of the consummation of the transactions contemplated in the Business Combination Agreement is \$30,000,000, then the number of Equity Reduction Shares to be forfeited by the Sponsor shall be 1,444,0432. The Sponsor hereby (a) agrees and acknowledges that any other rights that it might have to the Equity Reduction Shares are hereby terminated and shall be of no force or effect and (b) authorizes Legacy to take such actions as shall be necessary to evidence such surrender and forfeiture of the Equity Reduction Shares as of immediately prior to the consummation of the transactions contemplated in the Business Combination Agreement.

The Sponsor further agrees that if, and to the extent, that Buyer pays Buyer Transaction Expenses from the Trust Fund upon or after the consummation of the transactions contemplated in the Business Combination Agreement in excess of \$16,400,000, then the Sponsor shall surrender and forfeit to Legacy and shall cease to have any rights with respect to any Expense Reduction Shares (as defined below) held by it, including not having any right to receive any Class A Common Stock or any other securities of Legacy or its affiliates in respect thereof. The number of such Expense Reduction Shares shall be calculated as follow: one (1) share of Class F common stock of Legacy for each \$10.00 of such Buyer Transaction Expenses paid from funds in the Trust Fund immediately prior to the consummation of the transactions contemplated in the Business Combination Agreement in excess of \$16,400,000 (such number of shares subject to forfeiture, in the aggregate, the "Expense Reduction Shares"). In no event shall the sum of the Expense Reduction Shares and the Equity Reduction Shares exceed 3,250,000 shares of Class F Common Stock of Legacy. For example, if \$17,400,000 of Buyer Transaction Expenses are paid from the Trust Fund on or after the consummation of the transactions contemplated in the Business Combination Agreement, then the number of Expense Reduction Shares to be forfeited by the Sponsor shall be 100,0003. The Sponsor hereby (a) agrees and acknowledges that any other rights that it might have to the Expense Reduction Shares are hereby terminated and shall be of no force or effect and (b) authorizes Legacy to take such actions as shall be necessary to evidence such surrender and forfeiture of the Expense Reduction Shares on or after the consummation of the transactions contemplated in the Business

¹ For any shortfall below \$54,000,000, Class F Shares will be forfeited on a sliding scale for any shortfall of up to \$54,000,000 (from \$54,000,000 down to \$0), with a maximum of 3,250,000 Equity Reduction Shares.

² 1,444,043 Equity Reduction Shares in this example is equal to: (i) \$24,000,000 shortfall (\$54,000,000 threshold less \$30,000,000 in Trust Fund at closing in this example), divided by (ii) \$16.62 per the forfeiture formula.

³ 100,000 Expense Reduction Shares in this example is equal to: (i) \$1,000,000 excess (\$17,400,000 paid from the Trust Fundat or after the Closing less \$16,400,000 threshold in this example), divided by (ii) \$10.00 per the forfeiture formula.

Combination Agreement upon Buyer Transaction Expenses paid from the Trust Fund exceeding \$16,400,000. For the avoidance of doubt, Buyer Transaction Expenses paid prior to the consummation of the transactions contemplated in the Business Combination Agreement and Buyer Transaction Expenses paid with monies not held in the Trust Fund shall not be counted in determining whether Buyer Transaction Expenses exceed \$16,400,000 for purposes of this Section 2(c).

- (d) To the extent that the volume weighted average per share price for the shares of Class A Common Stock of Legacy on the New York Stock Exchange (or, if the Class A Common Stock of Legacy is not then listed on the New York Stock Exchange, then on such other stock exchange or market on which such shares are then listed) from 9:30 a.m. to 4:00 p.m. Eastern Time for any thirty (30) day trading period, as reported by Bloomberg Financial Markets, during the 730 calendar days after the Closing exceeds \$15.00, Legacy shall issue to the Sponsor a number of shares of Class A Common Stock of Legacy equal to 50% of the sum of the number of Equity Reduction Shares and the number of Expense Reduction Shares (the "Sponsor Deferred Shares").
- 3. <u>Sponsor Equity Reduction Warrants</u>. The Sponsor hereby agrees that, immediately prior to the Closing, the Sponsor shall automatically be deemed to irrevocably assign and transfer to Legacy, as partial consideration for the Sponsor Deferred Shares, 14,587,770 warrants to purchase shares of Class A Common Stock of Legacy held by the Sponsor (such warrants, the "<u>Equity Reduction Warrants</u>"), which excludes 2,912,230 warrants that are currently allocated to and beneficially owned by certain institutional investors of the Sponsor (the "<u>Allocated Warrants</u>") and that from and after such time, such Equity Reduction Warrants shall be cancelled and no longer outstanding.

4. Covenants.

- (a) Subject to the terms and conditions of this Agreement, the Sponsor hereby unconditionally and irrevocably agrees to take, or cause to be taken, all actions and to do, or cause to be done, all things, in each case, necessary, proper or advisable to consummate and make effective the transactions contemplated by Sections 2 and 3 of this Agreement.
- (b) From the date hereof until the earlier of the Closing and the termination of the Business Combination Agreement in accordance with its terms, the Sponsor hereby unconditionally and irrevocably agrees that it shall not, without the prior written consent of the Company, other than the transfer to any of Sponsor's direct or indirect equityholders of any Sponsor Shares or Sponsor Warrants that are not Equity Reduction Shares, Expense Reduction Shares or Equity Reduction Warrants, (i) sell, offer to sell, contract or agree to sell, hypothecate, pledge, grant any option to purchase or otherwise dispose of or agree to dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder, with respect to any equity securities of Legacy or any securities convertible into, or exercisable, or exchangeable for, equity securities of Legacy owned by it, (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any equity securities of Legacy or any securities convertible into,

or exercisable, or exchangeable for, equity securities of Legacy owned by it, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (iii) publicly announce any intention to effect any transaction specified in clauses (i) or (ii).

- (c) Prior to the Closing, the Sponsor may not acquire any equity securities in Legacy without the prior written consent of the Company.
- (d) The Sponsor hereby unconditionally and irrevocably agrees, and agrees to use its commercially reasonable efforts to take or cause to be taken actions necessary to reflect, that each Allocated Warrant is exchanged for (x) if, at the Closing, the aggregate gross cash in the Trust Fund, plus the aggregate gross proceeds received by Legacy pursuant to a potential private offering is at least equal to \$60,000,000, \$0.35 in cash and 0.065 of a share of Class A Common Stock of Legacy, (y) if, at the Closing, the aggregate gross cash in the Trust Fund, plus the aggregate gross proceeds received by Legacy pursuant to a potential private offering is less than \$60,000,000, but at least equal to \$44,000,000, \$0.25 in cash and 0.075 of a share of Class A Common Stock of Legacy, or (z) if, at the Closing, the aggregate gross cash in the Trust Fund, plus the aggregate gross proceeds received by Legacy pursuant to a potential private offering is less than \$44,000,000, \$0.18 in cash and 0.082 of a share of Class A Common Stock of Legacy. Notwithstanding the foregoing, if any of the Allocated Warrants ceases to be beneficially owned by such institutional investors of the Sponsor and become beneficially owned by the Sponsor (the "Reverted Warrants"), such Reverted Warrants shall be treated in accordance with Section 3.
- 5. <u>Termination</u>. This Agreement shall terminate, and have no further force and effect, if the Business Combination Agreement is terminated in accordance with its terms prior to the Closing.
- 6. <u>Counterparts.</u> This Agreement may be executed in one or more counterparts, all of which will be considered one and the same agreement, and will become effective when one or more counterparts have been signed by each of the Parties and delivered, in person or by facsimile or electronic image scan, receipt acknowledged, to the other Party.
- 7. Assignment; Binding Effect. Neither this Agreement nor any right or obligation hereunder will be assigned, delegated or otherwise transferred (by operation of law or otherwise) by either Party without the prior written consent of the other Party, except as otherwise provided in this Agreement. This Agreement will be binding on and inure to the benefit of the respective permitted successors and assigns of the Parties. Any purported assignment, delegation or other transfer not permitted by this Section is void.
- 8. <u>Amendment</u>. This Agreement may not be amended or modified except by an instrument in writing signed by, or on behalf of, all of the Parties hereto.
- 9. Governing Law. This Agreement will be construed and enforced in accordance with the substantive laws of the State of Delaware without reference to principles of conflicts of law to the extent such principles would require or permit the application of laws of another jurisdiction.

- 10. Severability: Blue-Pencil. If any term of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced, then all other terms of this Agreement will nevertheless remain in full force and effect, and such term is automatically will be amended that it is valid, legal and enforceable to the maximum extent permitted by Applicable Law, but as close to the Parties' original intent as possible.
- 11. Notices. All notices, requests, permissions, waivers, consents, and other communications hereunder must be in writing and will be deemed to have been given only (a) three Business Days following sending by registered or certified mail, postage prepaid, (b) when sent, if sent by facsimile transmission (provided that (i) the sender receives confirmation that the delivery was successful, (ii) such notice or communication is promptly thereafter delivered in accordance with clause (a), (c), or (d), and (iii) if such notice is received after 5:00 p.m. local time at the location of the recipient or is sent on a day other than a Business Day, such notice will be deemed given as of 9:00 a.m. local time at the location of the recipient on the next succeeding Business Day), (c) when delivered, if delivered personally to the intended recipient, or (d) one Business Day following sending by overnight delivery via a national courier service (receipt requested) and, in each case, addressed to a Party at the following address for such Party or to such other address, facsimile or email as is furnished in writing by any such Party to the other Party in accordance with the provisions of this Section 11:

If to Legacy prior to the Closing:

Address: 1308 Race Street Suite 200 Cincinnati, Ohio 45202

Attention: Darryl McCall Telephone: (505) 820-0412

Email: darrylmccall@legacyacquisition.com

with a copy to: DLA Piper

Address: 1201 West Peachtree Street, Suite 2800, Atlanta, Georgia 30309-3450

Attention: Gerry Williams Telephone: (404) 736-7891

Email: Gerry.Williams@us.dlapiper.com

If to the Sponsor:

Address: 1308 Race Street Suite 200 Cincinnati, Ohio 45202

Attention: Darryl McCall Telephone: (505) 820-0412

Email: darrylmccall@legacyacquisition.com

with a copy to: DLA Piper

Address: 1201 West Peachtree Street, Suite 2800, Atlanta, Georgia 30309-3450

Attention: Gerry Williams Telephone: (404) 736-7891

Email: Gerry.Williams@us.dlapiper.com

If to Stockholder Representative:

Address: 950 17th Street, Suite 1400, Denver, CO 80202

Attention: Managing Director Telephone: (303) 648-4085 Email: deals@srsacquiom.com

12. Entire Agreement. This Agreement, the Business Combination Agreement, and the Ancillary Documents constitute the entire agreement among the Parties hereto with respect to the subject matter hereof, and supersede all prior and contemporaneous understandings and agreements, both written and oral, with respect to such subject matter.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

7

IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Agreement as of the date first written above.

LEGACY:
LEGACY ACQUISITION, CORP.
By: Name: Edwin J. Rigaud
Title: Chairman and Chief Executive Officer
SPONSOR:
LEGACY ACQUISITION SPONSOR I LLC
Ву:
Name: Edwin J. Rigaud
Title: Managing Member

[Signature Page to Sponsor Support Agreement]

TRADEMARK REEL: 007280 FRAME: 0739 IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Agreement as of the date first written above.

STOCKHOLDER REPRESENTATIVE:	
SHAREHOLDER REPRESENTATIVE SERVICES LLC	
Ву:	
Name:	
Γitle:	

10512787.6

[Signature Page to Sponsor Support Agreement]

EXHIBIT 8.7

Form of Registration Rights Agreement

[attached]

REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of [●], is made and entered into by and among Legacy Acquisition Corp., a Delaware corporation (the "Company"), and the undersigned parties listed under Holder on the signature page hereto (each such party, together with the Company and any person or entity who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, a "Holder" and collectively the "Holders").

RECITALS

WHEREAS, on [●], 2020, the Company completed its initial Business Combination, pursuant to which the undersigned were issued an aggregate of 26,000,000 shares of Class A Common Stock, par value \$0.0001 per share, of the Company (the "Common Stock") pursuant to the terms of the Business Combination Agreement, dated as of September ___, 2020, by and among the Company, Excel Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of the Company and directly owned subsidiary of Merger Sub 2, Excel Merger Sub II, LLC, a Delaware limited liability company and direct wholly owned subsidiary of the Company ("Merger Sub 2"), Onyx Enterprises Int'l Corp., a New Jersey corporation, and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as the Stockholder Representative (as may be amended or restated, the "Business Combination Agreement") and on the terms and conditions provided in the Company's amended and restated certificate of incorporation;

WHEREAS, the Company and the Holders desire to enter into this Agreement, pursuant to which the Company shall grant the Holders certain registration rights with respect to certain securities of the Company, as set forth in this Agreement.

NOW, **THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

1.1 <u>Definitions.</u> The terms defined in this <u>Article I</u> shall, for all purposes of this Agreement, have the respective meanings set forth below. Any capitalized terms used herein and not defined shall have the meanings ascribed to them in the Business Combination Agreement.

"Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Board, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) the Company has a bona fide business purpose for not making such information public.

"Agreement" shall have the meaning given in the Preamble.

"Board" shall mean the Board of Directors of the Company.

- "Business Combination" shall mean any merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination with one or more businesses, involving the Company.
- "Business Day" means any day other than a Saturday, Sunday or a legal holiday on which commercial banking institutions in New York are authorized to close for business.
 - "Commission" shall mean the Securities and Exchange Commission.
 - "Common Stock" shall have the meaning given in the Recitals hereto.
 - "Company" shall have the meaning given in the Preamble.
 - "Demand Registration" shall have the meaning given in subsection 2.1.1.
 - "Demanding Holder" shall have the meaning given in subsection 2.1.1.
- "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
 - "Form S-1" shall have the meaning given in subsection 2.1.1.
 - "Form S-3" shall have the meaning given in subsection 2.3.1.
 - "Holders" shall have the meaning given in the Preamble.
 - "Maximum Number of Securities" shall have the meaning given in subsection 2.1.4.
- "Misstatement" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus (in the case of a Prospectus, in the light of the circumstances under which they were made) not misleading.
- "Permitted Transferees" means a person or entity to whom a holder of Registrable Securities is permitted to transfer such Registrable Securities pursuant to Section 5.2.5 of this Agreement.
 - "Piggyback Registration" shall have the meaning given in subsection 2.2.1.
 - "Pro Rata" shall have the meaning given in subsection 2.1.4.
- "Prospectus" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.
- "Prospectus Date" shall mean the date of the final prospectus filed with the Commission and relating to the Company's initial public offering.
- "Registrable Security" shall mean (a) any outstanding share of the Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of such equity security) of the Company held by a Holder as of the date of this Agreement, and (b) any other equity security of the Company issued or issuable with respect to any such share of the Common Stock by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization,

merger, consolidation or reorganization; <u>provided</u>, <u>however</u>, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred, new certificates for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; (D) such securities may be sold without registration pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) (but with no volume or other restrictions or limitations on such sale); or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"Registration" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

"Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:

- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;
- (B) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
 - (C) printing, messenger, telephone and delivery expenses;
 - (D) reasonable fees and disbursements of counsel for the Company;
- (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
- (F) reasonable fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders initiating a Demand Registration to be registered for offer and sale in the applicable Registration.

"Registration Statement" shall mean any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.

"Requesting Holder" shall have the meaning given in subsection 2.1.1.

"Securities Act" shall mean the Securities Act of 1933, as amended from time to time.

"Selling Holders" shall have the meaning given in subsection 2.3.4(a)(ii).

"Underwriter" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.

"Underwritten Registration" or "Underwritten Offering" shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.

ARTICLE II REGISTRATIONS

2.1 Demand Registration.

Request for Registration. Subject to the provisions of subsection 2.1.4 and Section 2.4 hereof, at any time and from time to time on or after the expiration of the lock-up to which such shares are subject, if any, the Holders of at least a majority in interest of the then-outstanding number of Registrable Securities (the "Demanding Holders") may make a written demand for Registration of all or part of their Registrable Securities, which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand a "Demand Registration"). The Company shall, within ten (10) days of the Company's receipt of the Demand Registration, notify, in writing, all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Requesting Holder") shall so notify the Company, in writing, within fifteen (15) days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Requesting Holder(s) to the Company, such Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall effect, as soon thereafter as practicable, but not more than forty-five (45) days immediately after the Company's receipt of the Demand Registration, the Registration of all Registrable Securities requested by the Demanding Holders and Requesting Holders pursuant to such Demand Registration, subject to Section 2.1.4 and Section 2.4. Under no circumstances shall the Company be obligated to effect more than an aggregate of three (3) Registrations pursuant to a Demand Registration under this subsection 2.1.1 with respect to any or all Registrable Securities; provided, however, that a Registration shall not be counted for such purposes unless a Form S-1 or any similar long-form registration statement that may be available at such time ("Form S-I") has become effective and all of the Registrable Securities requested by the Requesting Holders to be registered on behalf of the Requesting Holders in such Form S-1 Registration have been sold, in accordance with Section 3.1 of this Agreement.

2.1.2 Effective Registration. Notwithstanding the provisions of subsection 2.1.1 above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; provided, further, that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency the Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in

writing, but in no event later than five (5) days, of such election; <u>provided</u>, <u>further</u>, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration becomes effective or is subsequently terminated.

- 2.1.3 <u>Underwritten Offering.</u> Subject to the provisions of <u>subsection 2.1.4</u> and <u>Section 2.4</u> hereof, if a majority-in-interest of the Demanding Holders so advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such Demanding Holder or Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.1.3</u> shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Demanding Holders initiating the Demand Registration.
- Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Registration pursuant to a Demand Registration, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other Common Stock or other equity securities that the Company desires to sell and the Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggy-back registration rights held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and the Requesting Holders (if any) (pro rata based on the respective number of Registrable Securities that each Demanding Holder and Requesting Holder (if any) has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Requesting Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities of Holders (Pro Rata, based on the respective number of Registrable Securities that each Holder has so requested) exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), the Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.
- 2.1.5 <u>Demand Registration Withdrawal.</u> A majority-in-interest of the Demanding Holders initiating a Demand Registration or a majority-in-interest of the Requesting Holders (if any), pursuant to a Registration under <u>subsection 2.1.1</u> shall have the right to withdraw from a Registration pursuant to such Demand Registration for any or no reason whatsoever upon written notification to the

5

Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such Demand Registration. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration prior to its withdrawal under this <u>subsection 2.1.5</u>.

2.2 Piggyback Registration.

- 2.2.1 Piggyback Rights. If, at any time and from time to time on or after the expiration of a lock-up to which such shares are subject, if any, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including, without limitation, pursuant to Section 2.1 hereof), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.
- 2.2.2 Reduction of Piggyback Registration. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of the Common Stock or other securities that the Company desires to sell, taken together with (i) the Common Stock or other securities, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant Section 2.2 hereof, and (iii) the Common Stock or other securities, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:
- (a) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A),

the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock or other securities, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;

- If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities or the Company, then the Company shall include in any such Registration (A) first, the Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 pro rata based on the number of Registrable Securities that each Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Holders have requested to be included in such Underwritten Registration, which can be sold without exceeding the Maximum Number of Securities: (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A), (B) and (C), the Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.
- 2.2.3 Piggyback Registration Withdrawal. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this subsection 2.2.3.
- 2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a Registration pursuant to a Demand Registration effected under Section 2.1 hereof.

2.3 Registrations on Form S-3.

2.3.1 Form S-3 Rights. The Holders of Registrable Securities may at any time, and from time to time, request in writing that the Company, pursuant to Rule 415 under the Securities Act (or any successor rule promulgated thereafter by the Commission), register the resale of any or all of their Registrable Securities on Form S-3 or similar short form registration statement that may be available at such time ("Form S-3"); provided, however, that the Company shall not be obligated to effect such request through an Underwritten Offering. Within five (5) days of the Company's receipt of a written request from a Holder or Holders of Registrable Securities for a Registration on Form S-3, the Company

shall promptly give written notice of the proposed Registration on Form S-3 to all other Holders of Registrable Securities, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in such Registration on Form S-3 shall so notify the Company, in writing, within ten (10) days after the receipt by the Holder of the notice from the Company. As soon as practicable thereafter, but not more than twelve (12) days after the Company's initial receipt of such written request for a Registration on Form S-3, the Company shall register all or such portion of such Holder's Registrable Securities as are specified in such written request, together with all or such portion of Registrable Securities of any other Holder or Holders joining in such request as are specified in the written notification given by such Holder or Holders; provided, however, that the Company shall not be obligated to effect any such Registration pursuant to Section 2.3.1 hereof if (i) a Form S-3 is not available for such offering; or (ii) the Holders of Registrable Securities, together with the Holders of any other equity securities of the Company entitled to inclusion in such Registration, propose to sell the Registrable Securities and such other equity securities (if any) at any aggregate price to the public of less than \$10,000,000. The Company shall use reasonable best efforts to cause the any Registration on Form S-3 pursuant to this Section 2.3.1 to be declared effective as soon as possible after filing, and once effective, to keep such Registration continuously effective under the Securities Act at all times until all Registrable Securities and other securities covered by such Registration Statement have been disposed of in accordance with the intended method(s) of distribution set forth in the Registration Statement or such securities have been withdrawn.

Restrictions on Registration Rights. Notwithstanding anything to the contrary contained herein, if (A) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to subsection 2.1.1 and it continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective; (B) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (C) in the good faith judgment of the Board such Registration would be seriously detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be seriously detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement. In such event, the Company shall have the right to defer such filing for a period of not more than thirty (30) days; provided, further, that the Company shall not file a registration statement (or any amendment or supplement thereto) with respect to any securities of the Company for any other holder of registration statements.

2.5 Lock-Up Agreement.

2.5.1 <u>Lock-Up Period</u>. Each Holder agrees that such Holder shall not Transfer any Buyer Common Stock issued to such Holder as part of the Closing Share Consideration that may have been issued to such Common Stockholder prior to the earlier of (i) the first anniversary of the Closing Date, (ii) the date, following the 180th day after the Closing Date, on which the VWAP of Buyer Common Stock equals or exceeds \$15.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like), (iii) the date, following the 270th day after the Closing Date, on which the VWAP of Buyer Common Stock equals or exceeds \$13.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like), or (iv) Buyer's completion of a liquidation, merger, stock exchange or other similar transaction that results in all of Buyer's stockholders having the right to exchange their shares of Buyer Common Stock for cash, securities or other property (the "Lock-Up Period").

- 2.5.2 Permitted Transfers. Notwithstanding the provisions set forth in paragraphs 2.5.1, during the Lock-Up Period, Transfers of the Buyer Common Stock held by any Holder or any of their permitted transferees (that have complied with this paragraph 2.5.2), are permitted (i) in the case of an entity, as a distribution to its partners, stockholders or members; (ii) in the case of an individual, by a bona fide gift to one of the members of the individual's immediate family or to a trust, the beneficiary of which is a member of the individual's immediate family, for estate planning purposes; (iii) in the case of an individual, by virtue of laws of descent and distribution upon death of the individual; (iv) in the case of an individual, pursuant to a qualified domestic relations order; (v) if consented to in advance by the Post-Closing Buyer Board; provided, however, that in each case of clauses (i) through (iv), the permitted transferees must enter into a written agreement agreeing to be bound by these transfer restrictions and the other terms described in this Section 2.5.2 to the extent and for the duration that such terms remain in effect at the time of the Transfer.
- 2.5.3 Stop-Transfer Notice. Each Holder, inclusive of any Person to whom shares of Buyer Common Stock have been Transferred by a Common Stockholder in accordance with Section 2.5,2 of Buyer Common Stock issued pursuant to the Business Combination Agreement hereby authorizes Buyer during each applicable Lock-Up Period to cause its transfer agent for the shares of Buyer Common Stock that are subject to restriction on Transfer during the applicable Lock-Up Period ("Lock-up Shares") to decline to transfer, and to note stop transfer restrictions on the stock register and other records relating to, such Lock-up Shares for which the undersigned is the record holder and, in the case of Lock-up Shares for which the Holder is the beneficial holder but not the record holder, such Holder agrees during the applicable Lock-Up Period to cause the record holder to cause the relevant transfer agent to decline to transfer, and to note stop transfer restrictions on the stock register and other records relating to. such Lock-up Shares, if such Transfer would constitute a violation or breach of this Agreement. In addition, any Holder who is not a resident of the United States and is not an accredited investor as defined under Rule 501 of Regulation D under the Securities Act hereby authorizes Buyer to instruct its transfer agent to note that such Transaction Securities were issued under an exemption from registration under Regulation S and to utilize the CREST electronic clearing system for the purpose of compliance with legend and transfer restriction requirements as provided for by Regulation S. In addition, each Holder agrees that Buyer may issue appropriate stop transfer instructions to its transfer agent in order to ensure compliance with the Securities Act, including as to Regulation S as applicable.
- 2.5.4 <u>Release</u>. During the Lock-Up Period, the Company agrees that if it grants an early release or waiver of the lock-up restrictions imposed by the Company upon the Buyer Sponsor pursuant to that certain Amended and Restated Sponsor Lock-Up Agreement dated as of the same date hereof, then a similar release or waiver shall be provided to each Holder on a pro rata basis.

ARTICLE III COMPANY PROCEDURES

- 3.1 <u>General Procedures</u>. If at any time on or after the date of this Agreement the Company is required to effect the Registration of Registrable Securities, the Company shall use its best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:
- 3.1.1 prepare and file with the Commission as soon as practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;
- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be requested

by the Holders or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;

- 3.1.3 prior to filing a Registration Statement or prospectus, or any amendment or supplement thereto, furnish at least five Business Days prior to the anticipated filing, without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4 prior to any public offering of Registrable Securities, use its best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise he required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company arc then listed;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;
- 3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;
- 3.1.8 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus or any document that is to be incorporated by reference into such Registration Statement or Prospectus, furnish a copy thereof to each seller of such Registrable Securities or its counsel;
- 3.1.9 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;

- 3.1.10 permit a representative of the Holders (such representative to be selected by a majority of the participating Holders), the Underwriters, if any, and any attorney or accountant retained by such Holders or Underwriter to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; provided, however, that such representatives or Underwriters enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information;
- 3.1.11 obtain a "cold comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration which the participating Holders may rely on, in customary form and covering such matters of the type customarily covered by "cold comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;
- 3.1.12 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;
- 3.1.13 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- 3.1.14 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);
- 3.1.15 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$50,000,000, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any Underwritten Offering; and
- 3.1.16 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.
- 3.2 <u>Registration Expenses</u>. The Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing the Holders.
- 3.3 Requirements for Participation in Underwritten Offerings. No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary

questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.

- 3.4 Suspension of Sales: Adverse Disclosure. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than thirty (30) days, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this Section 3.4.
- 3.5 Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such fillings. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of the Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

4.1 Indemnification.

4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.

- 4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company.
- 4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.
- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.
- 4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by

reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this <u>subsection 4.1.5</u> shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in <u>subsections 4.1.1, 4.1.2</u> and <u>4.1.3</u> above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>subsection 4.1.5</u> were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this <u>subsection 4.1.5</u>. No person guilty of fraudulent misrepresentation (within the meaning of <u>Section 11(f)</u> of the Securities Act) shall be entitled to contribution pursuant to this <u>subsection 4.1.5</u> from any person who was not guilty of such fraudulent misrepresentation.

ARTICLE V MISCELLANEOUS

5.1 Notices. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, electronic mail, telecopy, telegram or facsimile. Each notice or communication that is mailed, delivered, or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third business day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, electronic mail, telecopy, telegram or facsimile, at such time as it is delivered to the addressee (with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: [•]¹, and, if to any Holder, at such Holder's address or facsimile number as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this Section 5.1.

5.2 <u>Assignment: No Third Party Beneficiaries.</u>

- 5.2.1 This Agreement and the rights, duties and obligations of the Company hereunder may not he assigned or delegated by the Company in whole or in part.
- 5.2.2 Prior to the expiration of the lock-up to which such shares are subject, if any, no Holder may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee.
- 5.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.

¹ Note to draft: To be updated to reflect post-closing address of the Company.

- 5.2.4 This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and this <u>Section 5.2</u> hereof.
- 5.2.5 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 5.1</u> hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company to be bound by the terms of this Agreement. Any transfer or assignment made other than as provided in this <u>Section 5.2</u> shall be null and void.
- 5.3 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced.
- 5.4 Governing Law; Venue; Waiver of Jury Trial. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (1) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THE AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN NEW YORK COUNTY IN THE STATE OF NEW YORK.

EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT.

- 5.5 Amendments and Modifications. Upon the written consent of the Company and the Holders of at least a majority in interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder, solely in its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.
- 5.6 Other Registration Rights. The Company represents and warrants that, except for the rights granted under that certain Registration Rights Agreement, dated as of November 16, 2017, by and among the Company and Legacy Acquisition Sponsor I LLC, a Delaware limited liability company, and the parties listed as a holder thereto (and any person or entity who thereafter becomes a party to the agreement), as may be amended or restated, no person, other than a Holder of Registrable Securities, has

any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. Further, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions with respect to the Registrable Securities and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.

5.7 Term. This Agreement shall terminate upon the earlier of (i) the tenth anniversary of the date of this Agreement or (ii) the date as of which (A) all of the Registrable Securities have been sold pursuant to a Registration Statement (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)) or (B) the Holders of all Registrable Securities are permitted to sell the Registrable Securities under Rule 144 (or any similar provision) under the Securities Act without limitation on the amount of securities sold or the manner of sale. The provisions of Section 3.5 and Article IV shall survive any termination.

[SIGNATURE PAGE FOLLOWS]

16

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

COMPANY:
LEGACY ACQUISITION CORP., a Delaware corporation
By: Name: William C. Finn Title: Chief Financial Officer
HOLDERS:
Name:

EXHIBIT 8.10(a)(i)

Buyer Stockholders Written Consent (Transaction Approval)

[attached]

LEGACY ACQUISITION CORP.

WRITTEN CONSENT OF STOCKHOLDERS

The undersigned (the "Stockholder"), being the holder of record of at least a majority of the outstanding shares of Class F common stock, par value \$0.0001 per share ("Class F Common Stock"), of Legacy Acquisition Corp., a Delaware corporation (the "Corporation"), and Class A common stock, par value \$0.0001 per share (the "Class A Common Stock" and, together with the Class F Common Stock, the "Common Stock"), of the Corporation, entitled to vote on the matters approved hereby, acting by written consent in accordance with Section 228 of the Delaware General Corporation Law (the "DGCL"), Section 7.3 of the Corrected Amended and Restated Certificate of Incorporation of the Corporation, as amended (the "Charter") and Section 2.9 of Article II of the Bylaws of the Corporation (the "Bylaws"), hereby adopts the following resolutions:

WHEREAS, the Corporation has entered into that certain Business Combination Agreement, dated as of September [●], 2020 (the "Business Combination Agreement"), by and among the Corporation, Excel Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of the Corporation and directly owned subsidiary of Merger Sub 2 ("Merger Sub 1"), Excel Merger Sub II, LLC, a Delaware limited liability company and direct wholly owned subsidiary of the Corporation ("Merger Sub 2"), [●], a New Jersey corporation (the "Seller") and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as the stockholder representative pursuant to Section 11.16 of the Business Combination Agreement, a copy of which has been provided to the Stockholder and is attached hereto as Exhibit A (capitalized terms used herein without definition shall have the respective meaning ascribed to them in the Business Combination Agreement);

WHEREAS, pursuant to the terms and subject to the conditions of the Business Combination Agreement, the Corporation desires to purchase the Seller for the Aggregate Purchase Price, which amount includes that number of shares of Class A Common Stock (as defined herein) representing the Adjustment Reserve Amount;

WHEREAS, pursuant to the terms and subject to the conditions of the Business Combination Agreement, (a) Merger Sub 1 will merge with and into the Seller (the "First Merger"), with the Seller surviving the First Merger as a wholly owned subsidiary of Merger Sub 2 (the Seller, in its capacity as the surviving corporation of the First Merger, is sometimes referred to as the "Surviving Corporation"), and (b) immediately thereafter, the Surviving Corporation will merger with and into Merger Sub 2 (the "Second Merger" and, together with the First Merger, the "Mergers"), with Merger Sub 2 being the surviving entity of the Second Merger (Merger Sub 2, in its capacity as the surviving company of the Second Merger, is sometimes referred to as the "Surviving Company");

1

WHEREAS, pursuant to the terms and subject to the conditions of the Business Combination Agreement, (a) each share of common stock, par value \$0.01 per share, of Merger Sub 1 (the "Merger Sub 1 Common Shares") issued and outstanding immediately prior to the effectiveness of the First Merger will be converted into and become one validly issued, fully paid and nonassessable share of common stock, par value \$0.01 per share, of the Surviving Corporation (the "Merger Sub 1 Share Consideration"), (b) the shares of preferred stock, no par value per share, of the Seller (other than Dissenting Shares and Cancelled Shares) shall be automatically converted into the right to receive an amount in cash equal to \$20,000,000, (c) each share of the common stock, par value \$0.01 per share, of the Seller (the "Seller Common Shares") (other than Dissenting Shares and Cancelled Shares) shall be automatically converted into the right to receive (i) a number of shares of the Class A Common Stock equal to the Estimated Purchase Price divided by \$10.00, (ii) the per share amount of payments, if any, to holders of the Seller Common Shares under Section 2.5(c) of the Business Combination Agreement, and (iii) the per share amount of any other release to holders of Seller Common Shares generally under the Business Combination Agreement, in each case determined by dividing the aggregate amount of such released amounts by the aggregate number of shares of Seller Common Shares outstanding immediately prior to the effectiveness of the First Merger (collectively, clauses (i) - (iii), the "Seller Common Share Consideration" and, together with the Merger Sub 1 Share Consideration, the "Share Consideration"):

WHEREAS, pursuant to the terms and subject to the conditions of the Business Combination Agreement, upon the effectiveness of the Second Merger, (i) each membership interest of Merger Sub 2 issued and outstanding immediately prior to the effectiveness of the Second Merger shall remain issued and outstanding as a membership interest of the Surviving Company and (ii) all shares of common stock of the Surviving Corporation shall no longer be issued and outstanding and shall automatically be cancelled and shall cease to exist without any consideration being payable therefor;

WHEREAS, the board of directors of the Corporation (the "Board of Directors") has (i) directed that, for purposes of complying with applicable provisions of Section 312.03 of the NYSE Listed Company Manual, the Seller Common Share Consideration and the related issuance of more than 20% of the Corporation's issued and outstanding shares of Class A Common Stock to the holders of Seller Common Shares in connection with the transactions contemplated by the Business Combination Agreement, and the related change of control (the "Stock Issuance"), should be submitted to the stockholders of the Corporation for approval and (ii) recommended and advised that the stockholders of the Corporation authorize and approve the Stock Issuance;

WHEREAS, pursuant to Section 251 of the DGCL, the Business Combination Agreement and the transactions contemplated thereby shall be effective and valid if declared advisable by the Board of Directors and approved

by the affirmative vote of stockholders entitled to cast a majority of all the votes entitled to be cast thereon;

WHEREAS, the Board of Directors has unanimously (i) authorized the execution and delivery of the Business Combination Agreement, (ii) declared that the Business Combination Agreement and the transactions contemplated thereby are advisable, in the best interests of the Corporation and its stockholders, and in accordance with the DGCL, (iii) directed that the Business Combination Agreement and the transactions contemplated thereby be submitted for consideration by the Corporation's stockholders and (iv) subject to the Business Combination Agreement, recommended that the stockholders of the Corporation approve the transactions contemplated by the Business Combination Agreement in accordance with the terms described therein; and

WHEREAS, the undersigned Stockholder has reviewed the Business Combination Agreement and the transactions contemplated thereby, and desires to consent to and approve the business combination and the other transaction contemplated by the Business Combination Agreement, including without limitation, the Mergers, the Adjustment Reserve Amount, the Share Consideration, and the Stock Issuance, in all respects.

NOW, THEREFORE, BE IT RESOLVED AS FOLLOWS:

RESOLVED, that the Business Combination Agreement and the transactions contemplated thereby, including without limitation, the Mergers, the Adjustment Reserve Amount and the Share Consideration, be, and each hereby is, approved, consented to, authorized and ratified in all respects, pursuant to and in accordance with the terms and subject to the conditions set forth in the Business Combination Agreement, and that the undersigned Stockholder hereby votes all of the Common Stock held by such Stockholder and entitled to vote thereon in favor of the approval, consent, authorization and ratification of the Business Combination Agreement and the transactions contemplated thereby in all respects;

FURTHER RESOLVED, that, for purposes of complying with applicable provisions of Section 312.03 of the NYSE Listed Company Manual, the Stock Issuance, and the related change of control, be, and hereby is, approved, consented to, authorized and ratified in all respects;

FURTHER RESOLVED, that this written consent shall become effective at 11:59 p.m., New York City time, on [●], 2020¹ (the "Written Consent Effective Time");

FURTHER RESOLVED, that the undersigned Stockholder acknowledges and agrees that this written consent shall be irrevocable, except that

NTD: Insert the date on which the Business Combination Agreement is signed pursuant to Section 8.10(a) of the Business Combination Agreement.

this written consent shall be null and void, rescinded and of no further force or effect following any termination of the Business Combination Agreement in accordance with its terms;

FURTHER RESOLVED, that this written consent shall have the same force and effect as if adopted and approved by the requisite vote at a duly called and convened meeting of the stockholders of the Corporation, and that the undersigned Stockholder hereby approves, consents to, authorizes and ratifies the transaction of all business represented by and set forth in this written consent; and be it

FURTHER RESOLVED, that, in accordance with Section 228(e) of the DGCL, the Corporation shall provide prompt notice of the foregoing action to each holder of Common Stock that, if such action had been taken at a meeting of the stockholders of the Corporation, would be entitled to notice of the meeting;

FURTHER RESOLVED, that this written consent may be executed in multiple counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument, and that signatures to this written consent transmitted by facsimile or PDF copy shall be deemed original signatures for all purposes, and that such execution and transmission shall be considered valid, binding and effective for all purposes.

This written consent shall be filed with the minutes and other records of the meetings of the stockholders of the Corporation.

[Signature page follows]

4

IN WITNESS WHEREOF, the undersigned Stockholder has duly executed this written consent on the date stated by the undersigned across from its signature below:

	Legacy Acquisition Sponsor I LLC
Date:	By: Name: Edwin J. Rigaud
•	Title: Managing Member

[Signature Page to Written Consent of Stockholders]

EAST\176362967.4

Exhibit A

Business Combination Agreement

EAST\176362967.4

EXHIBIT 8.10(a)(ii)

Buyer Stockholders Written Consent (Incentive Plan & ESPP Approvals)

[attached]

LEGACY ACQUISITION CORP.

WRITTEN CONSENT OF STOCKHOLDERS

The undersigned (the "Stockholder"), being the holder of record of at least a majority of the outstanding shares of Class F common stock, par value \$0.0001 per share ("Class F Common Stock"), of Legacy Acquisition Corp., a Delaware corporation (the "Corporation"), and Class A common stock, par value \$0.0001 per share (the "Class A Common Stock" and, together with the Class F Common Stock, the "Common Stock"), of the Corporation, entitled to vote on the matters approved hereby, acting by written consent in accordance with Section 228 of the Delaware General Corporation Law (the "DGCL"), Section 7.3 of the Corrected Amended and Restated Certificate of Incorporation of the Corporation, as amended (the "Charter") and Section 2.9 of Article II of the Bylaws of the Corporation (the "Bylaws"), hereby adopts the following resolutions:

WHEREAS, the Corporation has entered into that certain Business Combination Agreement, dated as of September [●], 2020 (the "Business Combination Agreement"), by and among the Corporation, Excel Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of the Corporation and directly owned subsidiary of Merger Sub 2 ("Merger Sub 1"), Excel Merger Sub II, LLC, a Delaware limited liability company and direct wholly owned subsidiary of the Corporation ("Merger Sub 2"), [●], a New Jersey corporation (the "Seller") and Shareholder Representative Services LLC, a Colorado limited liability company, solely in its capacity as the stockholder representative pursuant to Section 11.16 of the Business Combination Agreement, a copy of which has been provided to the Stockholder (capitalized terms used herein without definition shall have the respective meaning ascribed to them in the Business Combination Agreement);

WHEREAS, in connection with the Business Combination Agreement and contingent upon Stockholder approval thereof, the Board of Directors desires to adopt an equity incentive plan, to be effective post-Closing, in form and substance reasonably acceptable to Seller, that provides for the issuance of restricted stock units and other forms of equity incentive awards by the Corporation subject to an aggregate cap on such awards equal to 12% of the number of shares of Class A Common Stock on a fully-diluted basis (the "Post-Closing Equity Incentive Plan");

WHEREAS, in connection with the Business Combination Agreement and contingent upon Stockholder approval thereof, the Corporation and the Seller desire to adopt an employee stock purchase plan, in form and substance reasonably acceptable to the Seller, that provides for the issuance of Class A Common Stock by the Corporation subject to an aggregate cap on such awards equal to 5% of the number of shares of Class A Common Stock on a fully diluted basis (the "Employee Stock Purchase Plan");

1

WHEREAS, in connection with the Business Combination Agreement and contingent upon Stockholder approval thereof, the Corporation and the Seller desire to file an amended and restated certificate of incorporation (the "Amended Charter") with the Secretary of State of the State of Delaware, in form and substance reasonably acceptable to the Corporation and the Seller, which among other things, will (a) change the Corporation's name to PARTS iD, Inc., (b) designate the classes of members of the Board of Directors immediately after the Closing and (c) remove stockholders' ability to act by written consent;

WHEREAS, in connection with the Business Combination Agreement, and contingent upon Stockholder approval thereof, the Corporation and the Seller desire to approve amended and restated by laws of the Corporation (the "Amended By Laws") to remove stockholders' ability to act by written consent;

WHEREAS, pursuant to Section 251 of the DGCL, the Business Combination Agreement and the transactions contemplated thereby shall be effective and valid if declared advisable by the Board of Directors and approved by the affirmative vote of stockholders entitled to cast a majority of all the votes entitled to be cast thereon:

WHEREAS, the Board of Directors has unanimously (i) authorized the execution and delivery of the Business Combination Agreement, (ii) declared that the Business Combination Agreement and the transactions contemplated thereby are advisable, in the best interests of the Corporation and its stockholders, and in accordance with the DGCL, (iii) directed that the Business Combination Agreement and the transactions contemplated thereby be submitted for consideration by the Corporation's stockholders and (iv) subject to the Business Combination Agreement, recommended that the stockholders of the Corporation approve the transactions contemplated by the Business Combination Agreement in accordance with the terms described therein; and

WHEREAS, the undersigned Stockholder has reviewed the Business Combination Agreement and the transactions contemplated thereby, and desires to consent to and approve the business combination and the other transaction contemplated by the Business Combination Agreement, including without limitation, the Post-Closing Equity Incentive Plan, the Employee Stock Purchase Plan, the Amended Charter and the Amended By Laws, in all respects.

NOW, THEREFORE, BE IT RESOLVED AS FOLLOWS:

RESOLVED, that the Post-Closing Equity Incentive Plan, in substantially the form attached hereto as Exhibit A, be, and hereby is, approved, consented to, authorized and ratified in all respects;

FURTHER RESOLVED, that the Employee Stock Purchase Plan, in substantially the form attached hereto as Exhibit B, be, and hereby is, approved, consented to, authorized and ratified in all respects;

2

FURTHER RESOLVED, that the Amended Charter and the Amended By Laws, in substantially the forms attached hereto as <u>Exhibit C</u> and <u>Exhibit D</u>, respectively, be, and hereby are, approved, consented to, authorized and ratified in all respects;

FURTHER RESOLVED, that this written consent shall become effective at 11:59 p.m., New York City time, on [•]¹, 2020 (the "Written Consent Effective Time");

FURTHER RESOLVED, that the undersigned Stockholder acknowledges and agrees that this written consent shall be irrevocable, except that this written consent shall be null and void, rescinded and of no further force or effect following any termination of the Business Combination Agreement in accordance with its terms:

FURTHER RESOLVED, that this written consent shall have the same force and effect as if adopted and approved by the requisite vote at a duly called and convened meeting of the stockholders of the Corporation, and that the undersigned Stockholder hereby approves, consents to, authorizes and ratifies the transaction of all business represented by and set forth in this written consent; and be it

FURTHER RESOLVED, that, in accordance with Section 228(e) of the DGCL, the Corporation shall provide prompt notice of the foregoing action to each holder of Common Stock that, if such action had been taken at a meeting of the stockholders of the Corporation, would be entitled to notice of the meeting;

FURTHER RESOLVED, that this written consent may be executed in multiple counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument, and that signatures to this written consent transmitted by facsimile or PDF copy shall be deemed original signatures for all purposes, and that such execution and transmission shall be considered valid, binding and effective for all purposes.

This written consent shall be filed with the minutes and other records of the meetings of the stockholders of the Corporation.

[Signature page follows]

3

¹ NTD: Insert the date on which the Business Combination Agreement is signed pursuant to Section 8.10(a) of the Business Combination Agreement.

IN WITNESS WHEREOF, the undersigned Stockholder has duly executed this written consent on the date stated by the undersigned across from its signature below:

	Legacy Acquisition Sponsor I LLC
Date:	By: Name: Edwin J. Rigaud
	Title: Managing Member

[Signature Page to Written Consent of Stockholders]

Exhibit A

Post-Closing Equity Incentive Plan

EAST\176363023.4

Exhibit B

Employee Stock Purchase Plan

Exhibit C

Amended Charter

Exhibit D

Amended By Laws

Execution Copy

Schedule 1.3(e)

Company & Merger Sub 2 Officers

(i) Set forth below is the Company Officer List as of immediately prior to the First Effective Time:

Chief Executive Officer:

Antonino Ciappina

Chief Financial Officer:

Kailas Agrawal

Chief Operating Officer:

Ajay Roy

Vice President - Vendor Relations:

Mark Atwater

(ii) Set forth below is the Merger Sub 2 Officer List as of immediately prior to the Second Effective

Time:

Chief Executive Officer:

Antonino Ciappina

Chief Financial Officer:

Kailas Agrawal

Chief Operating Officer:

Ajay Roy

Vice President - Vendor Relations:

Mark Atwater

Graydon: 10533138.5 US.129321755.06

Schedule 2.2

Calculation of Closing Payments

Step	Description	USD	Remarks
1	Base consideration	\$260,000,000	Base consideration in Sec. 2.1(a) of the
			Business Combination Agreement
			("BCA")
2	plus/minus: Closing Net	+/- [●]	Such adjustment, as described in Sec.
	Working Capital adjustment		2.1(b)-(c) of the BCA
3	plus: Cash that will be retained	\$25,000,000	See Sec. 2.1(d) of the BCA, Cash has the
	by the Company		meaning defined in Article 12 of the BCA
4	minus: Closing Indebtedness	[•]	Closing Indebtedness, as defined in Sec.
			2.1(e) of the BCA
5	minus: Preferred Payment	\$20,000,000	Preferred Payment, as defined in Sec.
	·		2.1(f) of the BCA
6	minus: Company Transaction	[•]	Company Transaction Expenses, as
- 1	Expenses		defined in Article 12 of the BCA
7	minus: Adjustment Reserve	\$3,000,000	Adjustment Reserve Amount, as defined
	Amount		in Sec. 2.1(h) of the BCA
8	minus: Stockholder	\$350,000	Stockholder Representative Reserve Fund,
	Representative Reserve Fund		as defined in Sec. 11.16(c) of the BCA
9	minus: Indemnification Expense	\$7,500,000	See Sec. 2.1(j) of the BCA.
	Reserve Amount		
	Estimated Purchase Price	[0]	ра.

1 - Base consideration

Particulars	USD
Base consideration	\$260,000,000
Total	\$260,000,000

2 - Closing Net Working Capital adjustment

[Details of calculation be provided in accordance with the Agreement and Schedule 12.1]

3 - Cash

Particulars Particulars	USD
Cash	\$25,000,000
Total	\$25,000,000

[continues on following page]

Graydon: 10533138.5 US.129321755.06

4 - Closing Indebtedness

Particulars Particulars	USD
[separately itemize items of Closing Indebtedness]	[•]
Total	[•]

5 - Preferred Payment

Particulars Particulars	USD
Preferred Payment	\$20,000,000
Total	\$20,000,000

6 - Company Transaction Expenses

Particulars	USD
[separately itemize Company Transaction Expenses]	[•]
Total	[•]

7 - Adjustment Reserve Amount

Particulars Particulars	USD
Adjustment Reserve Amount	\$3,000,000
Total	\$3,000,000

8 - Stockholder Representative Reserve Fund

Particulars Particulars	USD
Stockholder Representative Reserve Fund	\$350,000
Total Total	\$350,000

9 – Indemnification Expense Reserve Amount

Particulars Particulars	USD
Indemnification Expense Reserve Amount	\$7,500,000
Total	\$7,500,000

Graydon: 10533138.5 US.129321755.06

Schedule 7.1

Pre-Closing Conduct of Business by the Company

None.

Graydon: 10533138.5 US.129321755.06

Schedule 12.1

Net Working Capital Calculation Policies and Procedures

Following is an illustrative example of the calculation of Net Working Capital if such calculation was completed as of July 31, 2020:

Current Assets Cash & Credit Card Receivables	42,009,252.71
Accounts Receivable (net)	321,268.85
Inventory	3,286,341.01
Prepaid Expenses/Advances/Deferred taxes	974,205.60
Total Current Assets	46,591,068.17
Current Liabilities	
Current Liabilities	48,213,980.10
Deposit/Return Reserve	1,137,374.60
Customer deposits	8,599,914.20
Sales tax payable	2,336,219.85
	60,287,488.75
Net Working Capital	(13,696,420.58)
Further de tails: Inventory	
Goods in Transit	2,662,932.66
Inventory Asset	570,682.54
Inventory Returned Not Credited	52,725.81
	3,286,341.01
Current Liabilities	- Aman - Camara - C
Accounts Payable (net) Accounts payable -GAAP reversals	47,648,683.23 (4,100,100.64)
Credit Cards Payable	1,570,680.64
Accrued Expense	1,708,749.94
AcetsPayable Allowance A/c	384,294.70
Gift Certificate	89,727.87
Inventory Received Not Billed	(13,507.98)
NY Tire Fees	71,259.95
Payroll Liabilities	854,192.39
	48,213,980.10
Deposit Reserve	
Cancellation Reserve	(3,040,618.16)
Deposit Reserve	4,177,992.76
•	1,137,374.60

Net Working Capital will be calculated in accordance with the following policies and procedures:

Graydon: 10533138.5 US.129321755.06

- 1. Net Working Capital shall be prepared and calculated in accordance with the hierarchy outlined below:
 - a. The accounting process and policies specifically set forth in this Schedule 12.1;
 - b. To the extent not addressed in paragraph (a), the accounting policies, principles, procedures, rules, practices, methodologies, categorizations, and definitions used to prepare the Company's audited balance sheet as of December 31, 2019 attached at Section 3.4(a) of the Company Disclosure Schedule; and
 - c. To the extent not addressed in paragraphs (a) and (b), GAAP, as applicable.

For the avoidance of doubt, paragraph (a) shall take precedence over paragraphs (b) and (c); paragraph (b) shall take precedence over paragraph (c). The determination of Net Working Capital shall not reflect the impact of any accounting standards that are effective for public business entities but not yet required to be adopted for all other entities as of the date of the Agreement.

- 2. Current liabilities shall exclude:
 - Any public audit adjustments relating to any material change in accounting policy or estimates as compared to the accounting policies used in in previously audited financials;
 - b. All current or deferred Tax or Tax Liabilities, contingent or otherwise; and
 - c. All deferred revenue adjustments as of the Closing Date (including deferred revenue related to customer deposits, goods in transit and accounts payable amounts, as noted above).
 - d. Indebtedness
- 3. The Net working capital so arrived at will be further adjusted for:
 - a. Expenses that would have been Company Transaction Expenses;
 - b. Any other expenses incurred by the Company in connection with the performance of its obligations under **Section 7.3** or otherwise at the request of Buyer;
 - Any fees, expenses and other Liabilities incurred by or for the account of Buyer or any of its Affiliates;
- 4. Net Working Capital shall be determined in accordance with current NetSuite financials and shall not be affected by any future reclassification or grossing up of assets or liabilities.

Graydon: 10533138.5 US.129321755.06

STATE OF NEW JERSEY DEPARTMENT OF THE TREASURY FILING CERTIFICATE (CERTIFIED COPY)

Corporation Name: ONYX ENTERPRISES INT'L, CORP.

Business Id: 0100992057 Certificate Number: 6000135234

I, THE TREASURER OF THE STATE OF NEW JERSEY, DO HEREBY CERTIFY, THAT THE ABOVE NAMED BUSINESS DID FILE AND RECORD IN THIS DEPARTMENT A MERGER ON November 23, 2020 AND THAT THE ATTACHED IS A TRUE COPY OF THIS DOCUMENT AS THE SAME IS TAKEN FROM AND COMPARED WITH THE ORIGINAL(S) FILED IN THIS OFFICE AND NOW REMAINING ON FILE AND OF RECORD.

IN TESTIMONY WHEREOF, I HAVE HEREUNTO SET MY HAND AND AFFIXED MY OFFICIAL SEAL AT TRENTON, THIS

January 12, 2021 A.D.



Sulp of Muso
ELIZABETH MAHER MUOIO
STATE TREASURER

VERIFY THIS CERTIFICATE ONLINE AT

https://wwwl.state.nj.us/TYTR_StandingCert/JSP/Verify_Cert.jsp

MRG

NOV 23 2020

STATE TREASURER

Certificate of Merger

Onyx Enterprises Int'l, Corp. (a New Jersey corporation) into and with

Excel Merger Sub II, LLC (a Delaware Limited Liability Company)

The undersigned entities have adopted an agreement and plan of merger pursuant to 14A:10-1 and pursuant to which Onyx Enterprises Int't, Corp., a New Jersey corporation will be merged into and with Excel Merger Sub II, LLC, a Delaware Limited Liability Company. The agreement and plan of merger is the Business Combination Agreement among Legacy Acquisition Corp., Excel Merger Sub I, Inc., Excel Merger Sub II, LLC, Onyx Enterprises Int'l, Corp., and Shareholder Representative Services LLC, as stockholder representative, entered into on September 18, 2020, attached to this Certificate of Merger.

1. This is a merger filing.

٠. . ۴

- 2. Name of Surviving Business Entity: Excel Merger Sub II, LLC
- 3. Names and Jurisdictions of all participating business entities including Surviving Entity:

Name	Jurisdiction	Identification # Assigned by Treasurer (if applicable)
Onyx Enterprises Int'l, Corp.	New Jersey	0100992057
Excel Merger Sub II, LLC	Delaware	0450565091

- 4. Date Merger Adopted: November 20, 2020.
- 5. Voting: The Merger was adopted by unanimous written consent of the sole shareholder without a meeting.

[Section 6 continues below]

6. The surviving business entity agrees that it may be served with process in this State in any action, suit or proceeding for the enforcement of any obligation of any domestic or

5.3109375

foreign corporation, previously amenable to suit in this State, which is a party to this merger, and in any proceeding for the enforcement of the rights of a dissenting shareholder of such domestic corporation against the surviving business entity.

The Treasurer is hereby appointed as agent to accept service of process in any such action, suit, or proceeding which shall be forwarded to the surviving business entity at the surviving business entity's Service of Process address.

The surviving business entity also agrees that it will promptly pay to the dissenting shareholders of any such domestic corporation the amount, if any, to which they may be entitled under the provisions of Title 14A.

chance under the provision	s of the 14A.
7. The Effective Time and Da	te of the merger is the time and date when filed
	uals signed this Certificate of Merger on the 20 th day of of the respective entity to the merger listed above his name.
Excel Merger Sub II, LLC	Onyx Enterprises Int'l, Corp.
Name: Dayryl McCall	Name: Antonino Ciappina
Title: President	Title: Interim General Manager, acting President, acting Chief Executive Officer, and acting Secretary
	Onyx Enterprises Int'l, Corp.
	Name: Mark Atwater

Title: Vice President

foreign corporation, previously amenable to suit in this State, which is a party to this merger, and in any proceeding for the enforcement of the rights of a dissenting shareholder of such domestic corporation against the surviving business entity.

The Treasurer is hereby appointed as agent to accept service of process in any such action, suit, or proceeding which shall be forwarded to the surviving business entity at the surviving business entity's Service of Process address.

The surviving business entity also agrees that it will promptly pay to the dissenting shareholders of any such domestic corporation the amount, if any, to which they may be entitled under the provisions of Title 14A.

7.	The Effective Time and Date of the merger	is the time and date when filed
	of the undersigned individuals signed this rember, 2020 on behalf of the respective	
Exce	l Merger Sub II, LLC	Onyx Enterprises Int'l, Corp.
	e: Darryl McCall President	Name: Antonino Ciappina Title: Interim General Manager, acting President, acting Chief Executive Officer, and acting Secretary
		Onyx Enterprises Int'l, Corp.

Name: Mark Atwater

Title: Vice President

TRADEMARK REEL: 007280 FRAME: 0784

RECORDED: 05/05/2021